

The contents of this report reflect the views of Wyle Laboratories which is responsible for the facts and the accuracy of the data presented herein. The contents do not necessarily reflect the official views or policy of the Department of Transportation. This report does not constitute a standard, specification, or regulation.



 Ratio of se Axial pos diameters The results fr put, directive engine thrust the reduction the ratio of the to primary ve mately 0.5. perceived no a simple theory curves were of tically optim from this opt of test results Key Werds Acoustics Aircraft Noise Coaxial Jets Jet Noise 	ition of primary). rom these mode vity indices and is ranging from a in overall sou the secondary n elocity. The m The maximum lise level on a pretical model also reinterpret ium, practical imum configura s. Minimum N Noise Sour Noise Pred	v nozzle relative to tests were analyzed the sideline sound 10,000 pounds to 80 and power and maxim ozzle area to prima aximum reduction was reduction was 10 dB 1500 foot sideline. of noise generation red to indicate the figet engine and to sho tion. Volume I is con- Noise ree Control iction	 and scaled to gipressure spectra and scaled to gipressure spectra and boom of the second as found to occur bin overall power These results were by coplanar coaxilow characteristic ow the noise penal detailed discussion. 18. Distribution Ster Availability is released to the Scientific and Viminia 2215 	to 10). y (±11 primary nozzle we the overall sound power out- nd perceived noise levels for a results were used to determine lived noise level as a function of d the ratio of secondary velocity at a velocity ratio of approxi- level and 11 PNdB in maximum a shown to be in accordance with al jets. The noise reduction s required to produce an acous- lities that result due to deviations on and Volume II is a compilation we clearinghouse for Federal Technical Information, Sprint	be
 Ratio of se Axial pos diameters The results fr put, directive engine thrust the reduction the ratio of t to primary ve mately 0.5. perceived no a simple theo curves were tically optim from this opt of test result Key Werds Acoustics Aircraft Noise 	ition of primary). rom these mode rity indices and is ranging from a in overall southe secondary n elocity. The m The maximum bise level on a pretical model of also reinterpret ium, practical imum configura s. Minimum N Noise Sour	A nozzle relative to tests were analyzed the sideline sound 10,000 pounds to 80 and power and maxim tozzle area to prima tozzle area to prima tozimum reduction was reduction was 10 dB 1500 foot sideline. To noise generation to indicate the f jet engine and to sh thion. Volume I is control	 area (zero to secondary velocit secondary velocit secondary velocit di and scaled to gi pressure spectra ai 0,000 pounds. The um sideline percery nozzle area and as found to occur in overall power These results were by coplanar coaxi low characteristic ow the noise pena idetailed discussional detailed discussional d	tement b 10). y (±11 primary nozzie ve the overall sound power out- nd perceived noise levels for e results were used to determine lived noise level as a function of d the ratio of secondary velocity at a velocity ratio of approxi- level and 11 PNdB in maximum e shown to be in accordance with al jets. The noise reduction s required to produce an acous- lities that result due to deviations on and Volume II is a compilation to mant i Unlimited. Document may le	be
 Ratio of se Axial pos diameters The results fr put, directive engine thrust the reduction the ratio of t to primary ve mately 0.5. perceived no a simple theory curves were tically optim from this opt of test result Xey Werds 	ition of primary). rom these mode rity indices and is ranging from a in overall sou the secondary n elocity. The m The maximum lise level on a pretical model also reinterpret ium, practical imum configura s. Minimum N	A nozzle relative to I tests were analyzed the sideline sound 10,000 pounds to 80 nd power and maxim ozzle area to prima maximum reduction was reduction was 10 dB 1500 foot sideline. of noise generation red to indicate the fight engine and to sho tion. Volume I is con-	and scaled to gi pressure spectra and 0,000 pounds. The num sideline percer ry nozzle area and as found to occur in overall power These results were by coplanar coaxi low characteristic ow the noise pena idetailed discussion 18. Distribution Ster Availability is	to 10). y (±11 primary nozzle we the overall sound power out- nd perceived noise levels for a results were used to determine lived noise level as a function of d the ratio of secondary velocity at a velocity ratio of approxi- level and 11 PNdB in maximum a shown to be in accordance with al jets. The noise reduction s required to produce an acous- lities that result due to deviations on and Volume II is a compilation forment is Unlimited. Document may	be
 Ratio of set Axial pos diameters The results fr put, directive engine thrust the reduction the ratio of t to primary ve mately 0.5. perceived no a simple theo curves were tically optim from this opt of test result 	ition of primary). rom these mode rity indices and is ranging from a in overall southe secondary n elocity. The m The maximum lise level on a pretical model also reinterpret ium, practical imum configura s.	A nozzle relative to I tests were analyzed the sideline sound 10,000 pounds to 80 and power and maxim rozzle area to prima maximum reduction was reduction was 10 dB 1500 foot sideline. of noise generation red to indicate the f jet engine and to sh thion. Volume I is c	221e area (zero to secondary velocit d and scaled to gi pressure spectra and 0,000 pounds. The num sideline perce ry nozzle area and as found to occur in overall power These results werd by coplanar coaxi low characteristic ow the noise pena idetailed discussion	to 10). y (±11 primary nozzie we the overall sound power out- nd perceived noise levels for e results were used to determine lived noise level as a function of d the ratio of secondary velocity at a velocity ratio of approxi- level and 11 PNdB in maximum e shown to be in accordance with al jets. The noise reduction s required to produce an acous- lities that result due to deviations on and Volume II is a compilation	-
 Ratio of se Axial pos diameters The results fr put, directive engine thrust the reduction the ratio of t to primary ve mately 0.5. perceived no a simple theo curves were tically optim from this opt of test result 	ition of primary). rom these mode rity indices and is ranging from a in overall southe secondary n elocity. The m The maximum lise level on a pretical model also reinterpret ium, practical imum configura	A nozzle relative to I tests were analyzed the sideline sound 10,000 pounds to 80 and power and maxim rozzle area to prima maximum reduction was reduction was 10 dB 1500 foot sideline. of noise generation red to indicate the f jet engine and to sh thion. Volume I is c	ozzie area (zero to secondary velocit d and scaled to gi pressure spectra an 0,000 pounds. Th num sideline perce ry nozzie area and as found to occur in overall power These results werd by coplanar coaxi low characteristic ow the noise pena i detailed discussio	to 10). y (±11 primary nozzie we the overall sound power out- ad perceived noise levels for e results were used to determine lived noise level as a function of d the ratio of secondary velocity at a velocity ratio of approxi- level and 11 PNdB in maximum e shown to be in accordance with al jets. The noise reduction s required to produce an acous- lities that result due to deviations on and Volume II is a compilation	
6. Abstract Model scale produced by Primary n Primary fl Ratio of s	air jets were us coaxial circula ozzle Mach nus low total tempe econdary flow econdary nozzl	sed in an anechoic r ar jets. The followin mbers (from 0.85 to rature (60°F to 800° velocity to primary no e area to primary no	oom to conduct a ng parameters wer 1.47) ?F) flow velocity (zer	parametric study on the sound e varied. o to sonic secondary velocity)	
J, Jupplementary Notes					
Washington, D.C	e Avenue, S 2. 20590	• ¥¥ •			
Systems Research	and Develop	oment Service		14. Sponsoring Agency Code	
Federal Aviation	Administrati	on		March 1968 - November	r 197
2. Sponsoring Agency Na	ame and Address			EINIAL DEDADT	d
El Segundo, Cali	ifornia 9024	5		DOT FA68WA-1889	
Wyle Laboratorie	is – Researc	h Staff		550-001-05H 11. Contract or Grant No.	
9. Performing Organizatio	on Name and Addre	55		10. Work Unit No.	
Kenneth M. Eldr	ed, et al				
. Author(s)				8. Performing Organization Report No.	
EXHAUSTS. VO	LUME I - DE	TAILED DISCUSS	SION	6. Performing Organization Code	
	E GENIERAT			November 1971	
EAR ETEID NICTS	, 1	l			
FAA-RD-71-101, Title and Subtitle FAR FIELD NIOTS	T	1			

. .

were a service and the second

2.50 2.50

'n

c

a: [₽].

ľ

in street

PREFACE

The author wishes to thank Mr. William Sperry of the Office of Noise Abatement, FAA, for his counsel and patience throughout this program, and for his most helpful detailed review of the final report and its findings.

In addition, sincere appreciation is due to several members of the Wyle Laboratories Research Staff who contributed significantly to the program, the report, or both. Victor Mason accomplished much of the original data analysis and interpretation; Elizabeth Cuadra and Richard Potter planned and directed the experimental and preliminary analysis phases; John Wilson conducted the detailed experiments. All of these contributed significant portions of the drafts of this report and its value owes much to their efforts.

ീ

TABLE OF CONTENTS

11077 PC

1.0	INTR	ODUCTION	1
2.0	NOI	SE GENERATION BY COAXIAL JET FLOWS	3
	2.1 2.2 2.3	Noise Generation of Jets from Single Circular Nozzles Noise Prediction for Jet Flows from Mixing Nozzles Noise Prediction for Jet Flows from Coplanar Coaxial Nozzles	3 13 19
3.0	EXPE	RIMENTAL APPARATUS AND DATA ANALYSIS PROCEDURES	31
	3.1 3.2	Jet Flow Parameters and Design of the Experiment Test Hardware	31 33
		 3.2.1 Jet Rig 3.2.2 Jet Nozzles 3.2.3 Air Supply 3.2.4 Jet Traversing Mechanism 3.2.5 Anechoic Room 	33 36 39 40 40
	3.3	Instrumentation	41
		 3.3.1 Acoustical Data Acquisition System 3.3.2 Acoustical Data Analysis System 3.3.3 Thrust Measurement 3.3.4 Jet Profile Measurements 3.3.5 Discrete Tones and the Resulting Hardware Modifications 	41 41 46 46 47
	3.4	Analytical Basis for Data Reduction Computer Program	48
		 3.4.1 Flow Velocities and Thrust 3.4.2 Sound Pressure Levels 3.4.3 Sound Power Levels 3.4.4 Directivity Patterns 3.4.5 Acoustic Conversion Efficiency 3.4.6 Sideline Sound Pressure Levels for Full-Scale Engines 3.4.7 Perceived Noise Level 	50 51 53 53 53 54 58
4.0	DESC	RIPTION OF THE EXPERIMENTAL DATA	59
	4.1 4.2	Computer Output Data Some Experimental Anomalies	59 63

v

TABLE OF CONTENTS ... continued

.

5.0	DISC FULL	USSION OF RESULTS AND THEIR APPLICATION TO SCALE ENGINES	71
	5.1	Discussion of Principal Experimental Results	71
		5.1.1 Noise of the Primary Nozzles	71
		5.1.2 Reduction in Overall Sound Power Level Associated with Coaxial Jet Flows	75
		5.1.3 Changes in Sound Power Spectra and Directivity	86
		5.1.4 Change in Maximum Sideline Perceived Noise Level	89
	5.2	Prediction of the Noise Characteristics of Full–Scale Coaxial Jets	97
	5.3	Comparison of Noise Characteristics of Practical Coaxial Jet Engines with Optimum Single–Nozzle Engines	110
6.0	SUM	MARY AND CONCLUSIONS	119
REFE	RENC	ES	123
APPE	NDIX	A - DIRECTORY OF EXPERIMENTAL RUNS	A-1
APPE	NDIX	B - JET FLOW PROFILE MEASUREMENTS	B-1
APPE	NDIX	C - COMPUTER PROGRAM FOR DATA ANALYSIS	C-1
APPE	NDIX	D - SUMMARY OF OVERALL SOUND POWER LEVEL DATA	D-1
APPE	INDIX	E - SUMMARY OF MAXIMUM PERCEIVED NOISE LEVEL DATA	E-1
Appe	NDIX	F - PRIMARY JET SOUND POWER SPECTRA AND DIRECTIVITY DATA	F-1
APPE	NDIX	G - NOISE OF A SUPERSONIC NOZZLE OPERATING AT OFF-DESIGN PRESSURE RATIOS	G-1

3

LIST OF TABLES

....

E S

.

.....

Table		Page
1	Calculated Ideal Primary Flow Velocities for Various Temperatures and Pressure Ratios	32
2	Summary of Runs Completed	33
3	Primary Velocity Ratios (U _p /U _s) for a 3 dB and 6 dB Increase in Overall Power Level from 'Figure 64	109
4	Primary Velocity Ratios (U_p/U_s) for a 3 dB and 6 dB Increase in Sideline Perceived Noise Level from Figure 65	109
5	Secondary–Primary Velocity Ratio for Specified Increase in Overall Sound Power	120
6	Secondary–Primary Velocity Ratio for Specified Increase in Perceived Noise Level	121
	Appendix A	
A-1	Directory of Experimental Run Conditions	A-2
A-2	Experimental Run Conditions	A-3
	Appendix B	
8-1	Test Conditions for Flow Profile Data	8-2

LIST OF FIGURES

. 2.

- 1.25 Sec.

......

Figure	<u>ا</u>	Page
1	Relationship Between Overall Sound Power Level, Normalized by Nozzle Size and Velocity for Jets from Circular Nozzles at Pressure Ratios Less Than 2.3.	4
2	Normalized Sound Power Spectrum Level for Hot and Cold Model Jets and Turbojet Engines.	5
3	Far-field Directional Characteristics of the Overall Sound Pressure Level for Four Types of Jet Flow.	4
4	Variation of the Predominant Propagation Angle for Various Frequencies from Measurements Alung the 10–Degree Boundary of a J–57 Engine as a Function of a Non–dimensional Axial Frequency Parameter.	6
5	Comparison of Assumed Velocity Profile for Axisymmetric Constant Density Jet with Mach 0.7.	7
6	Examples of Flow for Constant Density Axisymmetric Single-Nozzle Jets.	9
7	Variation in Core Length (x _t) for Constant Derisity Axisymmetric Jet as a Function of Mach Number .	10
8	Calculated increase in the Ratio of the Length of the Jet Core with Non–Zero External Velocity (U_{∞}) to its Length with Zero External Velocity for Constant Density Jets.	10
9	Source Strength Distribution Along Jet Axis.	11
10	Jet Source Normalized Sound Power Spectra,	12
11	Idealized Power Spectra of Non-Interfering Axisymmetric Jets Having a Thrust of 10,000 Lbs, Exit Total Temperature 1720 ⁰ F, Together with Two Jets of Lower Temperature and Exit Velocity.	74
12	Sketch of Two Nozzles in the Plane Containing the Axis of a Simplified Axisymmetric Peripheral Tube Nozzle Illustrating Development of Final Mixed Jet and the Mixing of the Individual Jets.	15

ł,

Figure		Page
13	Example of Flow Data and Comparison with Calculated Profiles for Three–Lobed Nozzle at Mach 0.3.	16
14	Comparison of Calculated Power Spectra with Power Spectra Derived from Measured Power Reduction for Two 12–Lobe Nozzles with J–57 Engine.	17
15	Computed Variation in Total Acoustic Power for Idealized Mixing Nozzle Relative to that of Baseline Single Circular Nozzle for $T_p/T_o = 3$, as a Function of the Ratio of the Envelope Diameter of the Mixing Nozzle to the Diameter of the Baseline Single Nozzle, with Tube Element Diameter as a Parameter.	18
16	Approximate Velocity Profiles at Various Downstream Stations for Constant Density Annular Coplanar Jet with $U_s/U_p = .5$ and $r_s/r_p = 1.5$.	19
17	Sketch of Equivalent Jet Flow Profile for Two Cases,	21
18	Calculated Regions of Complete and Incomplete Shrouding of the Primary Core as a Function of Area and Velocity Ratios.	23
19	Relationship Between Radius and Velocity of Equivalent Jet and the Parameters of a Constant Density Coplanar Coaxial Jet.	25
20	Estimated Contributions to and Total Reduction in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with an Area Ratio of 10 and a Fixed Thrust.	28
21	Estimated Reduction in Overall Sound Power Level with Velocity Ratio for Coplanar Coaxial Jets of Fixed Thrust and Various Area Ratios.	29
22	The Jet Flow Assembly.	34
23	Photograph of Jet Flow Assembly.	34
24	Primary Nozzles and Nozzle Extention Pieces (Photograph).	37
25	The Four Secondary Nozzles (Photograph).	38

Figure		Page
26	Inverse Square Law Measurements in the Anechoic Room Using Full Octave Bands of Random Noise.	42
27	Block Diagram of Acoustic Data Acquisition System.	43
28	Block Diagram of Acoustic Data Analysis System.	44
29	The Functional Flow Loop for the Computer Program.	49
30	Geometry for Obtaining Sideline Sound Pressure Levels from Measured Data.	55
31	Atmospheric Attenuation Utilized to Compute Sideling Sound Pressure Level.	56
32	Example of Computer Output for One Computer Run.	61
33	Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.	62
34	Variation in Maximum Perceived Noise Level on a 1500–Foot Side– line with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.	62
35	Power Spectra for Unity Velocity Ratio and Various Area Railos for Coplanar Jet with Primary Nazzle Pressure Ratio of 1.6 and Temper- ature of 60°F.	64
36	Variation in Maximum Perceived Noise Level on a 1500–Foot Side– line with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.	66
37	Directivity Pattern for Run 128 (Coplanar) Showing the Anomaly in the High Octave Bands,	67
38	One-Tenth Octave Band Analysis of Directivity Anomaly for Run 128 at 110 Degrees.	68
39	Variation in Measured Overall Sound Power Level with Velocity for the Model Scale Primary Nozzles Alone.	72

Y

2

Figure		Page
40	Variation in Overall Sound Power Level with Velocity for the Primary Nozzles Alone, Scaled to 20,000 Lbs Thrust.	73
41	Variation in Maximum 1500–Foot Sideline Perceived Noise Level for the Primary Nozzles Alone, Scaled to 20,000 Lbs Thrust.	74
42	Effect of Jet Thrust on Calculated Maximum Perceived Noise Level on 1500–Foot Sideline for the Primary Nozzles of Figure 41.	76
43	Change in Overall Sound Power Level with Velocity Ratio for Coplanar Coaxial Jets of 20,000 Lbs Thrust and Four Area Ratios.	77-78
44	Change in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 1 and Thrust of 20,000 Lbs.	80
45	Change in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 2 and Thrust of 20,000 Lbs.	81
46	Change in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 5 and Thrust of 20,000 Lbs.	82
47	Change in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 10 and Thrust of 20,000 Lbs.	. 83
48	Effect of Axial Position or the Primary Nozzle as Measured by the Change in Overall Sound Power Level Relative to the Level for Each Nozzle and Pressure Combination in the Coplanar Position.	85
49	Model Jet Sound Power Spectrum for Coplanar Nozzles (Primary Nozzle Pressure Ratio 1.6, Primary Nozzle Temperature 450 ⁰ F, Area Ratio 10, Thrust Variable).	87
50	Madel Jet Sound Power Spectrum for Fully Extended Primary Nozzle (Primary Nozzle Pressure Ratio 1.6, Primary Nozzle Temperature 450 ⁰ F, Area Ratio 10, Thrust Variable).	88
51	Effect of Area Ratio of Shrouding of Model Primary Jet (Primary Nozzle Pressure Ratio 1.6, Primary Nozzle Temperature 450°F, Velocity Ratio 0.38, Thrust Variable).	90

9

.

٨

Figure		Page
52	Directivity Pattern for a Primary Nozzle Alone.	91
53	Directivity Pattern of Coplanar Nozzle with Velocity Ratio of 0.375.	92
54	Directivity Pattern for Coplanar Nozzle with Velocity Ratio of 0.731.	93
55	Directivity Pattern for Coplanar Nozzle with Velocity Ratio of 0.88.	94
56	Change in 1500–Foot Sideline Maximum Perceived Noise Level with Velocity Ratio for Coplanar Coaxial Jets of 20,000 Lbs Thrust and Four Area Ratios.	95-96
57	Change in 1500–Foot Sideline Maximum Perceived Noise Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 5 and Thrust of 20,000 Lbs.	98
58	Effect of Axial Position of the Primary Nozzle as Measured by the Change in Maximum Perceived Noise Level in the 1500–Foot Side– line Relative to the Level for Each Nozzle and Pressure Combination in the Caplanar Position.	99
59	Effect of Velocity Ratio of 1500-Foot Sideline Perceived Noise Level for Coplanar Nozzle (Primary Nozzle Pressure Ratio 1.6, Primary Nozzle Temperature 450 ⁰ F, Area Ratio 10, Data Scaled to 20,000 Lbs Thrust).	100
60	Prediction of Overall Sound Power Level Based on Primary Nozzle Data in Figure 40 and Mean Curves of Experimental Variation in the Decrease in Overall Sound Power with Velocity Ratio for Jets with Coplanar Coaxial Nozzles and Fixed Thrust.	102
61	Prediction of 1500–Foot Sideline Perceived Noise Level Based on Primary Nozzle Data in Figures 41 and 42 and Mean Curves of Experimental Variation in the Decrease in 1500–Foot Sideline Maximum Perceived Noise Level for Jets with Coplanar Coaxial Nozzles and Fixed Thrust.	103

Figure		Page
62	Variation of Minimum Sound Power as a Function of Area Ratio and Primary Velocity.	104
63	Variation of Minimum Value of the Maximum Perceived Noise Level on a 1500–Foot Sideline as a Function of Area Ratio and Primary Velocity.	105
64	Prediction of Sound Power Level for a Coaxial Jet Using Secondary Velocity for Reference.	107
65	Prediction of Maximum Perceived Noise Level on a 1500–Foot Side– line for a Coaxial Jet Using Secondary Velocity for Reference.	108
66	Sound Power Spectrum for Coplanar Nozzles, Primary Nozzle Pressure Ratio 1.6, Data Scaled to 20,000 Lbs Thrust.	111
67	Octave Band Sound Pressure Spectrum at Position of Maximum Per- ceived Noise Level on 1500-Foot Sideline. Primary Nozzle Pressure Ratio 1.6, Thrust 20,000 Lbs.	112
68	Directivity Pattern for Primary Nozzle Alone.	114
69	Directivity Pattern for Coplanar Nozzles with Area Ratio 5, Velocity Ratio 0.722.	115
70	Directivity Pattern for Coplanar Nozzles with Area Ratio 10, Velocity Ratio 0.731.	116
71	1500–Foot Sideline Perceived Noise Level for Three Configurations Scaled to 20,000 Lbs Thrust, Primary Nozzle Pressure Ratio 1.6.	117
72	Effect of Jet Thrust on 1500–Foot Sideline Perceived Noise Level for Near–Optimum Jet with 20,000 Lbs Thrust (Nozzle Pressure Ratio 1.6, Temperature 60 ⁰ F, No Secondary Nozzle).	118

Appendix B

B-1	Typical Graphic Record of Total Pressure Profile for a Primary	R3
	Nozzle with Pressure Ratio 1.6.	0-0

Figure		Page
B-2	Effect of Temperature on Boundary Layer at the Exit of Primary Nozzle.	B-3
8-3	Typical Flow Velocity Profiles Through Exit Plane Boundary Layer for the Three Primary Nozzles.	B-4
B-4	Typical Profile of Total Pressure at Exit Plane of Supersonic Nozzle (Pressure Ratio 3.5) Operating at Off–Design Conditions.	B-4
B-5	Total Pressure Profiles at Exit Plane of Secondary Nozzle (Coaxial Nozzles with No Flow in Primary Nozzle).	B6
B-6	Total Pressure and Temperature Profiles at Several Axial Positions for Coplanar Nozzles with Area Ratio 10. Primary Pressure Ratio 2.1, Secondary Pressure Ratio 1.5. (Primary Operating Below Design Pressure Ratio.)	B-7
B7	Total Pressure and Temperature Profiles at Several Axial Positions for Pressure Ratio 3.5 Primary Nozzle Alone. (Operating Slightly Below Design Pressure.)	B-8
B-8	Total Pressure and Temperature Profiles at Exit Plane of Secondary Nozzle (Pressure Ratio 1.89, Area Ratio 10) with Primary Nozzle (Pressure Ratio 2.5, Total Temperature 790°F) Fully Retracted.	B-9
	Appendix C	
C1	Basic Flow Diagram for Complete Data Reduction Computer Program.	C-2
	Appendix D	
D-1	Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.	
	Experimental Run Numbers 102–117	D-2

Experimental Run Numbers 132–143 D–3

D-2

Experimental Run Numbers 119–130

D-1	continued	
	Experimental Run Numbers 145–156	D-3
	Experimental Run Numbers 158–165	D-4
	Experimental Run Numbers 167–174	D-4
	Experimental Run Numbers 176–187	D-5
	Experimental Run Numbers 189–196	D-5
	Experimental Run Numbers 198–205	D-6
	Experimental Run Numbers 215–230	D-6
	Experimental Run Numbers 231–242	D-7
	Experimental Run Numbers 243–254	D-7
	Experimental Run Numbers 255–266	D-8
	Experimental Run Numbers 267–274	D-8
	Experimental Run Numbers 275–282	D-9
	Experimental Run Numbers 283–292	D-9
	Experimental Run Numbers 311–326	D-10
	Experimental Run Numbers 327–338	D-10
	Experimental Run Numbers 339–350	D-11
	Experimental Run Numbers 407–422	D-11
	Experimental Run Numbers 423–434	D-12
	Experimental Run Numbers 435–446	D-12
	Experimental Run Numbers 447–458	D-13
	Experimental Run Numbers 459–466	D-13
	Experimental Run Numbers 467–474	D-14
	Experimental Run Numbers 503–516	D-14
	Experimental Run Numbers 519~530	D-15
	Experimental Run Numbers 531–542	D-15

Figure

語をおいたという

いたのであるかでい

Page

Figure		Page
	Appendix E	
E-1	Variation in Maximum Perceived Noise Level on a 1500–Foot Side– line with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.	
	Experimental Run Numbers 102–117	E-2
	Experimental Run Numbers 119–130	E-2
	Experimental Run Numbers 132–143	E-3
	Experimental Run Numbers 145–156	E-3
	Experimental Run Numbers 158–165	E-4
	Experimental Run Numbers 167–174	E-4
	Experimental Run Numbers 176–187	E-5
	Experimental Run Numbers 189–196	E-5
	Experimental Run Numbers 198–205	E6
	Experimental Pun Numbers 215–230	E6
	Experimental Run Numbers 231–242	E7
	Experimental Run Numbers 243–254	E-7
	Experimental Run Numbers 255–266	E8
	Experimental Run Numbers 267–274	E-8
	Experimental Run Numbers 275–282	E-9
	Experimental Run Numbers 283–292	E-9
	Experimental Run Numbers 311–326	E-10
•	Experimental Run Numbers 327–338	E-10
	Experimental Pun Numbers 339–350	E-11
	Experimental Run Numbers 407–422	E-11
	Experimental Run Numbers 423–434	E-12
	Experimental Run Numbers 435–446	E-12
	Experimental Run Numbers 447458	E-13
	Experimental Run Numbers 459–466	F-13

Figure		Page
E-1	continued	
	Experimental Run Numbers 467–474	E-14
	Experimental Run Numbers 503–516	E-14
	Experimental Run Numbers 519–530	E-15
	Experimental Run Numbers 531–542	E-15

Appendix F

F-1	Model Jet Sound Power Spectra for Primary Nozzles with Pressure Ratio 1.6.	F-2
F-2	Directivity Pattern for Run Number 101 (Pressure Ratio 1.6, Temperature 60°F).	F-3
F-3	Directivity Pattern for Run Number 144 (Pressure Ratio 1,6, Temperature 450°F).	F-4
F-4	Directivity Pattern for Run Number 175 (Pressure Ratio 1.6, Temperature 800 ⁰ F).	F-5
F-5	Model Jet Sound Power Spectra for Primary Nozzles with Pressure Ratio 2.5.	F-6
F-6	Directivity Pattern for Run Number 118 (Pressure Ratio 2,5, Temperature 60°F).	F-7
F-7	Directivity Pattern for Run Number 157 (Pressure Ratio 2.5, Temperature 450 ⁰ F).	F-8
F-8	Directivity Pattern for Run Number 188 (Pressure Ratio 2.5, Temperature 800 ⁰ F).	F-9
F-9	Model Jet Sound Power Spectra for Primary Nozzles with Pressure Ratio 3.5.	F-10
F-10	Directivity Pattern for Run Number 131 (Pressure Ratio 3.5, Temperature 60°F).	F-11

LIST OF FIGURES ... concluded

......

Figure		Page
F-11	Directivity Pattern for Run Number 166 (Pressure Ratio 3.5, Temperature 450 [°] F).	F-12
F-12	Directivity Pattern for Run Number 197 (Pressure Ratio 3.5, Temperature 800 ⁰ F).	F-13
	Appendix G	
G-1	Directivity Index for Overall Sound Pressure Level for a Supersonic Nozzle Operating at Off-Design Pressure Ratios.	G-3
G-2	Octave Band Sound Pressure Spectra at 45 Degrees to Jet Axis for Supersonic Nozzle Operating at Off–Design Pressure Ratios.	G-4
G-3	Effect of Off–Design Operation of the Sound Power Generation by a Supersonic Jet Nozzle.	G-5
G-4	Effect of Off-Design Operation on the Maximum Perceived Noise Level on a 1500–Foot Sideline. Scaled to 20,000 Lbs Thrust.	G-6
G - 5	Octave Band Sound Pressure Level at 110 Degrees, Showing Onset of the Third Anomaly Above 1200–ft/sec from Measured Raw Data.	G-7

LIST OF SYMBOLS

- radius of jet core at downstream station x ($0 \le a \le r_n$) in Equation (1) in feet
- a speed of sound in feet/sec
- a_o speed of sound for standard conditions (1120 feet/sec)
- a* critical speed of sound in flow with velocity Mach Number of }
- A area in square feet

a

- A_i incremental area on sound measurement sphere in square feet
- A_p area of primary nozzle in square feet
- A_s area of secondary nozzle in square feet
- b jet mixing width parameter in Equation (1) in feet
- d typical diameter or dimension of jet in feet
- d_n nozzle exit diameter in feet
- d_s secondary nozzle diameter in feet
- d₂ envelope diameter of mixing nozzles
- DI directivity index in dB
- E mechanical power of jet in watts
- f frequency in Hertz
- f_m octave band center frequency for model jet in Hertz
- f_S octave band center frequency for full-scale jet in Hertz
- F thrust in pounds
- Fe thrust of equivalent jet (defined in Equation (4)) in pounds

xix

Fm	thrust of model jet in pounds
Fn	thrust for a single-nozzle jet in pounds
Fp	thrust of primary jet in pounds
FS	thrust of full-scale jet in pounds
9	acceleration of gravity 32.2 feet/sec
I	Sound intensity in watts/sq. cm.
k	constant in Equation (16)
L	propagation distance to sideline in feet
۲d	sideline distance in feet
^m p	mass flow of primary nozzle in pounds/sec
m _s	moss flow of mass flow of secondary nozzle in pounds/sec
м	flow Mach Number
M _c	eddy convection Mach Number
Mx	jet momentum flow rate across axial station x in pounds
P	sound pressure in dynes/sq. cm.
Pa	measured ambient pressure in inches of mercury
P;	sound pressure in i th incremental area
PNL	perceived noise level in PNdB
PNL	perceived noise level at sideline distance L _d in PNdB
PWL	sound power level in dB re 10^{-13} watts (overall is implied unless otherwise stated)

e, (s

.

PWL	sound power level for model configuration in dB re 10 ⁻¹³ watts
PWLs	sound power level for full-scale configuration in dB re 10 ⁻¹³ watts
PWLOA	overall sound power level in dB re 10^{-13} watts
r	radius in feet
r _b	baseline jet radius as defined in Equation (13)
r _e	equivalent jet radius as defined in Equations (2) and (7)
r _n	jet nozzle radius in feet
r p	primary nozzle radius in feet
rs	secondary nozzle radius in feet
rj	radius of separation boundary between primary and secondary momentum flows for coaxial jets in feet
R	total pressure ratio across nozzle
R _C	critical pressure ratio across nozzle which just produces sonic flow through throat $(R_{c} = 1.89)$
S	Strouhal number based on diameter used in Figure 2
s _x	Strouhal number based on axial distance from nozzle used in Figure 10
SPL	sound pressure level in dB re 0.0002 dynes/sq. cm.
SPLm	sound pressure level measured for model in dB re 0.0002 dynes/sq. cm.
SPLS	sound pressure level scaled from model to full-scale in dB re 0.0002 dynes/sq. cm.
T	jet nozzle flow temperature in degrees Rankine
t _a	measured ambient temperature in degrees Rankine

ŝ

KENSEL DEGENES

. . .

T _t	jet total temperature in degrees Rankine
T _o	temperature for standard conditions (519 ⁰ R)
U	fluctuating turbulent velocity in feet/sec
U	flow velocity in feet/sec
U _c	jet centerline flow velocity in feet/sec
Ue	equivalent jet flow velocity as defined in Equations (2) and (8)
Un	jet nozzle exit velocity in feet/sec
U _p	primary jet nozzle exit velocity (fully expanded flow) in feet/sec
U r p	actual primary nozzle velocity when primary is retracted within secondary nozzle in feet/sec
U _r	jet velocity at radius r in feet/sec
Us	secondary jet nozzle exit velocity (fully expanded flow) in feet/sec
U	freestream velocity in feet/sec
W	total acoustic power in watts
wР	total acoustic power of baseline jet which has same thrust as coaxicl jet and same velocity as its primary, in watts
We	total acoustic power of equivalent jet which has the same thrust and down- stream flow as does the coaxial jet, in watts
W po	total acoustic power from unshrouded portion of primary jet mixing with ambient, in watts
W _{ps}	total acoustic power of primary jet mixing with secondary jet, in watts
W _{so}	total acoustic power of secondary jet mixing with ambient
Wt	total acoustic power of coaxial jet, in watts

のないのないとないというというないです。

÷

x	downstream axial distance from nozzle plane, in feet
×t	axial distance to tip of jet core, in feet
Greek	
a	atmospheric attenuation coefficient in dB/1000 ft
$\alpha_{0}^{\alpha}, \alpha_{1}^{\alpha})$	constants for atmospheric absorption, Equation (35)
β	ratio of area of secondary nozzle to area of primary nozzle (A_s/A_p)
δ	ratio of velocity of secondary jet to velocity of primary jet (U_s/U_p)
∆ _{DP}	decrease in 1500-foot sideline perceived noise level for coplanar coaxial jet (in Figure 61), in dB
Δ _{DW}	decrease in overall sound power level for coplanar coaxial jet (in Figure 60) in dB
Δ _{IP}	increase in 1500-foot sideline perceived noise level for coplanar coaxial jet (in Figure 65) in dB
∆ _{IW}	increase in overall sound power level for coplanar coaxial jet (in Figure 64) in d8
Δι	ratio of core tip length with external flow to core tip length without external flow in Figure 8
ΔU	velocity difference across stated portion of jet
ኤ	sound wavelength for standard conditions
η	nondimensional jet radius parameters; (r + a)/b
η	nondimensional jet radius parameters for coaxial equivalent jet; $(r_i - a)/b$
θ	angle from jet flow axis in degrees

LIST OF SYMBOLS ... concluded

ρ	density in slugs
ρ _e	density of equivalent jet in slugs
۹ _n	density of jet flow in plane of nozzle in slugs
ρ _p	density of primary jet flow (fully expanded flow) in slugs
ρ _s	density of secondary jet flow (fully expanded flow) in slugs

1.0 INTRODUCTION

The noise produced by jet engines comes from three sources – the intake compressor or fan, the internal engine components, and the jet flow of gases from the engine exhaust. This report is concerned solely with the latter source, which determines the jet noise level threshold of the engine after the first two sources have been sufficiently reduced.

Jet noise results from the mixing of the jet flow with the ambient air. This mixing process is turbulent in nature, consisting of velocity and aerodynamic pressure fluctuations in the mixing region which produce the noise audible in the acoustic far field. This far-field noise is reasonably well understood for a single circular jet and accurate prediction methods are available, although many details remain to be clarified.

However, modern jet engines are no longer pure jet engines but utilize the bypass principle. The exhaust from such an engine consists of a central jet of hot gas that has passed through the combustion chamber (termed the primary jet in this report) which is surrounded by the cooler, slower bypass flow (termed the secondary jet).

On some low bypass engines, the bypass flow does not completely surround the primary flow. This practice is being discontinued with increasing ratios, however, so that for all practical purposes the secondary bypass air can be considered to be coaxial with the primary flow and to completely surround it.

The mixing flow pattern from such a coaxial jet is much more complex than that from a single jet with the additional complication that there are many more variables to be considered. Consequently, accurate theoretical prediction of the noise is much more difficult. This difficulty in prediction motivated this experimental study to measure the noise from coaxial flows, covering the full range of physical and aerodynamic parameters to be expected from existing and future byposs jet engines. The parameters of interest include:

- Ratio of the secondary (bypass) mass flow to the primary mass flow.
- Ratio of the secondary flow velocity to the primary flow velocity.
- Ratio of the secondary flow area to the primary flow area.
- Relative axial positions of the secondary and primary nozzles
- Primary nozzle pressure ratio
- Temperature of the primary jet

This report gives the results of an experimental study using model jets in which all the above parameters were examined. The study included a series of more than three hundred separate model jet tests. The data from these tests were then analyzed to determine sound power output and directivity patterns of the model jet configurations.

and were scaled to give the equivalent acoustic output for full sized engines. The fullscale data are presented in terms of the sound pressure spectrum and perceived noise level along two sidelines at 500 feet and 1500 feet from the jet axis.

and the second second

The data were also used to determine the decrease in acoustic output that results from the use of a bypass flow configuration relative to the acoustic output of a single jet nozzle which has the same flow characteristics as the primary jet and the same thrust as that of the bypass flow configuration. The effects of area ratio, velocity ratio, and the relative axial position of the primary and secondary nozzles on this noise reduction were examined. Finally, these noise reduction data were reinterpreted to indicate the flow characteristics required to give a jet with minimum acoustic output for a given amount of gross thrust and the acoustic penalty that results if the actual flow characteristics depart from those optimal design conditions.

Section 2 reviews pertinent aspects of the noise generation of jets from circular and multiple nozzles, and develops a simple analytical model which describes the noise generation of coaxial jets. Section 3 details the experimental test hardware, instrumentation and the data analysis procedures. Section 4 presents the basic experimental data, and Section 5 contains a discussion of the results and the development of prediction techniques applicable to full-scale engines. The principal conclusions are given in Section 6.

Seven (7) appendices are included in this volume. They include a directory of experimental runs, the computer program for data analysis, some samples of measured jet flow profiles and a summary of the principal acoustic data obtained. The detailed flow and acoustic data are contained in Volume II.

2.0 NOISE GENERATION BY COAXIAL JET FLOWS

2.1 Noise Generation of Jets From Single Circular Nozzles

Jet flows generate sound by the turbulent mixing of the gas flow with the ambient atmosphere. This process was first properly documented by Lighthill (References 1 and 2), who recognized that the turbulent shear stresses resulting from the momentum exchange produced pressure fluctuations and a radiated sound field. The process of conversion of jet mechanical energy to acoustical energy is relatively inefficient. However, because of the high mechanical energy of a jet exhaust flow, the resulting noise is of great significance for the design of jet aircraft engines.

In the studies of the properties of jet noise generation, the importance of the velocity of the jet flow is easily identified. The overall sound power of single circular jet flows up to approximately 2000 fps is proportional to the eighth power of the velocity. Although the flow velocity is the most important parameter in the determination of the sound power output of a jet, the physical size and the temperature of the jet are also important. Lighthill showed that the sound output was directly proportional to the jet nozzle area. The flow velocity is a function of temperature; heating the flow will increase the velocity for a given nozzle pressure ratio, thus increasing the noise. Alternatively, if the velocity is maintained constant, the noise will decrease as temperature is increased because of a reduction in density. It is possible to reduce the noise slightly for fixed thrust by heating the gas while maintaining the same exit velocity, as documented by Plumblee et al (References 3 and 4).

The relationship between overall sound power level and jet flow parameters is given in Figure 1. The power spectrum can be normalized on the basis of a strouhal number (fd/U) modified by the ratio (a^*/a_0) of the critical velocity of sound in the flow to that in the surrounding ambient air (References 5 and 6). The results for typical jets are given in Figure 2. The sound field is directional, with a predominant sound radiation occurring at an acute angle to the jet flow direction, as illustrated in Figure 3. This directionality is believed to be the result of the convection of the disturbances within the flow and the refraction of the sound within the flow field, with the approximate relationships indicated in Figure 4 (References 5, 9, 10 and 11).













Figure 4. Variation of the Predominant Propagation Angle for Various Frequencies from Measurements Along the 10 Degree Boundary of a Jet from a J-57 Engine as a Function of a Non-dimensional Axial Frequency Parameter. Data from Howes, References 5 and 12.

2

The velocity profiles in the flow of a constant density circular jet can be approximated with the profile given in Figure 5. The flow momentum rate across any downstream station, which equals the net thrust of the jet, is given by:

$$M_{x} = F_{n} = \pi r_{n}^{2} \rho_{n} U_{n}^{2} = \pi \rho_{n} \left[a^{2} U_{n}^{2} + 2 \int_{a}^{\omega} U_{c}^{2} e^{-\eta^{2}} r dr \right]$$
(1)

where



Figure 5. Comparison of Assumed Velocity Profile for Axisymmetric Constant Density Jet with Mach 0.7. Data from Laurence, Reference 13.

This relationship, together with the assumption of a constant coefficient of momentum exchange throughout the mixing process in the flow, permits an analytical solution of the constant density jet velocities throughout the entire flow, as illustrated in Figure 6. This solution was obtained in terms of the normalized axial distance (x_t/r_n) from the nozzle to the tip of the potential core. The length of the core is a function of Mach number, as shown by the empirical curve in Figure 7. When the jet is surrounded by a moving medium, as for an aircraft in flight, the mixing process is less rapid and the distance from the nozzle to the tip of the nozzle to the tip of tip of

The characteristics of the noise generated along the jet vary with axial position, as do the parameters of the flow. Theoretical examination of Lighthill's basic theory of aerodynamic noise generation resulted in identification of the initial mixing region as the prime noise source for a subsonic jet flow (References 17 and 18). These analyses indicated that there should be a uniform source strength along the length of the initial mixing region, where the potential core still exists and the velocity difference across the mixing region is constant. Further, theory indicates that the magnitude of the noise generation should decrease rapidly, following an x^{-7} law, in fully developed mixing regions downstream of the core tip. These results were obtained by applying similarity arguments to the mixing flow properties (mean velocity profile and turbulence intensity).

Experimental measurements shown in Figure 9 appear to validate theory in the initial mixing region. However, the noise generation does not fall as rapidly as predicted by theory downstream of the core tip, perhaps because the turbulent eddies developed in the initial mixing region decay less rapidly as they are convected downstream than do the mean flow velocities. This possible explanation is reinforced by observing the nondimensional source noise spectra in Figure 10. In the initial mixing region, the frequency of maximum amplitude varies inversely with axial distance; in the downstream region, the spectrum appears almost independent of axial distance.

The basic similarity relationships between the total sound power spectrum and mean jet flow parameters, and between the local sound power spectra and the local flow relationships, provide a powerful tool for estimating the noise generation characteristics of jets with complex nozzles or non-standard initial flow profiles. The similarity laws between flow, turbulence and noise can enable solutions to the noise of a complex flow by analogy to the well known circular jet, once the velocity profiles of the complex flow are defined. This approach was used for the estimation of the noise characteristics of jet flows from mixing nozzles which have multiple tubes or corrugations (Reference 5) and is applied in this report to jet flows from coexial nozzles, as discussed in the following two subsections.



Figure 6. Examples of Flow for Constant Density Axisymmetric Single-Nozzle Jets. (From Reference 5.)

-9-



Figure 7. Variation in Core Length (x₁) for Constant Density Axisymmetric Jet as a Function of Mach Number. (From Reference 5.)



Figure 8. Calculated Increase in the Ratio of the Length of the Jet Core with Non-Zero External Velocity (U_m) to its Length with Zero External Velocity for Constant Density Jets. (From Reference 5.)





-11-



Figure 10. Jet Source Normalized Sound Power Spectra. Data from J-57, References 5 and 12.
2.2 Noise Prediction for Jet Flows from Mixing Nozzles

The possibility of modifying the noise of a single jet by changing the nozzle shape was first discovered by Westley and Lilley (Reference 19), when they observed a reduction in total noise as the result of inserting tooth-like projections into the nozzle. Initial full-scale application of this discovery was made by Greatrex of Rolls Royce and Callaghan, et al, of NASA (Lewis), (References 20 – 24). The concept was reduced to practice in this country in the 1950's with the use of 21- and 9-tube nozzle suppressors on the Boaing 707 series aircraft, corrugated nozzles on the Convair 880 aircraft and corrugated nozzles with ejectors on the Douglas DC-8 series aircraft. More recent developments include the multiple chutes proposed by General Electric for the U.S. supersonic transport engine.

These devices all act to subdivide the single jet into several smaller jets. If the smaller jets are spaced sufficiently far apart so that each jet mixes as a normal single jet, the length of the mixing region is reduced in proportion to the ratio of the diameter of one of the small jets to the diameter of a baseline circular single nozzle which has the same area as the total of the small jets. Similarly, the total acoustic power of each of the small jets relative to the baseline circular jet is reduced in proportion to the nozzle area ratio, with its characteristic frequency shifted up relative to the circular nozzle in inverse proportion to the diameter ratio. However, the acoustic power of the sum of all the smaller nozzles equals that of the baseline circular jet since the sum of the areas of the smaller nozzles equals that of the baseline line nozzle.

The power spectrum levels for a variety of sizes of non-interfering tube nozzles for a 10,000 lb thrust jet is illustrated in Figure 11. Also included in the figure is the power spectrum for two cases where the exit velocity was reduced by lowering temperature while retaining total pressure and the baseline circular nozzle.

In general, it is impractical to space the smaller jets sufficiently far apart to enable them to mix namally without interference. Consequently, for most practical configurations, only the peripheral small jets mix rapidly, as would be expected for independent small jets. The inner jets tend to recombine, with the degree of recombination critically dependent upon the amount of ambient air which can be induced into the central portion of the flow.

-13-





-14-

Power Spectrum Level, d8 re 10-13

zH/ettow

A simplified sketch of the flow from a circular ring of tube nozzles is illustrated in Figure 12. Reference 5 gives approximate solutions for the flow from idealized nozzles of this type and compares the solution with data taken by Lawrence and Benninghoff (Reference 25), as shown in Figure 13.



Figure 12. Sketch of Two Nozzles in the Plane Containing the Axis of a Simplified Axisymmetric Peripheral Tube Nozzle Illustrating Development of Final Mixed Jet and the Mixing of the Individual Jets. (From Reference 5.)

A method for computing the noise for ideal tube nozzles which were equally distributed within a circular envelope was developed in Reference 5. The method consisted of computing the sum of the total acoustic power for the peripheral small jets and the acoustic power for the slower, larger diameter, combined mixed jet. The velocity and size of the slower combined mixed jet were related to the nozzle exit velocity and the diameter (d_2) of the circle which envelopes the tube array.





Figure 14 shows two examples of a comparison of the results from the Reference 5 method with data for two configurations of 12-lobe nozzles. The low frequency peak in the sound power spectrum results from the combined jet and is lower in magnitude for the configuration with the larger envelope diameter. The high frequency peak in the calculated sound power spectrum results from the flow from the 12 lobes; its characteristic frequency is associated with the typical peripheral dimension of the lobes.



Figure 14. Comparison of Calculated Power Spectra with Power Spectra Derived from Measured Power Reduction for Two 12-Lobe Nozzles with J-57 Engine. (From References 5 and 26.)



An example of the variation in total acoustic power for a series of idealized circular arrays of uniformly distributed tube nozzles as a function of tube size, and the ratio of the envelope diameter to the diameter of a baseline single circular nozzle which has the same area as the sum of the areas of the tube nozzles, is shown in Figure 15. This method has not been generalized to the prediction of perceived noise level along the sideline, although recently it has been applied successfully in an individual case for the proposed supersonic transport (SST) engine, utilizing the appropriate directivity functions for the jet noise. Such generalization is required to determine optimum values for both the characteristic diameter of the tubes or corrugations and the envelope aiameter for application to perceived noise level for aircraft operations near airports. It is expected that the optimum values for this case will differ from those indicated in Figure 15 which apply to overall power level.



Ratio of Envelope Diameter to Diameter of Baseline Single Nozzle (d_2/d_n)

Figure 15. Computed Variation in Total Acoustic Power for Idealized Mixing Nozzle Relative to That of Baseline Single Circular Nozzle for T/T = 3, as a Function of the Ratio of the Envelope Diameter of the Mixing Nozzle to the Diameter of the Baseline Single Nozzle, with Tube Element Diameter as a Parameter. (From Reference 5.)

2.3 Noise Prediction for Jet Flows from Coplanar Coaxial Nozzles

The flow from a low bypass ratio coaxial jet nozzle is illustrated in Figure 16. It is characterized by two basic regions. In the upstream region, the secondary flow shrouds the primary flow. In the downstream region, the two flows have merged and behave as a single jet in accordance with the similarity parameters discussed in Section 2.1. The axial distance to the station separating these two regions is a function of both the area ratio (β) of the secondary to primary nozzles and their velocity ratio ($U / U = \delta$). When the secondary flow is large, relative to the primary, the flows will ^s ^p generally merge downstream of the tip of the primary core. However, as in Figure 16, when the secondary flow is small relative to the primary, the merger will occur upstream of the tip of the primary and only a small amount of the primary jet flow will be shrouded.



Figure 16. Approximate Velocity Profiles at Various Downstream Stations for Constant Density Annular Coplanar Jet with $U_s/U_p = .5$ and $r_s/r_p = 1.5$.

Greatrex (Reference 27) observed that the noise of a coaxial jet should be composed of three parts:

- Noise of the secondary jet mixing with the atmosphere (shrouded region)
- Noise of the primary jet mixing into the secondary (shrouded region)
- Noise of the large combined jet

Similar concepts were advanced in Reference 5 and approximate estimates were made of the reduction in acoustic power which could be obtained as a function of bypass ratio, for a velocity ratio of .5, considering only the noise produced by the combined jet.

In order to develop quantitative estimates of these three noise sources, it is necessary to define the extent of the shrouded region and the parameters of the combined jet. The velocity profiles in the combined jet could be duplicated by a portion of the flow from a single-nozzle jet which has the appropriate nozzle velocity and radius. Such a single-nozzle jet may be called an "equivalent jet," in both an aerodynamic and an acoustic sense.

The thrust of the equivalent jet must equal the thrust of the actual coaxial jet:

$$F_{e} = \pi \rho_{e} U_{e}^{2} r_{e}^{2} = \pi \rho_{p} U_{p}^{2} r_{p}^{2} + \pi \rho_{s} U_{s}^{2} \left(r_{s}^{2} - r_{p}^{2} \right)$$
(2)

where the subscripts e, p and s denote equivalent, primary and secondary, respectively.

For a constant density jet, which will be considered here, this equality gives:

$$U_{e}^{2}r_{e}^{2} = U_{p}^{2}r_{p}^{2} + U_{s}^{2}\left(r_{s}^{2} - r_{p}^{2}\right)$$
(3)

At the beginning of the combined jet from the coaxial nozzle, it is postulated that the flows have merged and obey the similarity profile for the single jet shown in Figure 5. Hence, at the most upstream or initial station at which this condition is fulfilled, the thrust for the combined jet is given by:

$$F_{e} = \pi \rho U_{e}^{2} r_{e}^{2} = \pi \rho \left[a^{2} U_{p}^{2} + 2 \int_{a}^{\infty} U_{c}^{2} e^{-\eta^{2}} r dr \right]$$
(4)

If the flows merge upstream of the primary core tip, the first term in the brackets is non-zero and the equivalent jet velocity equals the primary velocity. If the merger occurs downstream of the primary core tip, the first term in the brackets is zero and the equivalent jet velocity equals the jet centerline velocity. These two cases are illustrated in Figure 17.



Figure 17. Sketch of Equivalent Jet Flow Profile for Two Cases.

The radius of the separation boundary between the inner momentum flow due to the primary, and the outer momentum flow due to the secondary, may be defined for any axial station. At the initial station, however, where the flows merge and initially obey the similarity profile for the combined jet, the velocity associated with the radius (r_i) is approximately equal to the secondary flow velocity. Hence, at the initial station:

$$U_{r_1} = U_s = U_c e^{-\frac{\eta_1^2}{2}}$$
 (5)

For case (b) of Figure 17, the momentum of the outer flow at Station 1 may be written as:

$$\pi \rho U_{s}^{2} \left(r_{s}^{2} - r_{p}^{2} \right) = 2 \pi \rho \int_{r_{1}}^{\infty} U_{c}^{2} e^{-(r/b_{1})^{2}} r dr$$
$$= \pi \rho b_{1}^{2} U_{c}^{2} e^{-(r_{1}/b_{1})^{2}}$$
(6)

Substituting from Equation (5) and recognizing that the tip of the core of the equivalent jet $r_e = b_1$:

$$r_e^2 = r_s^2 - r_p^2$$
, or $\left(\frac{r_e}{r_p}\right)^2 = \beta$ (7)

where β is the area ratio of the secondary to primary jets.

From Equations (3) and (7), the velocity is given by:

$$\left(\frac{U_{e}}{U_{p}}\right)^{2} = \left(\frac{U_{s}}{U_{p}}\right)^{2} + \frac{r^{2}}{r^{2}_{s} - r^{2}_{p}}, \qquad (6)$$

or
$$\left(\frac{U_{e}}{U_{p}}\right)^{2} = \delta^{2} + \frac{1}{\beta}$$

where δ is the velocity ratio, U_{s}/U_{p} .

For case (a) of Figure 17, $U_e = U_p$ and the radius of the equivalent jet is given by:

$$\mathbf{r}_{e}^{2} = \mathbf{r}_{p}^{2} + \left(\frac{U}{s}\right)^{2} \left(\mathbf{r}_{s}^{2} - \mathbf{r}_{p}^{2}\right), \text{ or } \left(\frac{r}{e}\right)^{2} = 1 + \delta^{2} \beta \qquad (9)$$

This case of incomplete shrouding occurs whenever the combination of area ratio and velocity ratio are such that:

$$\left[\left(\frac{r_{s}}{r_{p}}\right)^{2}-1\right]\left[1-\left(\frac{U_{s}}{U_{p}}\right)^{2}\right] \leq 1, \text{ or } \beta (1-\delta^{2}) \leq 1 \quad (10)$$

The effect of this relationship is shown in Figure 18.



Figure 18. Calculated Regions of Complete and Incomplete Shrouding of the Primary Core as a Function of Area and Velocity Ratios.

:

The radius and velocity ratios for the equivalent jet from Equations (7), (8) and (9) are plotted in Figure 19 as a function of the area and velocity ratios of the coaxial jet. When the nozzle area ratio is less than 1, the equivalent jet velocity equals the primary velocity for all values of the secondary velocity. The minimum area ratio required to ensure complete shrouding of the primary jet core increases from 1 at low values of the velocity ratio to approximately 2 for $\delta = 0.7$ and to over 5 for $\delta = 0.9$. When the area ratio is less than the minimum required for complete shrouding, the equivalent jet radius is indicated by the dashed lines in the figure.

This information can be used to construct an approximate model of the entire flow-field and estimate the resulting acoustic power spectrum utilizing the general methods of Reference 5. However, the purpose of these experiments is to develop generalized curves of sideline perceived noise level, and the detailed computation of power spectra is not sufficient for this purpose. Rather, it is more relevant to examine the gross trends in overall acoustic power as a function of coaxial jet parameters to obtain guidance for understanding the experimental variations of perceived noise level.

For this discussion, the noise estimates for the coaxial jet flows will be compared to the noise of a baseline single-nozzle circular jet which has the same velocity as the primary nozzle of the coaxial jet and has the area required to make its thrust equal to that of the coaxial jet.

The acoustical power estimates are based on the Lowson and Pao relationships (References 28 and 29) which consider the total acoustic power (W) proportionality as:

$$W \sim \rho A u^4 M_c^4 \tag{11}$$

where u is the fluctuating turbulent velocity which is proportional to the velocity difference (ΔU) across a mixing layer, and M_c is the eddy convection Mach Number.

For a single jet, both u and M_c are proportional to the flow velocity U, so that Equation (11) is equal to the simple pAU^8 Lighthill relationship. For shrouded flows, however, u is proportional to the velocity difference, whereas M_c should be proportional to a typical value of the absolute velocity in the mixing flow. For this analysis, M_c will be considered proportional to the maximum velocity in the mixing layer.

Dashed lines for case of incomplete shrouding of core at indicated values of U_s / U_p Radius Ratio r_e/r Primary Shrouded to or Beyond 3 Core Tip $= \beta^{1/2}$ r_/r e p r<u>s</u> rp - 1 9 β 2 3 2 0 4 Ó 8 10 12 14 16 18 20 Area Ratio (β) 1.0 Velocity Ratio U Q .9 Velocity Ratio U_VU .8 , 8 7 .7 6 .6 5 . 5 3 0 2 6 12 8 10 4 14 16 18 20 Area Ratio (B)

Figure 19. Relationship Between Radius and Velocity of Equivalent Jet and the Parameters of a Constant Density Coplanar Coaxial Jet.

The acoustic power for the three principal source regions can be approximated by:

Equivalent jet:
$$W_{e} \sim r_{e}^{2} \cup_{e}^{8}$$

Primary mixing $W_{ps} \sim r_{p}^{2} (U_{p} - U_{s})^{4} \cup_{p}^{4} \Delta L$ (12)
with secondary: $W_{ps} \sim r_{p}^{2} \cup_{p}^{8} - U_{s}^{2} \cup_{p}^{8}$
Secondary mixing $W_{so} \sim r_{s}^{2} \cup_{s}^{8}$

where ΔL is the factor accounting for the lengthening of the primary jet by the presence of an external flow, as shown in Figure 8.

The size of the baseline jet which has the same thrust as the coaxial jet and the same velocity as its primary nozzle flow can be determined from the momentum equation:

$$U_{p}^{2}r_{b}^{2} = U_{p}^{2}r_{p}^{2} + U_{s}^{2}\left(r_{s}^{2} - r_{p}^{2}\right), \text{ or } \left(\frac{r_{b}}{r_{p}}\right)^{2} = 1 + \delta^{2}\beta \qquad (13)$$

Its acoustic power is proportional to:

$$W_{b} \sim r_{b}^{2} U_{p}^{\theta} = r_{p}^{2} U_{p}^{\theta} (1 + \delta^{2}\beta)$$
 (14)

The ratio of the power generated by the downstream flow of the equivalent jet (anehalf its total power) to that of the baseline jet is from Equations (7), (8), (12) and (14):

$$\frac{W_{e}}{2W_{b}} = \frac{r^{2} U^{8}_{e}}{2r^{2} U^{8}_{p} (1 + \delta^{2}\beta)} = 1/2 \left(\frac{\delta^{2}\beta + 1}{\beta}\right)^{3}$$
(15)

Similarly, the power ratio for the primary mixing with the secondary is:

$$\frac{W_{ps}}{kW_{b}} = \frac{(U_{p} - U_{s})^{4} U_{p}^{4} \Delta L}{k U_{p}^{8} (1 + \delta^{2} \beta)} = \frac{(1 - \delta)^{4} \Delta L}{k (1 + \delta^{2} \beta)}$$
(16)

and for the secondary mixing with the ambient:

$$\frac{W_{so}}{2W_{b}} = \frac{r_{s}^{2} U_{s}^{8}}{2r_{p}^{2} U_{p}^{8} (1 + \delta^{2}\beta)} = \frac{(\beta + 1) \delta^{8}}{2(1 + \delta^{2}\beta)}$$
(17)

The constant k in Equation (16) has a value between 1 and 2, with the lower value appropriate when the significant noise-generating region of the primary flow is shrouded. The upper value should apply when the primary flow is shrouded only to the tip of its core. Between these two cases, intermediate values should apply. For the estimates in this report, k = 2 was used for the area ratio of 1, and k = 1 was used for the area ratio 2, 5 and 10.

A first approximation of the noise from the unshrouded portion of the primary can be obtained from:

$$\frac{W_{po}}{2W_{b}} = 1/2 \left[1 - \left(\frac{(\beta+1)(1-\delta^{2})}{2-\delta^{2}} \right)^{1/2} \right] .$$
(18)

Figure 20 summarizes the estimates of the reduction in overall sound power level for a coaxial jet with an area ratio of 10 using Equations (15), (16) and (17). A maximum reduction occurs at a velocity ratio of approximately .55. Below that velocity, the noise output of the coaxial jet is dominated by the noise generated by the primary jet mixing into the secondary jet. Above a velocity ratio of .55, the noise output is dominated by the noise of the equivalent jet with a lesser contribution by the noise generated by the noise generated by the noise of the equivalent jet with a lesser contribution by the noise generated by the secondary mixing with the ambient.

Similar calculations have been made for area ratios of 1, 2 and 5, with the results summarized in Figure 21. The equivalent jet for an area ratio of 1 has a velocity equal to the primary velocity for all values of velocity ratio, as seen in Figures 18 and 19. Therefore, the moise of the equivalent jet is almost constant and the reduction is primarily controlled by the amount of the primary core which is shrouded.





Incomplete shrouding also affects the results for area ratic 2 for velocity ratios of .7 and above. However, for the remainder of the curves, the equivalent jet velocity is less than the primary and the maximum noise reduction occurs at a velocity ratio in the vicinity of .5. These simple theoretical estimates of sound power reduction are shown in Section 5 to be in general agreement with the experimental data, particularly for the higher values of area ratio.



Figure 21. Estimated Reduction in Overall Sound Power Level with Velocity Ratio for Coplanar Coaxiel Jets of Fixed Thrust and Various Area Ratios. Reduction is Measured Relative to a Single-Nozzle Jet Which has Equal Thrust and a Velocity Equal to Up.

:.

Intentionally Left Blank

-,:.)-

3.0 EXPERIMENTAL APPARATUS AND DATA ANALYSIS PROCEDURES

The experimental program was conducted using model jets located in an anechoic room in order to provide free-field conditions. In this section, details are given of the experimental rig, supporting measurement equipment and the analytical basis for computer data analysis.

3.1 Jet Flow Parameters and Design of the Experiment

The following jet flow and nozzle configuration parameters must be considered in the design of the experimental program for determining the noise characteristics of coaxial jets:

- Secondary and primary flow velocities
- Temperature (density) of the flows
- Relative size of the secondary and primary nozzles
- Relative axial position of the secondary and primary nozzles
- Thrust

A range of values must be chosen for each of these which covers the range to be expected from existing and projected jet engines (Reference 30). In practice, an engine designer works with additional factors which are related to those listed above, such as pressure ratios, bypass ratios and Mach numbers, so these also must be considered. On this basis, the following variation in jet engine parameters was chosen for investigation in this experimental model study:

Thrust

The thrust values of jet engines vary from less than 10,000 lbs for the smaller civil aircraft up to 30,000 to 40,000 lbs for large long-range commercial transport aircraft. Engines with even greater thrusts are under development. Four engine thrusts of 10,000 lbs, 20,000 lbs, 40,000 lbs and 80,000 lbs were therefore used in this program as the equivalent full-scale thrusts.

Primary Nozzles

Three different configurations were chosen with pressure ratios of 1.6, 2.5 and 3.5. The first is subsonic, whereas the latter two are supersonic. The pressure ratio of 3.5 is higher than that currently used in jet engines or proposed for future engines, but was included to give a significant variation in primary nozzle exit velocities.

Preceding page blank

-31-

Secondary Nozzles

Present indications are that engines with bypass ratios up to ten-to-one are being considered. In order to cover this range, the secondary nozzles were chosen to give secondary to primary area ratios of approximately unity, two, five and ten. All the secondary nozzles were simple convergent nozzles to be run at subsonic or just sonic conditions.

Flow Temperature

On existing bypass jets, the bypass air is not used for combustion. For the purposes of this program, the secondary flow was therefore assumed to be at ambient temperature. For the primary jet, three different temperatures were used. These were ambient, 450°F and 800°F. The latter temperature was a limitation provided by the experimental facility and is slightly less than conditions necessary to cover the normal operating range of present jet engines.

Flow Velocity

At design conditions, the calculated ideal primary flow velocities were as shown in Table 1.

Table 1

Calculated Ideal Primary Flow Velocities for Various Temperatures and Pressure Ratios

	Total Temperature				
Pressure Ratio	60 ⁰ F	450 ⁰ F	800 ⁰ F		
1.6 2.5 3.5	889 ft/sec 1201 ft/sec 1375 ft/sec	1174 ft/sec 1590 ft/sec 1818 ft/sec	1437 ft/sec 1943 ft/sec 2223 ft/sec		

Axial Position

Most existing bypass engines have a primary nozzle that protrudes beyond the secondary (bypass) nozzle, but the tendency is to reduce the axial distance between the two nozzles in order to facilitate the incorporation of silencing devices. The experimental facility was therefore designed so that the secondary nozzle could be moved axially relative to the primary nozzle over a range of plus or minus eleven primary nozzle diameters. In the experiments, axial displacements of 11, 5.5, 0, -5.5 and -11 primary nozzle diameters were used.

er men and an and a stand and a standard and a stan

The complete set of runs utilizing all possible combinations of the above parameters results in nearly five hundred (500) runs. In the actual experiments, approximately two-thirds of these possible runs were undertaken. Table 2 summarizes the runs for which experimental data was obtained and analyzed. Greater detail may be found in Appendix A.

Table 2

Summary of Runs Completed

	Primary Temperature			
	Ambient	450 [°] F	800 [°] F	
Primary Fully Extended	All	All	1.6 Primaries 3.5 Only	
Primary Half Extended	All	All	None	
Coplanar	All	All	All	
Primary Half Retracted	All	None	None	
Primary Fully Retracted	All	None	None	

= All Combinations

3.2 Test Hardware

3.2.1 Jet Rig.

Figures 22 and 23 show the configuration of the jet rig. It consisted of two coaxial pipes through which two separate air supplies were directed to demountable coaxial nozzles. Each air supply pipe contained flow straighteners at the inlet end, composed of bundles of thin wall tubes laid axially within the supply pipes. The construction of the assembly allowed a limited amount of longitudinal movement of the primary (inner) supply pipe within the secondary (outer) supply pipe. By this means, the relative positions of the primary and secondary nozzles could be adjusted. A means of locking the inlet ends of the coaxial pipes provided the longitudinal constraint for the primary supply pipe and nozzle. By supporting the forward end of the primary supply pipe on a three-point axial suspension system, relative longitudinal movement could occur between the coaxial pipes, as required to allow for differential expansion. Additional longitudinal adjustment of the relative positions of the primary and secondary nozzles was attained by the use of extension tubes on the primary nozzle supply.





.



The complete coaxial system was supplied with air through flexible pipes and was suspended in a framework by wire ropes. The ends of the wire ropes were connected via universal joints to runners extending the length of the framework. The thrust of the operating jets was applied to a thrust gage mounted on the framework at the rear of the jet assembly. To ensure that no gap existed between the jet assembly and the thrust gage, and to bring the latter into the linear part of its operating range, a pre-load was applied to the gage. This was accomplished by securing a known weight to a cable running over a pulley attached to the rear of the jet rig support structure. The entire framework was covered with fiber glass to prevent acoustic reflections.

3.2.2 Jet Nozzles.

Three primary nozzles and four secondary nozzles, illustrated in Figures 24 and 25, were fabricated for these experiments. Figure 24 also shows the primary nozzle extension lengths. The primary nozzles were designed for pressure ratios of 1.6, 2.5 and 3.5. The internal exit diameters of each nozzle were as follows:

0	Pressure ratio 1,6:	Exit diameter 0,7495 inches
8	Pressure ratio 2,5:	Exit diameter 0.7457 inches
•	Pressure ratio 3.5:	Exit diameter 0,7533 inches

The external diameter of the primary nozzles was one inch, except for the last halfinch near the nozzle exit which tapered from one inch diameter to 0.8 inch at the exit plane. These nozzles were constructed of stainless steel to withstand the high primary flow temperatures.

The four secondary nozzles had internal exit diameters as follows:

•	Nozzle No. 1:	1.288 inches
•	Nozzle No. 2:	1,491 inches
•	Nozzle No. 3:	1.955 inches
•	Nozzle No. 4:	2.567 inches

These were manufactured from resin-bonded fiber glass. The actual ratio of the secondary nozzle flow area to primary nozzle exit area can be calculated from the above dimensions. In practice, the formation of boundary layers resulted in effective flow areas somewhat less than the theoretical. The effective flow areas were calculated from a knowledge of the experimentally determined mass flows through the nozzles and their pressure ratios for a velocity ratio of one.





Figure 25. The Four Secondary Nozzles.

The nominal and effective area ratios for the four secondary nozzles are:

			(For Velocity Ratio of One)				
		Nominal	Effective Coplanar	1/2 Extended	Fully Extended		
÷	Nozzle No. 1:	1	1.21	1.00	1.00		
•	Nozzle No, 2:	2	2,39	2.02	1.95		
•	Nozzle No. 3:	5	5.23	4.98	4.99		
٠	Nozzle No. 4:	10	10.50	9.76	10.00		

These values apply only for the condition when the secondary flow velocity approximated that of the primary. When the velocities were different, there were small variations in the effective area. For all conditions, however, the effective areas of the primary nozzles were within one percent of 0.00287 sq. ft., indicating a nozzle coefficient of approximately 94 percent.

3.2.3 Air Supply.

The air supplies for the primary and secondary nozzles were obtained from three 2-stage reciprocating air compressors with combined mass flow rates of 275 lbs per minute and with delivery pressures in the range 125 psig to 255 psig. A booster compressor was also available to increase the pressure from the compressors, with the lower delivery pressure up to a maximum of 300 psig.

The primary air supply was heated by a gas-fired heat exchanger capable of supplying 550,000 Btu per hour. With the air flows involved in this program and the length of line to the jet rig, the maximum static temperature attainable at the primary nozzles was approximately 900°F; this was attainable only for certain air flow rates. As a result, a primary flow temperature of 800°F was used for the majority of the hottest runs. Provision also was made in the air supply system for the mixing of cold air with the hot primary flow in order to achieve precise temperature control.

The mass flow of air to both primary and secondary nozzles was determined from the pressure drop across the orifice, which was measured by a mercury or red-oil manometer. The supply line air pressures upstream of the flow measuring orifices and in the settling chambers of the primary and secondary nozzles were measured by Ashcroft dial pressure gages, Models 1082A or 1279A, which have an accuracy of better than 0.5 percent. The flow temperatures upstream of the flow orifice and in the nozzle settling chambers were measured with thermocouple probes and recorded on a standard temperature recorder.

3.2.4 Jet Traversing Mechanism.

To investigate the jet velocity and temperature profiles (particularly for the purpose of checking the nozzle flows), three probes were traversed across the jet flow (Appendix B). To accomplish these traverses, the probes were mounted on a carriage which was driven across the jet axis by a leadscrew. The leadscrew, in turn, was driven by a variable-speed electric motor via a speed reducer. The shaft of a multiturn potentiometer was coupled to the leadscrew, the potentiometer being supplied with direct current from a stabilized power supply. When the wiper of the potentiometer was connected to a chart recorder, the position of the probes across the jet was indicated as a chart pen deflection.

3.2.5 Anechoic Room.

The anechoic room used for the experiments was cylindrical in shape, approximately 30 feet in diameter and 21 feet high. To minimize acoustical background noise levels, the walls of the room were of high transmission loss construction. The lower section of the walls was constructed of cinder blocks, the courses filled with reinforced concrete and plastered on both sides, to a total wall thickness of 9 inches. The upper wall section was of staggered stud, wood frame construction, composed of the following layers (from interior to exterior wall surfaces):

- 1/2-inch gypsum wall board
- 1/2-inch wood fiber sound deadening board
- An air space of 5-1/2 inches
- 1-inch particle board mounted on the outer studs

Access to the room was via triple, solid-core doors separated by air spaces.

The basic anechoic treatment of the room was a curtain wall of 4-inch thick fiber glass spaced 6 to 10 inches from the interior wall surface. The floor of the room was treated in a similar fashion, and the suspended ceiling also consisted of a 4-inch layer of fiber glass. The basic acoustical treatment was supplemented by additional low frequency absorption in the form of wedges that were located in patches around the wallfloor perimeter.

The model jets used for the experiments were exhausted through a muffler to the external atmosphere. The muffler was designed to preserve the acoustical integrity of the anechoic room and consisted of a bent duct approximately 12 feet long and 4 feet square. The duct was lined with 4 inches of fiber glass spaced 4 inches from the interior walls. A fiber glass splitter vane divided the duct cross-section in half, over a longitudinal distance of 4 feet, to increase the high frequency absorption of the duct.

-40--

The suitability of the room for acoustic measurements such as were undertaken in this program is illustrated in Figure 26. These data show that the octave band acoustic levels in the room follow the inverse square law over the frequency band width and angular positions used for the measurement program.

3.3 Instrumentation

3.3.1 Acoustical Data Acquisition System.

The noise produced by the jets was measured by the system shown in Figure 27. This system consisted of a microphone mounted on a boom which slowly traversed a segment of a circle of radius 10.67 feet centered on the secondary nozzle. The segment ranged between 15 and 115 degrees from the jet axis. The microphone was a Bruel and Kjaer (B & K) half-inch plane wave condenser microphone, Type 4133, aligned to present the diaphragm at normal incidence to the noise from the jet. The microphone output was amplified by a B & K amplifier, Type 2603, and the amplified output fed to a B & K level recorder, Type 2305, and an Ampex two-track tape recorder. The system was calibrated using a B & K pistonphone, Type 4220. The system gain was increased as necessary to bring the signal level up toward the desired recording level (i.e., about 15 dB below the peak overload point). These gains were carefully noted for each run. The tape recordings were made at a tape speed of 15 inches per second to allow frequencies up to and including the 31.5 kHz octave band to be recorded.

The microphone traverse mechanism contained a microswitch trip mechanism which was actuated at 5-degree intervals (starting at 15 degrees from the jet axis). These 5degree invervals were marked on the level recorder trace, using the event marker, and used to interrupt a 1 kHz tone which was then recorded on the second channel of the tape recorder. This provided a synchronized index signal of microphone position on the tape recording.

3.3.2 Acoustical Data Analysis System.

The on-line recordings of the jet noise were analyzed at a later date by the system shown in Figure 28. The first channel of the tape recorder contained the recorded jet noise. This was fed, via an attenuator and amplifier to a 8 & K level recorder and a Wyls analog-to-digital octove band analyzer. The 1000 Hz signal on the second channel of the tape recorder indicated the position of the microphone-traversing boom at 5-degree angular increments. This signal operated through a simple relay device to trigger the octave band digital analyzer and to indicate the angular positions on the event marker of the level recorder.

-41-



Distance in Feet

Figure 26. Inverse Square Law Measurements in the Anechoic Room Using Full Octove Bands of Random Noise. The broken lines indicate a slope of 6 dB per doubling of distance. The hatched area indicates the level variation found over a 120-degree sweep of the microphone boom.

Brief Cake bertown



and the second sec

.....

i fizi na kata polizi ni ini ni polizi na na kata na kata na kata na kata na kata ni ini na kata na poznora. Na na kata na k

0

-43-



-0

Figure 28. Black Diagram of Acoustic Data Analysis System.

:

.

The recorded signal was fed from the tape recorder through an attenuator and a booster amplifier to the digital analyzer which contains a set of parallel 8 & K octave band filters with center frequencies ranging from 63 Hz to 31.5 kHz. The filter outputs were then converted to dc levels, digitized and punched sequentially on a standard 8-hole ASCII paper tape punch as a 3-digit number. The value punched is proportional to the input voltage, i.e., it is linear, not logarithmic. Additional punching included the overall level, a reference level derived from a separate input, and a run number which was set by thumb switches on the analyzer. For this experiment, the reference signal and the lowest three octave bands were not used. Suitable line feed, carriage return and spacing characters were included on the paper tape so that the data could be printed in a legible tabular form by a teletype machine. The paper tape punch could be triggered either manually or by a separate triggering input.

Calibration of the system was undertaken using the 124 dB acoustic calibration signal recorded directly on the data magnetic tape prior to each run. The digital analyzer was adjusted so that 124 dB gave a maximum paper tape punch count of 999. A preliminary analysis of the overall sound pressure level was then made using only the level recorder with the same gain setting as was used when the recordings were made. The gain was then increased so that the maximum octave band level would register near 999 counts, obtaining the maximum dynamic range of the digital analyzer. For compatibility, the overall output channel was attenuated by 5 dB to insure that it would not overflow the counter. These gains and the original recorded 124 dB calibration signal were utilized to calculate a calibration factor in millivolts/dyne/cm for use in computer decoding of the digital tape.

Systems such as that described above for data acquisition and analysis do not have a completely flat response over the frequency range of interest for this project. Therefore careful calibrations were made to determine the octave band corrections that had to be applied to the final octave band analysis produced by the paper tape punch. These calibrations indicated that the following corrections had to be added:

Octave Band Center Frequency (kHz)	0.5	1	2	4	8	16	31.5	Overall
Correction (dB)	0	0.5	1.5	1.0	2.0	3.0	4.5	6.0

-45-

W . . . WARKER AREA SO

The overall correction factor accounts for the 5 dB added in the digital analysis as well as an approximate correction for the octave band correction factors. This latter correction obviously will be a function of spectrum shape and therefore is not an exact correction. Hence, whenever calculations were made on the octave band figures (as in the computer program to be described later), the overall value was calculated from the sum of the octave band levels.

3.3.3 Thrust Measurement.

The thrust of the model jets was measured by suspending the whole jet assembly (as shown in Figure 22) from a supporting frame so that it acted as a parallel pendulum. The assembly was then made to react against a force gage of the proving-ring type. This gage was manufactured by Wiancko Engineering Company (thrust gage, Type F1021). It utilized a differential transformer excited by a carrier signal to measure thrusts in the range zero to 100 lbs. The gage was pretensioned by a constant 5-lb load to eliminate initial non-linearity at the low end of the thrust range. Calibration was achieved by adding extra known weights to the preload system.

3.3.4 Jet Profile Measurements.

To investigate the velocity and temperature profiles of the jets, three probes were traversed sequentially across the jet flow. The three probes were modified types manufactured by United Sensor Control Corporation, Watertown, Massachusetts. The modifications consisted of stiffening the probes so that they would not vibrate in the jet. The three probes were:

- Total pressure probe, Type PT, 1/16-inch diameter
- Static pressure probe, Type PS, 1/16-inch diameter
- Total temperature probe, Type TC, 1/8-inch diameter

The probe pressures were converted to electrical signals by the use of carrier-type pressure transducers, the outputs of which were recorded on a multiple-channel strip chart recorder. The total pressure probe was used in conjunction with a 0-100 psig Statham pressure transducer, Type PG7317C-100-350, while the static pressure probe used a Statham ± 5 psig pressure transducer, Type PG0aTC $\pm 5D-350$. Both transducers used Sanborn carrier preamplifiers, Model 150-1100AS. The total temperature probe was connected, via a reference junction contained in an ice bath, to the input of a Sanborn stabilized dc preamplifier, Model 150-1800.

The pressure transducers and associated instrumentation were calibrated as systems by the introduction of known reference pressures at the transducers, the gains of the amplifiers then being adjusted to calibrate the chart records. The reference pressures at the transducers were measured by means of mercury manometers. The total temperature probe system was calibrated by the insertion of known voltages into the ass. riated preamplifier, together with amplifier gain adjustment, to calibrate the chart records. The insert voltages were derived from standard tables of the generated electromotive forces of Chromel-Alumel thermocouples at different temperatures. The probe calibration was checked periodically at two fixed temperature points by inserting the probe alternately into a melting ice bath and into boiling water.

3.3.5 Discrete Tones and the Resulting Hardware Modifications.

CONFIGURATION PORT

With very few exceptions, full-scale jet engine exhausts produce random noise without discrete tones. However, discrete tones are sometimes produced by modelscale jets. Because of this and the fact that occurrence of discrete tones would invalidate the data, all runs were monitored aurally, via a loudspeaker amplifier system connected to a microphone in the test room. The discrete tones thus encountered and the hardware modifications required to eliminate them will now be described.

The first discrete tone was encountered during one of the equipment-checkout runs, under the following conditions. The primary nozzle (design pressure ratio (R) = 3.5) and secondary nozzle No. 2 were installed in the coplanar position, but with no flow through the primary nozzle. As the secondary supply pressure was gradually brought forward (through the range to sonic secondary flow at 13.0 psig), the following sounds were observed. A pure tone was first heard at 4.7 psig; it gradually grew more intense until 6 psig was reached, remaining present at apparently constant intensity thereafter. Between 6 psig and 13.0 psig, several sudden jumps in pitch were observed as the pressure was increased. It was found that the tone could be completely "turned off" at will by placing a fingertip (or screwdriver) very slightly into the flow at the exit plane at any point on the periphery. Stopping the tone apparently depends upon destroying the perfect circularity of the nozzle exit by the very slightest amount. From this observation, plus the previously described jumps in pitch, it was concluded that the phenomenon observed was probabily a ring tone which is known to depend on an acoustic feedback mechanism.

An experiment was made to determine whether a small button-shaped spoiler (washer) could be added at the periphery of the secondary nozzle without adding measurably to the random noise generation. Using the same conditions as Run No. 109, where there were no discretes present, a comparison was made of the overall noise (in a 120-degree sweep about the jet) with and without the spoiler, via on-line level recorder trace. There was no measurable difference. This test was followed by octave band sweeps (from the 125 Hz band through the 31.5 kHz band) using tape-recorded data. Again, with flow conditions set and held constant through the pair of tests, there was no measurable difference. Hence, it was concluded that the addition of the spoiler did not generate any measurable amount of additional random noise, and the spoiler was kept in place on secondary nozzle No. 2 during all subsequent production runs.

The second discrete tone was encountered during Run No. 125; by a process of elimination, it was found that the tone was being produced by the primary nozzle and occurred even when the secondary nozzle was removed. It was found that the tone or whistle disappeared if the operating pressure ratio was reduced from the design value of 2.5 to 2.45. This was confirmed by one-third octave band traces. Large, spatiallydependent spikes in several third-octave bands occurred during microphone sweeps when the audible whistle was present; these spikes disappeared when the whistle disappeared.

In later runs, it was found that this whistle (with this primary nozzle, and also with the high supersonic primary nozzle, R = 3.5) returned and could not be removed by slight alterations of the supply pressure. At this point, it was decided that these two primary nozzles would have to be modified slightly by adding a small brazing bead to the inner edge of the exit. This succeeded in eliminating the pure tone or whistle, again borne out by one-third octave band sweeps. The presence of the brazing beads did not add any measurable random noise, as determined from comparisons of level recorder traces in those third-octave bands in which tones were absent during occurrence of the whistle. All subsequent runs were made with these brazing beads affixed.

3.4 Analytical Basis for Data Reduction Computer Program

Upon completion of the model jet experiments, the acoustic data were in the form of octave band sound levels encoded as a series of 3-digit numbers. A computer program was therefore written to convert this data to model jet sound pressure levels in decibels relative to 2×10^{-4} dynes per square centimeter, and to scale these levels to the values for full-scale jet engines. Figure 29 indicates the calculations made by the computer program and the order in which they are carried out. The details of these calculations are given in this section. A description and listing of the computer program is given in Appendix C.


Figure 29. The Functional Flow Loop for the Computer Program.

3.4.1 Flow Velocities and Thrust.

During each model jet run, measurements were made of the primary and secondary mass flows and static (reservoir) temperature and pressure. These values were used to calculate the primary and secondary flow velocities. From Reference 31, the flow Mach number, M, as a function of the total pressure ratio, R, across the nozzle is given by:

$$M = \left[5 \left(R^{*2857} - 1 \right) \right]^{0.5}$$
(19)

and the nozzle flow temperature, T, is related to the total temperature, T_{t} , by:

$$T = T_{t} (1 + M^{2}/5)^{-1}$$
(20)

This temperature can then be used to determine the speed of sound, a, in the flow (assuming standard conditions of $a_0 = 1120$ feet per second at 519° R). Thus:

$$a = a_0 (T/T_0)^{1/2}$$
 (21)

so that the nozzle velocity, U, is given by the product of the Mach number and the speed of sound:

$$U = Ma \tag{22}$$

Although the thrust of the model jets was measured during each run, it was found that there were some inconsistencies in the measurements of the early runs due to instrumentation-drift problems. These problems were overcome in the latter runs but in order to maintain consistency, the thrusts, F, were also calculated using the measured primary and secondary mass flows (m_p, m_s) and the calculated velocities (U_p, U_s) . Thus:

$$F = \frac{m_p U_p + m_s U_s}{g}$$
(23)

A check was made between the calculated and measured values and the agreement found to be very good for all the latter run numbers. Therefore, calculated thrust values have been used for all subsequent analyses for these runs.

However, these equations must be modified for the runs in which the primary nozzle was retracted. The actual velocity, U_p^{-1} , at the exit of the retracted primary is determined by the ratio of the primary nozzle pressure to the secondary nozzle pressure. The secondary flow is augmented by the mixing between the primary and secondary flows upstream of the secondary nozzle exit, and its velocity is a function of position across the nozzle. The flow velocity ratios tabulated for these retracted cases are U_s/U_p^{-1} , where U_s is calculated for an ideal convergent nozzle, as previously. Also, the measured values of the thrusts were used for scaling and tabulation, since the calculated values are somewhat indeterminate because of the flow interaction. However, they were reasonably bounded by the minimum and maximum expected values given by:

$$\frac{\stackrel{m}{p} \stackrel{v}{p} \stackrel{r}{s} \stackrel{s}{s}}{g} \leq F \leq \frac{\stackrel{m}{p} \stackrel{v}{p} \stackrel{r}{s} \stackrel{s}{s}}{g}$$
(24)

where prime indicates conditions at the exit plane of the retracted primary nozzle.

3.4.2 Sound Pressure Levels.

The measured acoustic pressure, p, was calculated from the product of the calibration factor (see Section 3.3.2) and the encoded 3-digit number. The sound pressure level (SPL) was then determined using:

$$SPL = 20 \log p + 74, dB re 2 \times 10^{-4} dynes/cm^2$$
(25)

and the relevant octave band correction factors added (see Section 3.3.2). Since these correction factors are frequency-dependent, the correction factor for the overall level is dependent on spectrum shape. It is therefore not practical to calculate an overall level correction factor, so the measured overall level was replaced by the overall level computed from the sum of the octave band levels.

3.4.3 Sound Power Levels.

Since the acoustic data was taken in an anechoic room, the sound power radiated by the jets can be determined by integrating the acoustic intensity, I, over the surface of a sphere centered on the jet. The acoustic data was taken at 5-degree intervals. The sound pressure, p_i , measured at an angular position θ_i from the jet axis was therefore assumed to be representative of the sound pressure over an area A_i such that:

$$A_{i} = 2\pi r^{2} (\cos(\theta_{i} - 2.5^{\circ}) - \cos(\theta_{i} + 2.5^{\circ}))$$
 (26)

This is the area of a strip of the spherical surface (radius r) centered on the jet axis at the plane of the secondary nozzle exit. The strip subtends an angle of 2.5 degrees either side of the measurement position θ_i and extends all the way around the jet axis. The sound was measured at 21 angular positions ranging from $\theta_1 = 15^{\circ}$ to $\theta_{21} = 115^{\circ}$; so the sound power, W, radiated is:

$$W = \sum_{i=1}^{21} (p_i^2 A_i / \rho_a)$$
(27)

This summation was made for each octave band and the overall sound power calculated by summing the octave band powers. The octave band and overall sound power levels (PWL) were then determined as:

$$PWL = 10 \log W + 130, dB re 10^{-13} watts$$
(28)

The value of the characteristic acoustic impedance (ρa) used in the calculation was 41.5 cgs rayls at a pressure of 29.92 inches of mercury and absolute temperature of 519°R. The impedance was corrected for the ambient temperature, T_a , and pressure, P_a , measured for each run, so that the value of acoustic impedance used in Equation (27) was:

$$\rho a = 41.5 (P_2/29.92) (519/T_2)^{0.5} cgs rayls$$
 (29)

The radius, r, at which the sound measurements were made was measured in feet, so area A_1 from Equation (26) is in square feet. If the acoustic pressure, P_1 , is in dynes per square centimeter, the dimensional constant required in Equation (27) is 9.29 x 10⁻⁵ so that it should read:

$$W = 9.29 \times 10^{-5} \sum_{i=1}^{21} P_i^2 A_i / \rho_0, \text{ watts}$$
(30)

The above calculations have been limited to the acoustic power produced by the modul jet. Each model jet has a different thrust, so for comparison it is convenient to scale the overall sound power produced by the model jet, PWL_m , with thrust F_m to give the sound power produced by an engine with a standard thrust, F_S . This requires the scaling of the model jet by scaling the nozzle exit area, which is directly proportional to thrust, since the flow characteristics of the model and scaled jets are the same.

The scaled power level, PWL_S, is therefore given by:

$$PWL_{S} = PWL_{m} + 10 \log (F_{S}/F_{m})$$
(31)

Throughout the computer program, four standard thrusts with values of 10,000 pounds, 20,000 pounds, 40,000 pounds and 80,000 pounds were used.

3.4.4 Directivity Patterns.

Having calculated the model power spectra, the directivity index, DI, can be found using the expression:

$$DI = SPL - PWL + 20 \log r + 10.5 - 10 \log \rho_0/41.5$$
(32)

In this expression, SPL is the measured octave band sound pressure level at distance r (in feet) from the jet and PWL is the measured octave band power level. The constant in Equation (32) assumes a reference sound power of 10^{-13} watts and a reference sound pressure of 2×10^{-4} dynes per square centimeter.

At this point in the computer program, an indication is given of the level of the noise floor relative to the zero level on the directivity. This is a function of the calibration factor for the sound pressure measurements, and was provided as a system check.

3.4.5 Acoustic Conversion Efficiency.

The ratio of the acoustic power, W, produced by the jet to the mechanical power in the jet, E, is one additional measure of the acoustic performance of the jet. The mechanical power in watts is given by:

$$E = 0.0211 \ (m_p U_p^2 + m_s U_s^2) \tag{33}$$

when the primary and secondary mass flows (m_p, m_s) are in units of pounds per second and the velocities (U_p, U_s) in feet per second. Using the acoustic power, W, determined earlier, the acoustic efficiency, η , is given as:

$$\eta = W/E \tag{34}$$

3.4.6 Sideline Sound Pressure Levels for Fuli-Scale Engines.

The scaling of the model acoustic data to give the sound pressure levels on a sideline parallel to the jet axis due to a full-scale angine requires the following operations on the model data:

- Determine scale factor to scale-model thrust to standard thrust.
- Shift model frequency to frequency of standard thrust nozzle by the inverse of the square root of the scale factor.
- Apply inverse square law appropriate for the desired distance.
- Apply atmospheric attenuation as appropriate.

In scaling the measured data to the equivalent values on a sideline, the physical situation is as shown in Figure 30. The data taken on the circular arc (radius r) must be scaled to lie on the sideline. The position of this sideline is characterized by the sideline distance L_d (see Figure 30). For the purposes of this project, two sideline distances of 500 feet and 1500 feet were chosen. The sound pressure measured on the model jet at coordinate (r, θ) is therefore scaled to the point (L, θ) on one of these sidelines. The sound pressure decreases by 20 log (L/r) due to this change.

In order to scale the results to allow for a change of thrust, the engine thrust must be increased without altering any of the jet flow parameters. Thus, the scaling is done purely by changing the nozzle area. The thrust change is then directly proportional to the change in area. For the same flow velocity and density, the acoustic output is also proportional to area. Hence, if the thrust is scaled from a model thrust of F_m to a full-scale thrust of F_S , the sound pressure level at a fixed radius will change by 10 log (F_S/F_m) .

For this project, the full-scale thrusts (F_5) lie in the range 10,000 pounds to 80,000 pounds while the model jets had thrusts ranging from 6 pounds to 120 pounds. Similarly, for the 1500-foot sideline, the propagation distances range from 1500 feet to 6000 feet, compared to a model-jet measurement distance of 10.67 feet. For these values, the combined distance and thrust scaling correction varies over a range from -2 dB to -36 dB.

The atmospheric absorption data were taken from Reference 32. The data for a standard day with a temperature of 519^oR and relative humidity of 70 percent were used and are plotted in Figure 31.



Figure 30. Geometry for Obtaining Sideline Sound Pressure Levels from Measured Data.



be in the state of the state of



For the computer program, it was desirable to fit a curve through the data points. This resulted in the following empirical equation for the absorption $\alpha(f_{S})$ in dB per 1000 feet:

$$\alpha(f_{\rm C}) = 10^{b(f_{\rm C})} \tag{35}$$

where

มสะ แสมให้ปฏกเชีย และใจเสียงสองสีของเห็มชิมิตน้ำต่อสีขึ้น ให้ก็มีชื่อแทดง แสน้ำ ก็มีก้องได้ และครั้งไป 4 กา

Second and the second start field of

 $b(f_{S}) = \alpha_{0} + \log(f_{S}/1000)(\alpha_{1} + \alpha_{2} \log(f_{S}/1000))$ fs is the octave band center frequency for the full-scale engine $\alpha_{0} = 0.127$ $\alpha_{1} = 1.338$ $\alpha_{2} = -0.207$

This expression was chosen salely for convenience in the computer analysis. It provided values for air absorption over the full-scale frequency range which were in substantial agreement with values computed by a more exact theoretical model for the measured data (Reference 33). Absorption corrections were not required for the model data because the distance from the model jet to the measured point was small.

Combining these factors, the total charge from the model sound pressure level, SPLm, to full-scale sound pressure level, SPLS, is given by:

$$SPL_{s} = SPL_{m} + 10 \log (F_{s}/F_{m}) + 20 \log (r/U - L \sigma(f_{s})/1000)$$
(36)

The change in frequency accompanying this change in level is based on the use of a constant Strauhal number, S, before and after the scaling process. Thus:

$$S = f d/U = constant$$
 (37)

where f is a frequency, d is a typical dimension of the jet, and U is a typical jet valacity. We have specified constant flow conditions in the scaling process. Hence, U is unaltered by the scaling. However, the area of the jet is altered in proportion to the thrust, so a typical linear dimension changes according to the square root of the thrust. Thus, if $f_{\rm sp}$ is a typical frequency of the model jet, applying Equation (37) yields:

$$f_{S} = f_{m} \left(F_{m} / F_{S} \right)^{0.3}$$
 (38)

This aquation governs the frequency scaling and also is used to define the full-scale frequency used in Equation (35).

3.4.7 Perceived Noise Level.

Once the sideline octave band levels have been determined, the parceived noise levels can be calculated. The method used is specified in Reference 34, utilizing the equations given in Table II of that reference to calculate the Noy values. Under certain circumstances, the lowest and highest octave bands of sideline sound pressure fall outside the range of 50 Hz to 10 kHz covered by this reference. For octave band frequencies below 50 Hz, the Noy value was calculated as if the octave band frequancy were 50 Hz. Frequencies above 10 kHz were assumed to make no contribution to the parceived noise level and were therefore ignored.

The original model-jet sound pressure measurements were taken in seven octave bands up to and including the 31.5 Hz band. When these duta are scaled to the higher thrust values, the upper frequency band does not always extend above the frequency containing the peak in the Noy curves. This condition often occurred with the largest scaled thrust of 80,000 pounds, sometimes resulting in erroneously low perceived noise level values. To overcome this deficiency, the model data were extrapolated by adding four extra actave bands above the 31.5 kHz band, assuming a 3 dB per active decrease in level from the 31.5 kHz band. In many cases, the atmospheric attenuation reduces these extra bands to negligible small levels when they are scaled to the sideline. In certain cases for the higher scaled thrusts, the first of these added bands has a significant effect on the perceived noise level. In order to indicate the magnitude of this effect, the perceived noise level calculations were made twice and printed out by the computer. The first value of PNL was obtained using all actave bands, including the four extra bands. The second value of PNL (printed in parenthesis) utilizes only the original seven octave bands of data. In this way, the effect of the added bands is clearly seen for each case.

4.0 DESCRIPTION OF THE EXPERIMENTAL DATA

During the course of this program, the sound from over 310 model jet configurations was measured and processed by the computer program. An effective presentation of all of these detailed data could not be made within one cover. Thus, the principal results are given in this report, supplemented by a detailed tabulation of the results in Volume II. Appendix A, in Volume I, gives a detailed listing of the run numbers and associated run conditions. This section describes the format of the detailed data data and discusses a few pertinent points concerning the results.

4.1 Computer Output Data

For each experimental run condition, the output of the computer program consists of the following major items of information:

- The basic run conditions, including such items as mass flow, velocity, thrust and pressure ratio.
- (2) The octave band sound pressure levels for the model jet. The data are presented in seven octave bands with frequencies ranging from 500 Hz to 31.5 kHz, and an overall level computed as the sum of the octave band levels. Levels are given at 5-degree angular increments, starting at an angle of 15 degrees from the jet axis in the flow direction and extending to 115 degrees around an arc of 10.67 fect radius.
- (3) The model jet sound power octave band spectrum and overall level. This information is the result of an integration of the sound power over a spherical surface within the angular limits specified in (1) above.
- (4) The overall sound power level for full-scale jets with thrusts of 10,000 pounds, 20,000 pounds, 40,000 pounds and 80,000 pounds.
- (5) The model jet directivity index.

- (6) The model jet mechanical power and acoustical conversion efficiency.
- (7) The octave band sound pressure levels, overall sound pressure levels and perceived noise levels on a 500-foot sideline. These levels are given for each of the 5-degree angular positions used in the model sound pressure level measurement (see (1) above). The coordinate defining position on this output (L) is the distance from the jet to the sideline at various angles. Two values of the perceived noise level are printed, as explained earlier in Section 3.4.7. The first value is computed from

the measured plus extrapolated octave band levels. The value in parentheses utilizes only the seven measured octave bands in the perceived noise level calculation.

(8) The same information as in item (7) above, but for a 1500-foot sideline.

This information occupies many pages of computer output and was condensed for presentation in Volume II in the format shown in Figure 32. The data presented include items (i), (2), (3), (4) and (8) for the 20,000 pound thrust case only. From these data, two sets of summary plots were made. The first set gives the overall sound power level for a 20,000 pound thrust engine as a function of mass flow ratio (ratio of secondary mass flow to primary mass flow), and the second presents the maximum 1500-foot sideline perceived noise level as a function of mass flow ratio. For each of these sets of plots, each graph represents one combination of primary nozzle temperature, primary pressure ratio and primary axial position. Thus, 28 plots are required for each set to cover all run conditions measured in the experiment. The complete set of plots for overall power level and for perceived noise level are given in Appendices D and E, respectively.

For illustration, an example from each set is presented in Figures 33 (overall sound power level) and 34 (perceived noise level). Each of these graphs contains a point on the vertical axis, giving the result for the primary nozzle alone, and four solid lines for the four different values of the nominal ratio of secondary nozzle area to primary nozzle area. (See Section 3.2.1 for the exact values.) The experimental points or each of the four lines correspond to various velocity ratios.

į

The experiments documented in this report include the determination of the directivity of the jets measured in an anechoic room where there was no reflecting ground plane and the possibility of the jet exhaust being deflected by low velocity air currents was eliminated. Many of the measurements of directivity patterns previously reported were made in the open where the presence of a ground plane can significantly affect the measured results (References 5 and 35), and a slight breeze can markedly bend the jet exhaust plume and hence distort the directivity. It is therefore felt that a useful side benefit from the present tests is a set of single-nozzle directivity patterns from which all ground plane and cross-wind effects have been removed. Consequently, a set of directivity plots and sound power spectra for each of the primary nozzles acting alone is presented in Appendix F.

| | 411 | N NUMBER

 |

 |
 |
 |
 | ·
 | 103.600 |
 | - | |
 | |
|--|---
--
--

--
--
--
---|---
--

--
--|---|--|--
---|--|---|
| | 41 | AL POSITION

 | OF PRIMAN

 | W NOT. SE
 | CONDARY LINS
 | ial
 |
 | 4.041 |
 | | |
 | |
| | 560 | LOIDYBA ILAGLET
Infaa llaglet

 | (1-)4F [4]
WATOUS 16

 | **
 |
 |
 |
 | 525,000 |
 | | |
 | |
| | P91 | HANY PHESSUN

 | E WATIO

 | ··
 |
 |
 |
 | 1.400 |
 | ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~ | |
 | |
| | A76 | LE HATIO

 |

 |
 |
 |
 |
 | 1.000 |
 | | |
 | |
| | VEL | OCITY RATIO

 |

 |
 |
 |
 |
 | .700 |
 | | |
 | |
| | | INARY VELOCIT

 | T WINE

 | 1
 |
 |
 |
 | 101.542 |
 | | |
 | |
| | PR | HARY MASS FL

 |

 | (C)
 |
 |
 |
 | .217 |
 | | |
 | |
| | THE | RUST (LAS)

 |

 |
 |
 |
 | •
 | 9.932 |
 | | |
 | ***** |
| | ENI | ¥1905+E5141 T

 |

 | RE (R)
 |
 |
 | ! <u>*</u> !
 | 523.504 |
 | | |
 | |
| | ENI | TPON A HTAL H

 | DATELLA .

 | PFR CFNTI
 |
 |
 |
 | 67.004 |
 | | |
 | |
| | TAJ | THRATIOS TAC

 | 111-141-1

 | 10 DY750 C
 | 45
 |
 |
 | .007 |
 | | |
 | |
| | IN | STAIMENTATION

 | NOISE FL

 | DON (OB)
 |
 |
 |
 | 48.541 |
 | | |
 | |
| | |

 |

 |
 |
 |
 |
 | |
 | | |
 | |
| • | • • |

 |

 |
 | ~~
 |
 |
 | |
 | | |
 | |
| | |

 |

 |
 |
 |
 |
 | |
 | | · · · · · · · · · · · · · · · · · · · | ·····
 | |
| #40031 | 10 -0 |

 |

 |
 | HUNCE JEI
 |
 |
 | |
 | | |
 | |
| | |

 |

 |
 | ****
 |
 | ·····
 | |
 | | |
 | |
| F # | REGUEN | 1CY PO

 | VERIVATTS

 | 1 PO
 | VER LEVELIDE
 | 1
 | 07
 | ERALL SO | JND POVER
 | LEVEL | SCALED | FOR THRU
 | JST |
| 0 | | ,
1 1.

 | 153538-01

 |
 | 120.4
 |
 |
 | THRU | ST PI
 | WER LI | EVEL (OR) |
 | |
| | |

 |

 |
 |
 | ·
 |
 | 100 | 0
 | | | .
 | |
| | 50 | . 1.

 | 30182F+03

 | L
 | 101.1
 |
 |
 | 200 | 90
 | 153 | .7 |
 | |
| | 100 | 7.

 | 302256+03

 |
 | 108.6
 |
 |
 | - 400 | 30
 | 156 | •7 |
 | |
| | - 400 | ð 3.

 | 125++2-02

 |
 | 114.9
 |
 |
 | |
 | 154 | |
 | |
| | 800 | 2.

 | 74616E-32

 |
 | 114.4
 |
 |
 | |
 | | |
 | |
| | 1600 | 1.

 | 6026+F+02

 |
 | 112.0
 |
 |
 | |
 | | | ~
 | |
| | 3150 | <u>0</u>

 | 65123E-03

 |
 | 108.2
 |
 |
 | |
 | | / | へ
 | |
| | |

 |

 |
 |
 |
 |
 | |
 | | - Ka | 70
 | |
| | |

 |

 |
 |
 |
 |
 | |
 | | / | 3 3 1-
 | |
| | | BEGINENTH - N

 |

 |
 |
 |
 |
 | |
 | | / | ~ °^
 | · |
| | ŤÂ | BLE SHOWING (

 | OCTAVE BA

 | NO AND OV
 | FRALL SOUND
 | PRESSURE L
 | EVEL VARI
 | ATTON WE | H ANGLE
 | | | 126
 | \mathbf{N} |
| | |

 | هو تستقيم بند

 |
 |
 |
 |
 | |
 | | | -/
 | · / |
| | |

 |

 |
 |
 |
 |
 | |
 | | · | \`&
 | °\ |
| | NO # |

 | OCTAY

 | F. 84ND
 | Southe
 | PRESSURE
 | 1
 | |
 | ivro | | \o
 | <u> </u> |
| î | DEGY- | 500

 | - 1000

 | 2000
 | 4000
 | 8000
 | 16000
 | 3150 | ·
 | ALL | | /
 | ∾ %/ |
| | |

 |

 |
 |
 |
 |
 | |
 | | |
 | \&^ \. |
| | 15.0 | 30.5

 | 88.4

 | 42.5
 | 87.8
 | 02.5
 | 10.3
 | 71+6 | , ,
 | 12+1 | |
 | 12 1 |
| | - | / 7.6

 |

 | ~ ~ ~ ~
 |
 |
 |
 | |
 | | |
 | · · · · · · · |
| | 5.1 |

 |

 |
 |
 | 84,2
 | 78.0
 | 73,0 |
 | 75.9 | |
 | -\12 _ |
| | 25.0 | 77.8

 | 86.1

 | 91.4
A9.9
 | 90.6
9.09
 | 84.2
86.0
87.4
 | 78.0
79.9
82.0
 | 73,0
75,
77,0 |
 | 75.9
75.5
75.0 | |
 | -12 |
| | 25.0
30.0
35.0 | 77.8
76.8
74.6

 | 86.1
84.5
82.3

 | 01.4
A9.9
88.1
 | 90.6
90.2
89.1
 | 84,2
86.0
87:4
87.8
 | 78.0
79.9
82.0
83.4
 | 73,0
75,
77,0
78, |
 | 75.9
75.5
75.0
75.1 | |
 | |
| | 25.0 | 77.8
76.8
74.6
72.9

 | 86.1
84.5
82.3
80.6

 | 91.4
A9.9
88.1
RA.7
 | 90.6
90.2
89.1
88.9
 | 84,2
86.0
87.4
87.8
87.8
87.9
 | 78,0
79,9
82,0
83,4
84,3
 | 73,
75,
77,
78,
79, |
 | 25.9
25.5
25.0
24.1
23.6 | |
 | |
| | 25.0
30.0
35.0
45.0 | 77.8
76.8
74.6
72.9
71.3
70.2

 | 86.1
84.5
82.3
80.6
78.8
78.8

 | 91.4
A9.9
88.1
R6.7
84.9
 | 90.6
90.2
89.1
88.9
87.2
86.3
 | 84,2
86.0
87.4
87.8
87.8
87.9
87.3
86.6
 | 78.0
79.9
82.0
83.4
84.7
84.7
 | 73,0
75,
77,0
78,
79,0
40,0 |
 | 25.5
25.5
25.6
2.6
2.7 | |
 | |
| | 25.0
30.0
35.0
45.0
50.0
55.0 | 77.6
76.8
74.6
72.9
71.3
70.2
69.7

 | 86.1
84.5
82.3
82.3
80.6
78.8
77.6
76.7

 | 91.4
A9.9
88.1
R6.7
84.9
84.1
B2.9
 | 90.6
90.2
89.1
68.5
87.2
46.3
65.2
 | 84.2
86.0
87.4
87.8
87.3
87.3
86.6
85.8
 | 78,0
79,9
82,0
83,4
A4,3
84,7
84,7
84,6
84,0
 | 73,
75,
77,
78,
79,
40,
40,
80, |
 | 75.5
75.5
75.6
73.6
72.7
72.0 | |
 | |
| | 25.0
30.0
35.0
45.0
50.0
55.0
60.0 | 77.6
76.8
74.6
72.9
71.3
70.2
69.7
69.8

 | 86.1
84.5
82.3
#0.6
78.8
77.6
76.7
76.0

 | 91.
A9.9
88.1
86.7
84.9
84.1

 | 90.6
90.2
89.1
88.5
87.2
46.3
65.2
84.5
 | 84,2
86.0
87:4
87.8
87.9
87.3
86.6
85.6
85.6
 | 78.0
79.9
82.0
83.4
84.7
84.7
84.6
84.0
83.4
 | 73,
75,
77,
78,
79,
40,
80,
79,
80,
79, | 5
5
5
7
7
 | 75.9
75.5
75.0
73.6
72.7
72.0
71.2
70.6 | |
 | |
| | 25.0
30.0
35.0
45.0
50.0
55.0
60.0 | 77.8
76.8
74.6
72.9
71.3
70.2
69.7
69.7
69.8

 | 86-1
84-5
82-3
#0.6
78-8
77-6
76-7
76-0
76-1

 | 84.1
84.1
84.7
84.9
84.1
82.9
84.1
82.9
81.6
 | 90,6
90,2
89,1
88,5
87,2
46,3
65,2
84,5
84,4
 | 84,2
86.0
87.4
87.4
87.9
87.4
86.6
85.6
85.6
85.1
 | 78,0
79,9
82,0
83,4
84,3
84,7
84,6
84,6
84,0
83,4
83,4
83,2
 | 73,
75,
77,
78,
79,
40,
80,
79,
80,
79,
80, |
 | 25.9
25.5
25.0
24.1
23.6
22.7
22.0
21.2
20.6
20.5 | |
 | |
| | 25.0
30.0
35.0
45.0
50.0
55.0
60.0
65.0
70.0 | 77.8
76.8
74.6
72.9
71.3
70.2
69.7
69.8
69.8
69.8

 | 86.1
84.3
82.3
80.6
78.8
77.6
76.7
76.0
76.1
75.6

 | 82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
81.6
81.6
 | 90.6
90.2
89.1
88.5
87.2
85.2
85.2
84.5
84.4
83.6
 | 84,2
85.0
87.4
87.4
87.4
87.9
87.3
86.6
85.6
85.6
85.6
85.1
85.1
 | 78.0
79.9
82.0
83.4
84.7
84.7
84.6
84.6
83.0
83.4
83.2
83.0
 | 73,1
75,77,
78,
79,4
40,4
40,4
40,4
40,4
40,4
40,4
40,4
79,1
70 |
 | 23.9
25.5
25.6
24.1
23.6
22.7
22.0
21.2
20.6
20.5
39.9 | |
 | |
| | 25.0
30.0
35.0
45.0
55.0
55.0
65.0
75.0
80.0 | 77.8
76.8
74.6
72.9
71.3
70.2
69.7
69.8
69.8
69.8
69.8
69.8
69.4
68.7
68.4

 | 86.1
84.5
82.3
82.3
82.3
82.3
82.4
78.8
77.6
76.0
76.1
75.6
75.3
74.5

 | 91.4
A9.9
98.1
R6.7
84.9
84.1
82.9
82.0
81.6
81.6
81.9
41.0
40.5
 | 90.6
90.2
89.1
88.9
87.2
46.3
65.2
84.5
84.4
83.6
93.7
63.2
 | 84,2
86.0
87.4
87.8
87.3
86.6
85.6
85.6
85.1
85.1
84.1
83.6
 | 78,0
76,9
82,0
83,4
84,7
84,7
84,7
84,7
84,7
84,7
84,6
83,2
83,2
83,0
83,2
83,0
82,6
82,0
 | 73,
75,
77,
78,
79,
40,
40,
40,
79,
90,
79,
79,
79,
79,
79, |
 | 25.9
25.5
25.5
25.0
2.0
2.0
2.0
2.0
2.0
2.0
2.0
2 | |
 | |
| | 20.0
30.0
35.0
45.0
50.0
55.0
60.0
70.0
75.0
80.0
85.0 | 77.8
76.8
74.6
72.9
71.3
70.2
69.7
69.7
69.7
69.4
68.7
68.4
68.3

 | 86.1
84.5
82.3
78.8
77.6
76.7
76.0
76.1
75.6
75.3
74.5

 | 91.4
A9.9
88.1
R6.7
84.9
84.1
82.0
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.6
 | 90.6 90.2 89.1 88.5 87.2 85.2 84.5 85.2 84.5 83.6 83.7
 | 84,2
86.0
87.4
87.8
87.9
87.3
86.6
85.6
85.6
85.1
84.1
83.6
83.6
83.6
 | 7810
79.9
82.0
83.4
84.7
84.7
84.7
84.7
83.0
83.4
83.0
83.4
83.0
83.4
83.2
83.0
82.6
82.0
81.8
 | 73,
775,
778,
794
40,
40,
40,
794
80,
794
40,
794
794
794
794,
794, |
 | 25.9
95.5
95.6
95.6
95.6
95.6
93.6
93.6
93.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6
90.6 | |
 | |
| | 20.0
30.0
35.0
45.0
50.0
55.0
65.0
65.0
65.0
70.0
70.0
80.0
85.0 | 77.8
76.8
74.6
72.9
70.2
69.7
69.8
69.4
69.4
65.4
65.4
65.1
66.3

 | 86.1
84.5
82.3
78.8
77.6
76.0
76.7
76.0
76.1
75.6
75.3
74.5
74.5
74.2

 | 91.4
A9.9
98.1
R6.7
84.1
B2.9
82.0
82.0
82.0
81.6
60.9
A1.0
A0.5
80.1
79.7
 | 70 6 90 2 87 1 88 4 87 2 86 3 85 2 84 5 84 5 84 5 83 7 83 7 83 7
 | 86.0
87.4
87.8
87.9
87.9
87.3
86.6
85.6
85.6
85.2
85.4
84.1
83.6
83.4
84.1
83.6
83.4
84.1
 | 7810
79.9
82.0
83.4
A4.7
84.7
84.7
84.7
84.7
83.0
83.2
83.0
83.6
82.6
81.8
91.8
 | 73,
775,
778,
794,
40,
40,
40,
40,
79,
79,
79,
79,
79,
78,
78,
78, |
 | 25,9
5,5
5,0
4,1
7,6
7,6
7,6
7,6
7,6
7,6
7,6
7,6 | |
 | |
| | 20.0
30.0
35.0
45.0
55.0
65.0
65.0
65.0
65.0
65.0
70.0
70.0
80.0
85.0
80.0
85.0 | 778
76.8
76.8
72.9
71.3
70.2
69.7
69.6
69.4
69.4
68.4
68.5
68.5
68.5
67.4

 | 86.1
94.5
82.3
82.3
78.8
76.7
76.0
76.1
75.6
75.3
74.5
74.5
74.2
74.2
74.2

 | 91.4
A9.9
98.1
RA.7
84.9
B2.9
82.0
82.0
82.0
81.6
60.9
A1.0
A0.5
80.1
79.7
79.7
79.7
 | 00,6 00,6 00,2 89,1 88,4 83,7 83,7 83,7 83,7 83,7 83,7 83,7 83,7 83,7 83,7 83,7 83,7 83,7
 | 84,2
86.0
87.4
87.8
87.3
85.6
85.6
85.2
85.1
85.1
85.1
85.1
85.1
85.1
85.1
85.1
 | 78,0
79,9
82,0
83,4
83,4
84,7
84,6
83,4
83,4
83,4
83,4
83,6
83,6
83,6
83,6
83,6
83,6
83,6
83,6
 | 73,
775.
778.
794.
40.
40.
40.
794.
794.
794.
794.
794.
794.
794.
794 |
 | 25,9
15,5
15,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6
17,6 | |
 | |
| | 235.0
35.0
45.0
55.0
65.0
65.0
75.0
65.0
75.0
85.0
75.0
85.0
95.0
65.0
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
65.0
1
7
8
7
8
7
8
7
8
7
8
7
8
7
8
7
8
7
8
7 | 77.8
76.8
76.8
77.9
71.3
70.2
69.7
69.7
69.7
69.7
69.7
69.7
69.7
69.7

 | 86.1
94.5
82.3
80.6
78.8
77.6
76.0
76.1
75.6
75.3
74.5
74.5
74.5
74.2
74.2
73.6
73.0

 | 91.4
R9.9
88.1
RA.7
84.2
82.0
82.0
81.6
81.6
80.9
A1.00
A0.5
80.1
79.7
N0.1
79.5
79.5
 | 70 n 90.6 90.2 87.1 86.4 87.2 84.4 83.6 83.6 83.7 82.7 83.7 83.4 83.8 83.4 83.4 83.4 83.4 83.4
 | 84,2
86.0
87.4
87.8
87.3
86.6
85.6
85.6
85.6
85.6
85.6
85.6
85.6
 | 78,0
79,9
82,0
83,4
84,7
84,7
84,7
84,6
83,2
83,2
83,2
83,2
83,2
83,2
83,2
83,2
 | 73,
77.
77.
78.
79.
40.
40.
40.
79.
79.
79.
79.
79.
79.
79.
79.
79.
79 |
 | 25.9
25.5
25.6
25.6
25.6
22.7
22.0
22.0
20.6
20.5
20.5
20.5
20.6
20.5
20.6
20.5
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6
20.6 | |
 | |
| | 235.0
35.0
35.0
45.0
55.0
55.0
65.0
75.0
75.0
75.0
75.0
75.0
75.0
10.0 | 7 7 5
7 7 5
7 7 6
7 6

 | 86.1
84.5
82.3
82.3
82.3
78.8
77.6
76.7
76.7
76.7
76.7
76.7
75.3
74.5
74.5
74.5
74.5
74.2
73.6
73.0

 | 91.4
A9.9
88.1
88.7
84.9
84.1
82.0
82.0
81.6
81.6
81.6
81.6
81.6
81.7
8.5
75.5
79.5
78.5
78.5
 | 00.6 90.2 87.1 88.9 84.4 83.6 83.7 83.7 83.8 83.7 81.9 81.9 81.4 83.7 81.4 83.8 83.9 81.9
 | 84,2
86.0
87.4
87.3
87.3
86.6
85.4
85.4
85.4
85.4
85.4
85.4
83.6
83.4
83.5
83.4
83.5
83.4
83.5
83.4
83.5
83.4
83.5
83.4
83.5
83.4
83.5
83.5
83.5
83.5
81.9
 | 78.0
76.9
82.0
83.4
84.7
84.7
84.7
84.7
83.4
83.2
83.0
83.4
83.4
83.4
83.4
83.4
83.4
83.4
83.4
 | 73,
75,
77,
78,
78,
78,
78,
40,
40,
40,
79,
79,
79,
79,
79,
74,
74,
74,
77,
75,
75, |
 | 25.9
25.5
25.6
24.1
23.6
22.7
22.0
20.6
20.5
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6
30.6 | |
 | |
| | 235.00
35.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
55.00
5 | 77.5
76.8
76.8
74.6
72.9
70.2
60.7
60.7
60.7
60.6
60.4
60.4
60.4
60.4
60.4
60.4
60.4

 | 86.1
94.5
98.2
78.6
78.6
75.6
75.6
75.6
75.3
74.5
74.5
74.2
74.2
73.6
73.6
73.6
73.6
73.6
73.6
73.6

 | 91.4
91.4
84.9
98.1
84.7
84.9
84.1
84.7
84.9
84.1
84.7
84.9
84.1
84.7
84.9
84.1
84.7
84.9
84.1
84.7
84.9
84.1
84.7
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.9
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
 | 00.6 00.6 00.2 87.1 88.4 83.6 85.2 84.4 83.6 83.6 83.6 81.5 81.6 81.6 81.6 80.6
 | 84,2
86,0
87,4
87,4
87,4
87,4
85,6
85,6
85,6
85,6
85,6
85,6
85,1
84,1
84,1
83,6
83,4
83,4
83,4
83,4
83,4
83,4
83,4
83,4
 | 78.0
76.9
82.0
83.4
84.7
84.7
84.7
84.7
83.4
83.0
83.4
83.0
83.4
83.0
83.4
83.0
83.4
83.0
83.4
83.0
83.4
83.0
83.4
83.0
81.5
81.7
81.1
81.1
81.0
85.9
 | 73,
75,
77,
78,
79,
40,
80,
79,
80,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 |
 | 25.9
5.5
95.0
95.0
95.0
95.0
95.0
95.0
92.0
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
90.5
9 | |
 | |
| | 235.0
35.0
55.0
55.0
55.0
55.0
55.0
55.0
65.0
75.0
65.0
75.0
85.0
75.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0 | 77.5
76.8
76.8
77.6
72.6
71.3
70.2
69.7
69.8
69.7
69.8
69.4
69.4
69.4
69.4
69.4
69.4
69.4
65.4
65.5
65.5
65.5
65.5

 | 86.1
84.5
82.3
78.6
78.6
78.6
75.6
75.6
75.6
75.6
75.6
75.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.4
93.4
93.8
93.8
93.8
93.8
93.8
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.4
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
 | 00.6 00.2 87.1 86.5 77.2 87.1 86.5 87.1 86.5 87.1 87.1 86.5 87.1
 | 84,2 86,0 87,4 87,3 86,6 85,6 85,6 85,6 85,6 85,6 85,6 83,16 83,16 83,16 83,16 83,16 83,16 83,16 83,16 83,16 83,16 83,17 83,16 83,17 83,16 83,17 83,16 83,17
 | 78,0
79,9
82,0
83,4
84,7
84,6
83,4
84,7
84,6
83,2
83,6
83,2
83,6
83,2
83,6
83,2
83,6
83,2
83,6
83,2
83,6
81,5
81,7
81,2
81,1
81,0
80,9
 | 73,
75,
77,
78,
79,
40,
40,
40,
40,
79,
79,
79,
74,
74,
74,
74,
74,
77,
75,
75,
76,
76, |
 | 25.9
5.5
75.0
14.1
13.6
22.0
12.7
22.0
12.6
22.0
12.6
22.0
12.6
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.6
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
20.5
2 | |
 | |
| | 25.0
35.0
45.0
55.0
55.0
55.0
55.0
65.0
65.0
65.0
85.0
65.0
85.0
65.0
85.0
65.0
15.0
15.0 | 77.8
76.8
76.8
77.9
71.3
76.2
69.7
69.7
69.7
69.7
69.7
69.7
69.7
69.7

 | 86.1
86.1
82.3
70.6
78.6
75.6
75.6
75.3
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.4
92.4
84.1
84.1
84.1
82.7
84.9
82.7
84.9
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
82.7
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.1
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
 | 00.6 00.6 00.2 87.1 88.4 87.2 84.4 83.4 83.4 83.7 83.7 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.6 83.6
 | 84,2
86,0
87,4
87,9
87,9
86,6
85,6
85,6
85,6
85,1
84,4
83,1
83,1
83,1
83,3
83,3
83,3
83,3
83,4
83,5
83,4
83,5
83,4
83,5
83,4
83,5
83,4
83,5
83,5
83,5
83,5
83,5
83,5
83,5
83,5
 | 78,0
79,9
82,0
83,4
84,7
84,6
84,6
83,6
83,6
83,6
83,6
83,6
83,6
83,6
83
 | 73,
75,
77,
78,
79,
40,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 |
 | 25,9
25,6
25,6
25,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,6
23,7
24,6
25,7
24,6
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7
25,7 | |
 | |
| | | 77.5
76.8
74.6
72.9
71.3
70.2
69.7
69.8
09.4
69.4
69.4
69.4
69.4
69.4
69.4
69.4
6

 | 86.1
86.1
82.3
70.6
78.0
78.0
75.0
75.0
75.1
75.0
75.3
74.5
74.5
74.2
73.6
73.6
72.6
72.6

 | 91.4
91.4
98.1
84.1
82.0
81.0
82.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
 | 00.6 00.6 00.2 87.1 88.4 81.4 83.7 83.7 83.7 81.4 83.6 83.6 83.6
 | 84,2
86,0
87,4
87,4
87,9
87,3
86,6
85,4
85,4
85,4
85,4
85,4
85,4
83,5
83,5
83,5
83,5
83,5
83,5
83,5
83,5
 | 78,0
79,9
82,0
83,4
84,7
84,6
83,4
83,4
83,4
83,4
83,6
83,4
83,6
83,4
83,6
83,4
83,6
83,4
83,6
83,4
83,6
83,4
83,6
83,4
83,6
83,4
83,6
83,6
83,6
83,6
83,6
83,6
83,6
83,6
 | 73,
75.
77.
78.
79.
79.
79.
79.
79.
79.
79.
79.
79.
79 |
 | 25,9
25,5
5,5
3,6
3,6
3,7
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
2,0
1,2
1,2
1,2
1,2
1,2
1,2
1,2
1,2 | |
 | |
| | 2550
3500
3500
4500
4500
4500
4500
6500
7500
8500
7500
8500
7500
8500
7500
8500
7500
8500
7500
8500
7500
8500
8 | 77.5
76.8
76.8
74.6
72.9
70.2
69.7
69.6
69.4
69.4
69.4
69.4
69.4
69.4
69.4

 | 86.1
86.1
82.3
78.8
77.6
76.0
75.6
75.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.6
93.6
98.1
98.1
98.7
84.1
92.7
92.7
92.7
92.7
92.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
93.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
 | Yo L. Yo
 | 84,2
86,0
87,4
87,4
87,9
87,3
86,6
85,6
85,6
85,6
85,6
85,6
85,6
85,6
 | 78,0
79,9
82,0
83,4
83,4
84,7
84,7
84,6
83,4
83,6
83,6
83,2
83,6
83,6
83,6
81,6
91,6
91,6
91,2
81,1
81,0
80,9
80,9
 | 73,
75,
77,
79,
40,
80,
79,
79,
79,
79,
79,
79,
79,
74,
79,
76,
76,
76,
76, |
 | 25,9
25,5
25,5
25,5
25,6
25,6
25,6
22,7
22,0
21,2
22,0
21,2
22,0
21,2
22,0
21,2
22,0
21,2
22,0
21,2
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,7
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6
20,6 | |
 | |
| | 230.0
35.0
45.0
45.0
45.0
65.0
75.0
85.0
75.0
85.0
75.0
85.0
75.0
85.0
10.0
15.0 | 77.8
76.8
76.8
77.9
71.3
70.2
69.7
69.6
69.7
69.6
69.7
69.6
69.7
69.6
69.7
69.7

 | 86.1
86.1
82.3
70.6
78.8
77.6
76.0
75.0
75.3
74.5
74.5
74.5
74.5
73.0
72.6
73.0
72.6

 | 91.4
91.4
84.1
84.1
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
 | Vin 1 Vin 6 Vin 7
 | 84,2
86,0
87,4
87,9
87,9
87,9
87,9
85,6
85,6
85,6
85,6
85,6
85,6
85,1
84,1
83,1
83,1
83,1
83,1
83,1
83,1
83,1
83
 | 78,0
79,9
82,0
83,4
84,7
84,4
83,6
83,6
83,6
83,6
83,6
83,6
83,6
83,6
 | 73,
75,
77,
79,
79,
79,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 |
 | 25, 9
55, 5
55, 5
75, 6
72, 6
72, 7
72, 7
72, 6
72, 7
72, 7
72, 6
72, 7
72, 6
72, 7
70, 6
70, 70, 70, 70, 70
70, 70, 70, 70
70, 70, 70, 70, 70
70, 70, 70, 70, 70, 70, 70, 70, 70, 70, | | |
 |
| | 2000
30.0
30.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0
50.0 | 77.5
76.8
76.8
74.6
72.9
70.2
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.5
65.5
65.5
65.5
65.5
65.5
65.5
65.5

 | 86.1
86.1
82.3
78.0
78.0
77.0
76.0
76.0
76.0
76.0
76.0
76.0
76

 | 91.4
92.4
93.4
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.1
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
94.7
 | Vor A Vor A Vor A Born A AB A
 | 84,2
86,0
87,4
87,8
87,8
87,9
87,9
85,6
85,2
85,2
85,2
85,1
84,1
83,6
83,4
83,5
83,4
83,5
83,4
83,5
83,4
83,5
83,5
83,5
83,5
83,5
83,5
83,5
83,5
 | 78,0
79,9
82,0
83,4
84,7
84,6
84,7
84,6
83,4
84,7
84,6
83,6
83,6
83,6
83,6
83,6
83,6
83,6
83
 | 73,
75,
77,
78,
78,
78,
78,
78,
78,
78,
79,
79,
79,
79,
78,
77,
78,
76,
76,
76,
76,
76,
76,
76,
76,
76,
76 |
 | 25, 9
55, 5
55, 5
55, 6
55, 7
55, 7
55 | | |
 |
| | 23.0
30.0
35.0
45.0
55.0
75.0
75.0
75.0
85.0
75.0
85.0
10.0
10.0
10.0
10.0
10.0
10.0
10.0
1 | 77.5
76.8
76.8
74.6
72.9
69.7
69.6
69.4
69.4
69.4
69.4
69.4
69.4
69.4

 | 86.1
86.1
82.3
78.6
78.6
76.7
76.7
76.7
75.6
75.6
75.6
75.6
75

 | 91.6
92.7
98.1
98.1
98.1
98.1
92.0
91.6
91.6
91.6
91.6
91.6
91.6
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
91.7
 | Vir No Vor No Vor No 80.6 Vor 87.1 88.4 83.4 85.2 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.6 83.4 83.7 83.4 83.6 83.4 83.7
 | 84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 83.3 82.3 81.5 NPUST = 2 XNU SUU5
 | 78.0
79.9
82.0
83.4
83.4
83.4
83.6
83.6
83.6
83.6
83.6
81.6
81.6
81.6
81.7
61.2
81.1
61.2
81.1
61.2
81.1
61.2
81.1
91.0
90.0
90.0
90.0
90.0
90.0
90.0
9
 | 73,
75,
77,
78,
79,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
75,
76,
76,
76,
76,
76,
76,
76,
76,
76,
76 |
 | 25, 9
25, 5
25, 6
25, 6
25, 6
25, 6
25, 7
22, 7
22, 7
22, 7
22, 7
22, 7
24, 6
24, 7
24, 6
24, 6
24, 7
24, 6
24, 7
24, 6
24, 7
24, 7
24 | 2745.6 | 5571.3 | 11142.5
 |
| 1 | 230.0
35.0
45.0
55.0
55.0
70.0
75.0
75.0
75.0
75.0
10.0
10.0
10.0
10.0
10.0
10.0
10.0
1 | 77.6
76.8
76.8
76.8
76.2
70.2
69.7
69.6
69.7
69.6
69.7
69.6
69.7
69.6
69.7
69.6
69.7
69.6
69.7
69.7

 | 86.1
86.1
82.3
78.8
77.6
76.0
75.4
75.3
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.4
91.4
70.9
98.1
84.1
82.7
82.7
82.7
82.7
83.7
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
70.5
 | Vor R Vor R Vor R 80.6 Vor 87.1 R RA R R R R R R R <tr< td=""><td>84,2 86,0 87,4 87,8 87,8 87,8 87,8 87,9 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,7 83,3 62,7 81,5 81,5 HPUST = 2 AND 500 6,00,1 6,07,73</td><td>78.0
79.9
82.0
83.4
83.4
84.7
84.6
83.6
83.6
83.6
83.6
83.6
83.6
83.6
83</td><td>73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79</td><td>LEVELS
707.0 1
45.1A</td><td>25, 5
25, 6
25, 6
25, 6
25, 6
25, 7
25, 7
22, 7
22, 7
22, 7
20, 6
22, 7
20, 6
20, 7
20, 6
20, 7
20, 7
20</td><td>27A5.6
16.13</td><td><u>4571.3</u>
-18.35</td><td>11142.5</td></tr<> | 84,2 86,0 87,4 87,8 87,8 87,8 87,8 87,9 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,7 83,3 62,7 81,5 81,5 HPUST = 2 AND 500 6,00,1 6,07,73

 | 78.0
79.9
82.0
83.4
83.4
84.7
84.6
83.6
83.6
83.6
83.6
83.6
83.6
83.6
83
 | 73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707.0 1
45.1A | 25, 5
25, 6
25, 6
25, 6
25, 6
25, 7
25, 7
22, 7
22, 7
22, 7
20, 6
22, 7
20, 6
20, 7
20, 6
20, 7
20, 7
20 | 27 A 5.6
16.13
 | <u>4571.3</u>
-18.35 | 11142.5 |
| | 2230.0
30.0
45.0
35.0
35.0
35.0
60.0
80.0
80.0
95.0
95.0
95.0
15.0
15.0
15.0
15.0
15.0
15.0
15.0
1 | 775
76.8
76.8
76.8
76.8
76.2
60.7
60.7
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.7

 | 86.1
86.1
82.3
78.6
77.6
76.0
76.7
76.0
76.1
75.6
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.4 92.4 84.1 84.1 82.0 84.1 82.0 84.1 82.0 84.1 82.0 81.0 81.0 80.1 70.5 70.5 70.5 76.7 80.7 80.7 80.7 80.7 90.7 11.1 50.85 60.35
 | Y0 th Y1 th
 | 84,2 86.0 87.4 87.8 87.8 87.9 86.6 85.8 85.4 85.4 85.4 85.4 85.8 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.5 82.3 81.9 81.5 HPUST = 2 NNO SUU0 609.1 8 67.73 071.05
 | 78.0
79.9
82.0
83.4
84.7
84.6
83.4
83.6
83.6
83.6
83.6
83.6
83.6
83.6
83.6
 | 73,
75,
77,
78,
78,
78,
78,
78,
74,
78,
79,
79,
79,
79,
79,
79,
74,
79,
76,
76,
76,
76,
76,
76,
76,
76,
76,
76 | LEVELS
707.01
45.1A
 | 73,0
13,6
13,6
14,1
13,6
14,1
13,6
14,1
14,1
14,8
14,8
14,8
14,8
14,8
14,8
14,8
14,8
14,8
14,8
14,8
14,1
14,8
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1
14,1 | 27 A 5.6 | 4571.3
 | 11142.5
 |
| 1
1
1
1
1
1
1
1
1
1
1
1
1
1
1
1
1
1
1 | | 77.5
76.5
76.6
77.6
76.7
76.7
69.6
69.7
69.6
69.4
69.4
69.4
69.4
69.4
69.4
69.4

 | 86.1
86.1
82.3
78.6
77.6
76.7
76.7
75.6
75.6
75.6
75.6
75

 | 91.4
91.4
70.7
84.1
84.1
82.0
81.6
81.6
81.6
81.6
81.6
81.6
81.6
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
 | Vor R Vor R Vor R 80.6 Vor 87.1 R 88.4 R 83.6 R 83.7 R 83.7 R 83.7 R 83.6 R 83.7 R 83.6 R 83.7 R <
 | 84,2
86,0
87,4
87,8
87,9
87,3
86,6
85,6
85,6
85,6
85,6
85,6
85,6
83,1
83,1
83,1
83,3
83,3
83,3
82,7
82,3
81,5
81,5
81,5
81,5
81,5
81,5
81,5
81,5
 | 78.0
79.9
82.0
83.4
83.4
83.4
83.7
84.6
84.7
84.7
84.7
84.7
84.7
84.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81
 | 73,
75,
77,
78,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
75,
75,
76,
76,
76,
76,
76,
76,
76,
76,
76,
76 | LEVELS
707-0 1
45,14
 | 73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0 | 27 A 5.6
16.13
27.50
3.4.70 | 5571.3
-10.35
-12.42
 | |
| 579
3300
201 | | 77.5
76.8
76.8
76.8
76.8
77.9
69.8
69.7
69.6
69.4
69.4
69.4
69.4
69.4
69.4
69.4

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.0
75.3
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
75.5
74.5
74.5
74.5
75.5
74.5
74.5
74.5
74.5
74.5
75.5
74.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
74.5
75.5
75.5
74.5
75.5
75.5
74.5
75.5
75.5
74.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5

 | 91.4
91.4
P8.1
P8.7
84.1
P8.7
84.1
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
P8.7
 | Yu La Yu Y
 | 84,2 86.0 87.4 87.8 87.4 87.9 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.2 85.6 85.3 83.3 62.7 83.3 62.7 83.3 81.5 81.5 93.4 83.3 62.7 81.5 93.4 83.3 62.7 81.5 93.4 83.3 62.7 81.5 93.4 83.3 93.4 83.3 93.4 83.5 93.4 83.5 93.4 83.5 93.4 83.5 93.4 83.5 93.4 83.5 93.4 83.5 93.5 93.6 93.4 93.7 93.7 93.7 93.7 93.7 93.7 93.7 <td>78.0
79.9
82.0
83.4
83.4
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84</td> <td>73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79</td> <td>LEVELS
707.0 1
45,14
50,70
60,70
60,70
60,70
60,70
60,70
60,70
60,70</td> <td>27,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175</td> <td>2745_6
16.13
27.30
3.4,70
41.02</td> <td>4571.3
</td> <td></td>
 | 78.0
79.9
82.0
83.4
83.4
84.7
84.7
84.7
84.7
84.7
84.7
84.7
84
 | 73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707.0 1
45,14
50,70
60,70
60,70
60,70
60,70
60,70
60,70
60,70
 | 27,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175,0
175 | 2745_6
16.13
27.30
3.4,70
41.02 | 4571.3

 | |
| 579
579
300
243 | 225.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 775
768
768
768
768
768
768
768
668
667
668
668

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.0
76.0
76.0
76.0
76.0
76.0
76

 | 91 6 93 1 98 1 98 1 98 1 98 1 92 6 91 6 91 6 91 6 91 6 91 6 91 6 91 6 91 7 91 7 91 7 92 7 93 7 11 1 93 7 11 1 93 7 11 1 93 7 11 1 93 8 93 7
 | Vir 70 00.6 90.2 88.1 86.9 87.1 86.9 87.1 86.9 83.7 83.6 83.8 83.6 83.4 83.6 81.9 81.6 81.9 81.6 81.9 81.6 81.6 80.6 80.6 80.6 FULL SCALE T 7 UCTAVE 8 66.0 70.7 A8.51 70.7 A8.51 70.7 A8.51 70.7 A7.54 71.7 66.0 73.7
 | 84,2 86.0 87.4 87.8 87.9 87.9 87.9 85.8 85.2 85.4 85.4 85.4 85.8 85.9 <td>78.0
79.9
82.0
83.4
84.7
84.7
84.7
84.7
84.7
84.7
83.6
83.6
83.6
83.6
83.6
81.6
81.6
81.5
81.5
81.5
81.2
81.2
81.2
81.2
81.2
81.2
81.2
81.2</td> <td>73, 75, 77, 77, 77, 77, 77, 77, 77, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 77, 77, 76, 77, 76, 77, 76, 76, 76, 356, 6 52, 84 77, 39 61, 10 65, 17 70, 85, 70, 85</td> <td>LEVELS</td> <td>73,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0</td> <td>27A5.6</td> <td>5571.3
5571.3
10.35
.44
12.42
71.75
.7.50
</td> <td>11142.5
-77.98
-45.41
-25.24
-10.54
-7.98</td>
 | 78.0
79.9
82.0
83.4
84.7
84.7
84.7
84.7
84.7
84.7
83.6
83.6
83.6
83.6
83.6
81.6
81.6
81.5
81.5
81.5
81.2
81.2
81.2
81.2
81.2
81.2
81.2
81.2
 | 73, 75, 77, 77, 77, 77, 77, 77, 77, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 74, 77, 77, 76, 77, 76, 77, 76, 76, 76, 356, 6 52, 84 77, 39 61, 10 65, 17 70, 85, 70, 85 | LEVELS
 | 73,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0 | 27 A 5.6 | 5571.3
5571.3
10.35
.44
12.42
71.75
.7.50

 | 11142.5
-77.98
-45.41
-25.24
-10.54
-7.98 |
| 579
310
310
201
201
212 | 225.0
30.0
45.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0
55.0 | 77.5
76.8
74.6
72.9
76.3
77.3
70.2
60.7
60.6
60.4
60.4
60.4
60.4
60.4
60.4
60.4

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.7
76.7
75.6
75.6
75.6
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.4
91.4
70.7
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
98.1
97.1
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
97.5
 | Vor 1 Vor 0 Vor 0 80 0 87 1 88 1 87 1 88 1 87 1 87 1 87 1 83 1 1 1 1 1 1 1
 | 84,2
86,0
87,4
87,8
87,9
87,3
86,6
85,6
85,6
85,6
85,6
85,6
85,6
85,6
 | 78.0
79.9
82.0
83.4
83.4
83.7
84.7
84.7
84.7
84.7
83.6
83.6
83.6
83.6
81.6
82.6
82.6
82.6
81.7
81.7
81.2
81.1
81.7
81.2
81.1
91.0
80.0
9000_000
 | 73,
75,
77,
78,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707-0
1
45,14
53,70
74,95
74,95
74,95
74,95
74,95
74,95
74,95
74,95
75
74,95
75
74,95
75
74,95
75
75
75
75
75
75
75
75
75
75
75
75
75 | 73,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
13,0
14,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0
15,0 | 27745.6
16.13
27.34
70
41.02
41.02
41.02
41.02
51.29 |
4571.3
+18.35
-12.42
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49
-7.49 | 11142.5
-77.98
-25.24
-10.59
-10.59
-13.04 |
| 579
300
201
579
1
1
1
1
1
1
1
1
1
1
1
1
1 | 225.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 77.6 76.8 76.8 76.8 76.8 76.8 76.8 76.8 60.7 60.7 60.7 60.7 60.7 60.7 60.7 60.7 60.6 60.7 60.6 60.7 60.6 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 71.41 70.41 70.41 70.41 70.41 70.41 70.41 70.41 70.41 70.41

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.7
76.7
75.6
75.3
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
75.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
73.6
73.6
73.6
73.6
73.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6
76.6

 | 91.4
91.4
PB21
PR.7
B4.1
PB27
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B247
B441
B247
B441
B247
B441
B247
B441
B247
B441
B447
B441
B447
B447
B447
B447
B447
B447
B447
B447
B447
B447
B447
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B477
B4777
B4777
B4777
B4777
B4777
B4777
B4777
B4777
B4777
B47777
B47777
B47777777
B47777777777
 | Yu L Yu A
 | 84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 85.6 85.7 85.8 83.3 82.7 83.4 83.5 83.5 81.5 81.5 81.5 81.5 81.5 81.7 <td>78.0
79.9
82.0
83.4
83.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
83.7
83.7
83.7
83.7
83.7
81.2
81.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0</td> <td>73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79</td> <td>LLVELS
707.0 1
45,1A
50.70
74,47
707.0 1
45,1A
50.70
70
74,47
70
74,58</td> <td>75,0
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5</td> <td>2785.6
16.13
27.50
34.70
41.02
41.02
41.02
41.02
51.29
52.62</td> <td>4571.3
-18.35
-12.42
-77.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53</td> <td>11142.5
-77.08
-5.21
-25.21
-7.08
-10.59
-10.59
-10.05</td>
 | 78.0
79.9
82.0
83.4
83.7
84.7
84.7
84.7
84.7
84.7
84.7
84.7
83.7
83.7
83.7
83.7
83.7
81.2
81.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
 | 73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LLVELS
707.0 1
45,1A
50.70
74,47
707.0 1
45,1A
50.70
70
74,47
70
74,58
 | 75,0
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5
15,5 | 2785.6
16.13
27.50
34.70
41.02
41.02
41.02
41.02
51.29
52.62 | 4571.3
-18.35
-12.42
-77.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
 | 11142.5
-77.08
-5.21
-25.21
-7.08
-10.59
-10.59
-10.05 |
| 579
579
579
579
579
579
579
579
579
579 | 225.4
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0
45.0 | 77.5 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 66.7 66.7 66.6 66.3 66.7 66.6 66.3 66.4 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.7 71.4 70.4 70.4 70.4 70.4 70.4 70.4 70.4 70.4 70.4 70.4 70.4 70.4

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.0
75.6
76.7
76.7
76.7
76.7
75.6
74.5
74.5
74.5
74.5
74.5
73.6
73.6
73.6
72.6
73.6
72.6
73.6
72.6
73.6
72.6
72.6
73.6
72.6
73.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72

 | 91 6 93 6 94 1 98 1 98 1 98 1 92 6 91 6 91 6 91 6 91 6 91 6 91 6 91 6 91 7 91 7 91 7 92 7 93 7 10 7 11 1 93 7 11 1 93 7 11 1 93 7 11 1 93 7 93 7 93 7 93 7 93 7 93 7
 | Vir R Vir R Vir R 80.6 Y 86.7 R R <
 | 84,2 86.0 87.4 87.4 87.4 87.4 85.8 85.8 85.8 85.8 83.4 83.4 83.4 83.4 83.4 83.5 82.3 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 81.5 74.4.1 74.1 74.1 73.1 374.2 73.4 73.4
 | 78.0
79.9
82.0
83.4
84.7
84.7
84.7
84.7
84.6
83.6
83.6
83.6
83.6
83.6
81.6
81.6
81.5
81.7
81.1
81.0
80.9
80.9
80.9
80.9
80.9
80.9
80.9
80
 | 73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707-0 1
45,1A
50,70
707-0 1
45,1A
50,70
74,95
41,20
64,61
71,39
64,73
65,73
64,61
 | 73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,00,00
74,000,000,000,000,000,000,000,000,000,0 | 27785.6
16.13
27.26
34.70
41.02
51.79
52.87
52.87
52.41
15 |
5571.3
-19.35
-19.35
-19.44
-12.42
-7.75
-7.50
-7.75
-7.50
-7.75
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
-7.51
- | 111142.5
-77.98
-77.98
-77.98
-77.98
-77.98
-77.98
-75.24
-27.24
-27.24
-7.95
-13.05
-13.05
-13.05
-13.05 |
| 579
579
310
300
201
201
212
212
160
173
173 | | 77.5
76.8
76.8
76.8
76.8
77.9
76.2
60.7
60.7
60.6
60.4
60.4
60.4
60.4
60.4
60.4
60.4

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.7
76.7
75.6
75.6
75.6
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91.4
91.4
70.7
84.1
84.1
82.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
81.0
 | 00.6 00.6 00.2 87.1 88.4 81.4 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.7 83.6 83.7
 | 84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 83.3 82.7 82.7 82.7 82.7 82.7 82.7 81.5 1.9 81.5 1.9 81.5 1.9 1.7.3 0.71.05 0.71.05 1.74.18 1.74.18 7.74.18 7.73.40 73.240 73.240
 | 78.0
79.9
82.0
83.4
83.4
83.7
84.7
84.7
84.7
84.7
84.7
84.7
83.7
83.7
83.7
83.7
81.7
81.7
81.7
81.7
81.7
81.7
81.7
81
 | 73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707-0 1
45,14
5,707-0 1
45,14
5,707-0
1
45,14
5,707-0
1,6
4,9
5,73
1,6
4,9
5,73
1,6
4,9
5,73
1,6
4,9
5,73
1,6
4,9
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,73
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,7
5,75
1,75
1 | 73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0 | 27785.6
16.13
27.34
16.23
41.02
41.02
41.02
41.02
51.29
52.02
52.02
54.11
54.36 |
5571.3
=18.35
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42
-12.42 | 11147.5
-77.98
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
-25.41
- |
| 579
579
300
201
10
11
11
11
11
11
11
11
11 | 225.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 77.5 76.8 76.8 76.8 76.8 76.8 76.8 76.8 60.7 60.7 60.6 60.7 60.6 66.7 66.6 66.7 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.7 71.4170.65 72.4773.65 74.477.77 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 74.477.9 <tr<
td=""><td>86.1
86.1
82.3
82.3
78.6
77.6
76.0
75.6
75.6
75.6
75.7
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
75.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.7
72.7
75.6
72.7
72.7
75.6
72.7
75.6
72.7
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.7
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7</td><td>91.4
91.4
70.7
84.1
84.1
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0</td><td>00.6 00.6 00.2 88.4 88.4 87.1 88.4 83.6 83.7 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.6 83.6 83.7 83.6 83.7 83.6 83.7 83.7</td><td>84,2 86.0 87.4 87.4 87.3 86.6 85.2 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 83.3 82.7 83.4 83.5 82.7 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 9.1 8 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.2 <!--</td--><td>78.0
79.9
82.0
83.4
83.0
84.4
84.7
84.6
83.6
83.6
83.6
83.6
83.6
83.6
81.6
91.6
91.6
91.6
91.6
91.6
91.6
91.7
81.0
82.0
81.1
81.0
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9</td><td>73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td><td>LUVELS
707.0
1
45.1A
50.70
61.70
61.70
61.70
61.70
61.70
61.73
61.70
61.73
61.70
61.73
61.74
61.70
61.73
61.74
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
75
75
75
75
75
75
75
75
75
75
75
75
7</td><td>173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
17</td><td>2785.6
10.13
27.50
34.70
41.02
44.00
51.29
52.49
54.11
54.26
51.29
52.49
54.11
54.26</td><td>4571.3
-18.35
-18.35
-12.42
-77.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-12.42
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53</td><td>11142.5
-77.48
-5.41
-75.45
-77.48
-5.41
-7.48
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59

-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.</td></td></tr<> | 86.1
86.1
82.3
82.3
78.6
77.6
76.0
75.6
75.6
75.6
75.7
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
75.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.7
72.7
75.6
72.7
72.7
75.6
72.7
75.6
72.7
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.7
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.6
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7

 | 91.4
91.4
70.7
84.1
84.1
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
82.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
80.0
 | 00.6 00.6 00.2 88.4 88.4 87.1 88.4 83.6 83.7 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.6 83.6 83.7 83.6 83.7 83.6 83.7 83.7 | 84,2 86.0 87.4 87.4 87.3 86.6 85.2 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 83.3 82.7 83.4 83.5 82.7 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 9.1 8 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.7 9.2 </td <td>78.0
79.9
82.0
83.4
83.0
84.4
84.7
84.6
83.6
83.6
83.6
83.6
83.6
83.6
81.6
91.6
91.6
91.6
91.6
91.6
91.6
91.7
81.0
82.0
81.1
81.0
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9</td> <td>73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td> <td>LUVELS
707.0
1
45.1A
50.70
61.70
61.70
61.70
61.70
61.70
61.73
61.70
61.73
61.70
61.73
61.74
61.70
61.73
61.74
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
75
75
75
75
75
75
75
75
75
75
75
75
7</td> <td>173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
17</td> <td>2785.6
10.13
27.50
34.70
41.02
44.00
51.29
52.49
54.11
54.26
51.29
52.49
54.11
54.26</td> <td>4571.3
-18.35
-18.35
-12.42
-77.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-12.42
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53</td>
<td>11142.5
-77.48
-5.41
-75.45
-77.48
-5.41
-7.48
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.</td> | 78.0
79.9
82.0
83.4
83.0
84.4
84.7
84.6
83.6
83.6
83.6
83.6
83.6
83.6
81.6
91.6
91.6
91.6
91.6
91.6
91.6
91.7
81.0
82.0
81.1
81.0
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
9
 | 73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LUVELS
707.0
1
45.1A
50.70
61.70
61.70
61.70
61.70
61.70
61.73
61.70
61.73
61.70
61.73
61.74
61.70
61.73
61.74
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
61.75
75
75
75
75
75
75
75
75
75
75
75
75
7 | 173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
17 | 2785.6
10.13
27.50
34.70
41.02
44.00
51.29
52.49
54.11
54.26
51.29
52.49
54.11
54.26 |
4571.3
-18.35
-18.35
-12.42
-77.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-12.42
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53
-7.53 | 11142.5
-77.48
-5.41
-75.45
-77.48
-5.41
-7.48
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10.59
-10. |
| 579
579
300
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
320
243
243
243
24
24
24
24
24
24
24
24
24
24
24
24
24 | 225.0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 | 77.5 76.8 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 64.7 64.6 64.7 64.6 65.7 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.7 71.41 70.61 75.21 73.61 74.7 7.91 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71 87.41 70.71

 | 86.1
86.1
82.3
82.3
78.6
77.6
75.6
75.6
75.6
74.5
74.5
74.5
74.5
74.5
74.5
73.6
72.6
72.6
73.0
72.6
72.2
9
9
9
9
9
9
9
9
9
9
9
9
9

 | 91 6 93 6 98 11 84.1 8 84.1 8 84.1 8 84.1 8 84.1 8 84.1 8 84.1 8 84.1 8 81.6 8 81.7 8 81.6 8 81.7 8
 | Vor R Vor R Vor R 80.6 Vor 87.1 R RA R RA S S S S S S S S S S S S S S S S S S S S S S S S S
 | 84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 85.8 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.5 82.7 81.5 81.5 81.5 81.5 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 <t< td=""><td>78.0 79.9 79.9 82.0 83.4 83.4 78.9 84.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 81.6 81.6 81.7 81.7 81.8 80.9 00000.000 000 00000.000 9 00000.000 71.19 77.19 71.19 73.74 73.74 73.74 73.74 73.74 73.74</td><td>73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79</td><td>LLVELS
707-0 1
45,14
50,70
707-0
1
45,14
50,70
51,71
50,70
51,71
50,70
51,73
50,70
51,73
50,70
51,73
50,70
51,73
50,70
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
5</td><td>73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0</td><td>27755.6
16.13
27.70
34.00
34.00
51.70
52.67
52.47
54.11
52.47
54.15
52.65
52.65
52.65
52.65</td><td>4571.3
=18.35
=18.35
-12.42
-7.40
-7.40
-12.42
-7.40
-12.42
-7.40
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-1</td><td>11142.5
-77.48
-77.48
-75.41
-25.24
-10.59
-7.05
-13.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.0
5
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.0</td></t<> | 78.0 79.9 79.9 82.0 83.4 83.4 78.9 84.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 81.6 81.6 81.7 81.7 81.8 80.9 00000.000 000 00000.000 9 00000.000 71.19 77.19 71.19 73.74 73.74 73.74 73.74 73.74 73.74
 | 73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LLVELS
707-0 1
45,14
50,70
707-0
1
45,14
50,70
51,71
50,70
51,71
50,70
51,73
50,70
51,73
50,70
51,73
50,70
51,73
50,70
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
50,73
5 | 73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0 | 27755.6
16.13
27.70
34.00
34.00
51.70
52.67
52.47
54.11
52.47
54.15
52.65
52.65
52.65
52.65 |
4571.3
=18.35
=18.35
-12.42
-7.40
-7.40
-12.42
-7.40
-12.42
-7.40
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-1 | 11142.5
-77.48
-77.48
-75.41
-25.24
-10.59
-7.05
-13.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.0 |
| 579
579
380
201
217
217
217
217
217
217
217
217
217
21 | | 77.5
76.8
76.8
76.8
76.8
76.8
76.9
66.7
66.7
66.6
66.9
66.9
66.9
66.9

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.7
76.7
75.6
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5

 | 91. 4
91. 4
70. 7
84. 1
84. 1
82. 0
81. 0
81. 0
81. 0
82. 0
81. 0
81
 | Vor Cor Vor | 84,2 86.6 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 84.1 83.3 82.7 82.7 82.7 82.7 82.7 82.7 82.7 82.7 81.9 81.9 81.9 81.9 81.9 81.9 97.13 74.6 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94

 | 78.0
79.9
82.0
83.4
83.7
84.7
84.7
84.7
84.7
83.6
83.6
83.6
83.6
83.6
81.8
81.8
81.7
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.1
81.2
81.2
 | 73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707-0 1
45,14
5,707-0 1
45,14
5,707-0 1
45,14
5,707-0 1
45,14
5,707-0 1
6,707-0 1
7,64
6,77
7,16
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
6,73
7,39
7,39
7,39
7,39
7,39
7,39
7,39
7 |
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0 | 2785.6
16.13
27.5
34.70
41.92
44.92
54.11
54.36
54.13
54.36
54.13
54.13
54.14 | 4571.3
 |
11142.5
-77.48
-25.24
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
- |
| 579
579
300
201
1
31
31
21
21
21
21
21
21
21
21
21
21
21
21
21 | 2230.0
5.0
5.0
5.0
5.0
5.0
5.0
5.0 | 77.5
76.8
76.8
76.8
76.8
76.8
76.2
60.7
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.6
60.7
60.5
65.5
65.5
65.5
65.5
65.5
65.5
65.5

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.0
75.6
75.6
75.6
75.7
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
72.6
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
72.7
73.0
72.6
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.4
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
73.0
70.1
70.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
7 74.5
7 74.5
7 74.5
7 74.5
7 74.5
7 74.5
7 74.5
7 74.5
7 74.5
7 74

 | 91. c
91. c
70. c
84. 1
84. 1
82. n
81. c
81. c
81
 | Vor Port Vor Port Vor Port 80.6 Vor 80.6 Port 80.7 Port 80.6 Port 80.6 Port 90.7 | 84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 84.1 84.1 84.1 84.1 84.1 84.1 83.3 82.7 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.9 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 81.7 <t< td=""><td>78.0 79.9 79.9 82.0 83.4 83.4 84.6 83.4 84.6 83.4 83.6 83.2 83.6 83.2 83.6 81.6 81.6 81.6 81.7 91.7 81.1 81.8 81.1 81.9 81.1 81.0 80.0 90000.0000 9000 00000.0000 9000 00000.0000 9000 71.74 72.74 73.40 73.40 73.40 73.40 73.40 73.40</td><td>73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td><td>LEVELS
707.0 1
45.1A
5.7
707.0 1
45.1A
5.7
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0
1
45.1A</td><td>173,0
173,0
173,0
173,0
173,0
173,0
173,0
173,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
17</td><td>2785.6
16.13
27.56
34.70
41.62
44.10
51.79
52.49
51.79
52.49
53.11
54.36
54.57
54.14
54.57
54.16
54.57
54.16
54.57
54.16</td><td>4571.3
-18.35
-18.35
-242
-21.72
-7.53
-7.53
-19.42
-1.72
-7.53
-19.45
-41.65
-41.65
-41.65
-41.65
-41.65
-41.65
-41.65
-41.65
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-</td><td>11142.5
-77.98
-35.21
-77.98
-35.21
-77.98
-17.05
-17.05
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-</td></t<>
 | 78.0 79.9 79.9 82.0 83.4 83.4 84.6 83.4 84.6 83.4 83.6 83.2 83.6 83.2 83.6 81.6 81.6 81.6 81.7 91.7 81.1 81.8 81.1 81.9 81.1 81.0 80.0 90000.0000 9000 00000.0000 9000 00000.0000 9000 71.74 72.74 73.40 73.40 73.40 73.40 73.40 73.40
 | 73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707.0 1
45.1A
5.7
707.0 1
45.1A
5.7
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A
707.0 1
45.1A |
173,0
173,0
173,0
173,0
173,0
173,0
173,0
173,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
174,0
17 | 2785.6
16.13
27.56
34.70
41.62
44.10
51.79
52.49
51.79
52.49
53.11
54.36
54.57
54.14
54.57
54.16
54.57
54.16
54.57
54.16 | 4571.3
-18.35
-18.35
-242
-21.72
-7.53
-7.53
-19.42
-1.72
-7.53
-19.45
-41.65
-41.65
-41.65
-41.65
-41.65
-41.65
-41.65
-41.65
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
-41.75
- |
11142.5
-77.98
-35.21
-77.98
-35.21
-77.98
-17.05
-17.05
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
- |
| 579
579
579
579
579
579
579
579
579
579 | | 77.5 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 66.7 66.7 66.7 66.7 66.7 66.7 65.7 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.7 71.41 70.61 75.21 73.65 76.21 70.61 75.21 73.61 76.21 70.7 82.41 70.7 83.01 70.7 83.01 70.7 83.01 70.7 83.01 70.7 83.01 70.7 83.01 70.7 83.01 70.7 83.01 70.7 82.41 70.51

 | 86.1
86.1
82.3
82.3
78.6
77.6
75.6
75.6
75.6
74.5
74.5
74.5
74.5
74.5
73.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.6
72.5
72.6
72.6
72.5
72.6
72.5
72.5
72.6
72.5
72.5
72.5
72.6
72.5
72.5
72.5
72.6
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
72.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5
75.5

 | 91 6 93 6 94 1 98 1 98 1 98 1 92 6 91 6 91 6 91 6 91 6 91 6 91 6 91 6 91 7 91 7 91 7 91 7 92 7 93 7 11 1 93 7 11 1 93 7 11 1 93 6 94 6 94 6 94 6 94 6 94 6 94 6 94 6 94 6 94 6 94 <td>Vor Cor Vor Cor Vor Cor 80.6 Oo 87.1 BA BA Solar RA Solar</td> <td>84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.5 9 81.5 9 81.5 9 9 9 9 9 9 9 9 9 9 10 11 12 13 13 13<!--</td--><td>78.0 79.9 79.9 82.0 83.4 83.4 78.9 84.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.7 83.6 83.6 83.7 81.7 81.9 81.9 80.9 80.9 00000.000 000 00000.000 9 00000.000 9 00000.000 9 72.4 73.46 73.47 73.97 73.47 73.97 73.47 73.97 73.47 73.47</td><td>73,
75,
77,
79,
79,
79,
79,
79,
80,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td><td>LEVELS
707-0
1
45,14
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
51
51
51
51
51
51
51
51
51
51
51
51</td><td>73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0</td><td>27785.6
16.13
27.76
34.70
34.70
34.70
34.70
52.67
54.11
52.67
54.13
52.67
54.14
52.67
54.15
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75</td><td>4571.3
</td><td>11142.5
-77.48
-25.24
-10.55
-77.48
-25.24
-13.04
-13.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.</td></td> | Vor Cor Vor Cor Vor Cor 80.6 Oo 87.1 BA BA Solar RA Solar
 | 84,2 86.0 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.4 83.5 9 81.5 9 81.5 9 9 9 9 9 9 9 9 9 9 10 11 12 13 13 13 </td <td>78.0 79.9 79.9 82.0 83.4 83.4 78.9 84.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.7 83.6 83.6 83.7 81.7 81.9 81.9 80.9 80.9 00000.000 000 00000.000 9 00000.000 9 00000.000 9 72.4 73.46 73.47 73.97 73.47 73.97 73.47 73.97 73.47 73.47</td> <td>73,
75,
77,
79,
79,
79,
79,
79,
80,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td> <td>LEVELS
707-0 1
45,14
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
51
51
51
51
51
51
51
51
51
51
51
51</td> <td>73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0</td>
<td>27785.6
16.13
27.76
34.70
34.70
34.70
34.70
52.67
54.11
52.67
54.13
52.67
54.14
52.67
54.15
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75</td> <td>4571.3
</td> <td>11142.5
-77.48
-25.24
-10.55
-77.48
-25.24
-13.04
-13.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.</td> | 78.0 79.9 79.9 82.0 83.4 83.4 78.9 84.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.7 83.6 83.6 83.7 81.7 81.9 81.9 80.9 80.9 00000.000 000 00000.000 9 00000.000 9 00000.000 9 72.4 73.46 73.47 73.97 73.47 73.97 73.47 73.97 73.47 73.47

 | 73,
75,
77,
79,
79,
79,
79,
79,
80,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707-0 1
45,14
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
50,70
51
51
51
51
51
51
51
51
51
51
51
51
51 | 73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0 |
27785.6
16.13
27.76
34.70
34.70
34.70
34.70
52.67
54.11
52.67
54.13
52.67
54.14
52.67
54.15
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
52.63
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75
54.75 | 4571.3
 | 11142.5
-77.48
-25.24
-10.55
-77.48
-25.24
-13.04
-13.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7.05
-7. |
| 579
579
334
34
34
34
34
34
34
34
34
34
34
34
34 | | 77.5
76.8
76.8
76.8
76.8
76.8
76.8
76.8
66.7
66.7

 | 86.1
86.1
82.3
82.3
78.6
77.6
76.0
75.6
75.6
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
74.5
75.6
72.6
72.6
73.0
75.6
76.6
75.6
75.6
72.6
75.6
72.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6
75.6

 | 91. 4
91. 4
70. 2
98. 1
84. 1
82. 0
81. 0
82. 0
81. 0
82. 0
82. 0
81. 0
82. 0
81. 0
81
 | 00.6 00.6 00.2 87.1 88.4 81.4 83.6 83.7 | 84,2 86.6 87.4 87.4 87.4 87.4 87.4 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.6 85.7 84.1 83.3 82.7 82.7 82.7 82.7 82.7 82.7 81.5 1.5 XNU 6.1.5 XNU 5.00 71.73 74.6 73.74.73 74.73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94 73.94<

 | 78.0 79.9 82.0 83.4 83.7 84.7 84.7 84.7 84.7 84.7 84.7 83.7 83.7 83.7 83.7 83.7 83.7 81.8 81.8 81.7
 | 73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707-0 1
45,14
5,707-0 1
45,14
5,707-0 1
45,14
5,707-0 1
45,14
5,707-0 1
7,6,73
6,70
7,74
6,77
7,16
6,73
7,19
6,73
7,19
6,73
7,74
6,73
7,74
6,73
7,74
6,73
7,74
6,73
7,74
6,73
7,74
7,75
6,73
7,75
7,74
7,75
7,74
7,75
7,75
7,75
7,75
 | 75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0
75,0 | 2785.6
16.13
27.35.6
16.13
27.70
41.92
41.92
41.92
54.11
54.5
54.13
54.13
54.13
54.13
54.13
54.14
54.15
54.14
54.15
54.14
54.15
54.14
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
54.15
55.15
54.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.15
55.1 | 4571.3
4571.3
418.35
-12.42
-71.75
-71.75
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-12.42
-71.75
-7.45
-71.75
-7.45
-71.75
-7.45
-71.75
-7.45
-71.75
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45
-7.45 |
11142.5
-77,48
-25,24
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
-10.54
- |
| 579
579
300
201
201
201
201
201
201
201
201
201
2 | | 77.6 76.8 76.8 76.8 76.8 76.8 76.8 76.8 76.8 60.7 60.7 60.6 60.7 60.6 66.7 66.7 67.3 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.7 71.4 70.4 71.7 71.7 71.7 71.7 72.4 70.7 70.4 70.7 70.7 70.7 70.7 70.7 70.7 70.7 70.7 70.7 70.7 70.7

 | 86.1 86.1 82.3 82.4 78.6 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 76.7 72.6 72.7 72.7 72.7 72.7 72.7 73.0 72.7 72.7 73.0 72.7 73.0 72.7 73.0 72.7 73.0 72.7 73.0 72.7 73.0 73.0 72.7 73.0 73.0 73.0 73.0 73.0 73.0 73.0 73.0 73.0 73.0 73.0 74.7 75.8 77.8 70.6 70.5 70.5 70.5 <td>91. c
91. c
70. c
70. c
84. 1
84. 1
82. n
81. c
81. c
81</td> <td>Vor Nor Vor Nor Vor Nor 80.6 Vor 80.6 Nor 80.7 Nor 80.6 Nor 80.6 Nor 80.6 Nor 80.6 Nor 90.7 Nor 80.6 Nor 90.7 Nor 80.6 Nor 90.7 Nor</td> <td>84,2 86,0 87,4 87,8 87,4 87,3 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,7 84,1 84,1 84,1 84,1 84,1 84,1 84,1 84,1 83,3 62,7 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 9 7,1,0 9 7,2,0 9 7,2,0 <</td> <td>78.0 79.2 79.9 82.0 83.4 83.4 84.6 83.4 83.6 83.2 83.6 83.2 83.6 81.6 81.6 81.6 81.7 81.8 81.8 81.8 81.8 81.8 81.9 81.9 81.1 81.0 80.00 80.00 81.1 81.0 80.00 80.00 80.00 91.2 81.1 81.0 80.00 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 90000 900000.0000<!--</td--><td>73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td><td>LEVELS
707.0 1
45.1A
50.70
1
45.1A
61.70
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.75
75
75
75
75
75
75
75
75
75
75
75
75
7</td><td>173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
17</td><td>2785.6
16.13
27.50
34.70
51.79
52.42
54.11
54.36
51.79
52.42
54.14
54.57
54.14
54.57
54.14
54.57
54.14
54.57
54.14
54.57
54.14</td><td>4571.3
-10.35
-10.35
-2.42
-7.50
-7.50
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.55
-1.72
-1.72
-1.75
-1.72
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.</td><td>11142.5
-77.98
-35.21
-77.98
-35.21
-77.98
-10.59
-10.59
-10.59
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27
.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-</td></td> | 91. c
91. c
70. c
70. c
84. 1
84. 1
82. n
81. c
81. c
81
 | Vor Nor Vor Nor Vor Nor 80.6 Vor 80.6 Nor 80.7 Nor 80.6 Nor 80.6 Nor 80.6 Nor 80.6 Nor 90.7 Nor 80.6 Nor 90.7 Nor 80.6 Nor 90.7 Nor | 84,2 86,0 87,4 87,8 87,4 87,3 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,7 84,1 84,1 84,1 84,1 84,1 84,1 84,1 84,1 83,3 62,7 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 81,9 9 9 7,1,0 9 7,2,0 9 7,2,0 <

 | 78.0 79.2 79.9 82.0 83.4 83.4 84.6 83.4 83.6 83.2 83.6 83.2 83.6 81.6 81.6 81.6 81.7 81.8 81.8 81.8 81.8 81.8 81.9 81.9 81.1 81.0 80.00 80.00 81.1 81.0 80.00 80.00 80.00 91.2 81.1 81.0 80.00 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 9000 90000.0000 90000 900000.0000 </td <td>73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79</td> <td>LEVELS
707.0 1
45.1A
50.70 1
45.1A
61.70
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.75
75
75
75
75
75
75
75
75
75
75
75
75
7</td> <td>173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
17</td> <td>2785.6
16.13
27.50
34.70
51.79
52.42
54.11
54.36
51.79
52.42
54.14
54.57
54.14
54.57
54.14
54.57
54.14
54.57
54.14
54.57
54.14</td>
<td>4571.3
-10.35
-10.35
-2.42
-7.50
-7.50
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.55
-1.72
-1.72
-1.75
-1.72
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.</td> <td>11142.5
-77.98
-35.21
-77.98
-35.21
-77.98
-10.59
-10.59
-10.59
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-</td> | 73,
75,
77,
79,
40,
40,
40,
79,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707.0 1
45.1A
50.70 1
45.1A
61.70
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.73
61.75
75
75
75
75
75
75
75
75
75
75
75
75
7
 | 173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
173.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
174.0
17 | 2785.6
16.13
27.50
34.70
51.79
52.42
54.11
54.36
51.79
52.42
54.14
54.57
54.14
54.57
54.14
54.57
54.14
54.57
54.14
54.57
54.14 | 4571.3
-10.35
-10.35
-2.42
-7.50
-7.50
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-7.50
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.72
-1.55
-1.72
-1.72
-1.75
-1.72
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1.75
-1. |
11142.5
-77.98
-35.21
-77.98
-35.21
-77.98
-10.59
-10.59
-10.59
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
-27.00
- |
| 579
579
579
579
579
579
579
579
579
579 | | 77.5 76.8 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 74.6 64.7 64.6 64.7 64.6 65.7 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.5 65.7 71.4170.61 75.2173.61 74.4717.97 74.4717.97 74.4717.97 74.4717.97 74.4717.97 75.2173.61 76.217.93 77.417.97 74.4717.97 74.4717.97 74.4717.97 74.4717.97 74.4717.97 74.477.97 74.477.97 <t< td=""><td>86.1
86.1
82.3
82.3
78.6
77.6
75.6
75.6
75.6
74.5
74.5
74.5
74.5
74.5
74.5
73.6
72.6
72.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7</td><td>91 4 93 4 98 11 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 84 84.1 84 84.1 74 84.1 74 84.1 74 74.7 76 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7</td><td>Vorticity Vorticity 00.6 Vorticity 88.9 87.1 88.9 87.2 85.2 85.4 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.6 83.7 <td< td=""><td>84,2 86,0 87,4 87,8 87,4 87,8 87,4 87,8 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,1 84,1
 84,1 83,4 83,3 62,7 82,3 81,5 81,5 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 9,7,0,3 7,3,3,3 9,7,3,3 7,3,4 7,3,4 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,4,4 9,4 8,4,7 9,4</td><td>78.0 79.9 79.9 82.0 83.4 83.4 79.9 84.6 83.4 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.7 83.7 83.7 83.7 83.7 81.7 81.2 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7</td></td<></td></t<> <td>73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79</td> <td>LEVELS
707.0 1
45.14
5.7
707.0 1
45.14
5.7
6.7
707.0 1
45.14
5.7
707.0 1
45.14
7.7
7.7
7.7
7.7
7.7
7.7
7.7
7.7
7.7
7.</td> <td>73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,00
74,00
74,00
74,00
74,00
74,00
74,000
74,000
74,000
74,000
74,000
74,0000
74,0000
74,00000</td> <td>2745.6
16.13
27.50
41.02
41.02
51.70
52.67
54.11
54.16
51.79
52.67
54.11
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
555
5</td>
<td>9571.3
-18.35
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.45
-12.44
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-1</td> <td>11142.5
-77.98
-25.51
-77.98
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.52
-25.51
-25.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-</td> | 86.1
86.1
82.3
82.3
78.6
77.6
75.6
75.6
75.6
74.5
74.5
74.5
74.5
74.5
74.5
73.6
72.6
72.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
72.7
73.6
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7
77.7

 | 91 4 93 4 98 11 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 82 84.1 84 84.1 84 84.1 74 84.1 74 84.1 74 74.7 76 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7 74.7 76.7
 | Vorticity Vorticity 00.6 Vorticity 88.9 87.1 88.9 87.2 85.2 85.4 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.6 83.7 <td< td=""><td>84,2 86,0 87,4 87,8 87,4 87,8 87,4 87,8 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,1 84,1 84,1 83,4 83,3 62,7 82,3 81,5 81,5 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 9,7,0,3 7,3,3,3 9,7,3,3 7,3,4 7,3,4 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,4,4 9,4 8,4,7 9,4</td><td>78.0 79.9 79.9 82.0 83.4 83.4 79.9
 84.6 83.4 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.7 83.7 83.7 83.7 83.7 81.7 81.2 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7</td></td<> | 84,2 86,0 87,4 87,8 87,4 87,8 87,4 87,8 85,6 85,6 85,6 85,6 85,6 85,6 85,6 85,1 84,1 84,1 83,4 83,3 62,7 82,3 81,5 81,5 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 81,5 9 9,7,0,3 7,3,3,3 9,7,3,3 7,3,4 7,3,4 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,3,6 7,3,4 7,7,4,4 9,4 8,4,7 9,4
 | 78.0 79.9 79.9 82.0 83.4 83.4 79.9 84.6 83.4 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.6 83.7 83.7 83.7 83.7 83.7 83.7 81.7 81.2 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7 80.9 70.7

 | 73,
75,
77,
79,
79,
79,
79,
79,
79,
79,
79,
79 | LEVELS
707.0 1
45.14
5.7
707.0 1
45.14
5.7
6.7
707.0 1
45.14
5.7
707.0 1
45.14
7.7
7.7
7.7
7.7
7.7
7.7
7.7
7.7
7.7
7. | 73,0
73,0
73,0
73,0
73,0
73,0
73,0
73,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,0
74,00
74,00
74,00
74,00
74,00
74,00
74,000
74,000
74,000
74,000
74,000
74,0000
74,0000
74,00000 | 2745.6
16.13
27.50
41.02
41.02
51.70
52.67
54.11
54.16
51.79
52.67
54.11
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
54.57
555
5 |
9571.3
-18.35
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.44
-12.45
-12.44
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-12.45
-1 | 11142.5
-77.98
-25.51
-77.98
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.51
-25.52
-25.51
-25.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
-27.55
- |

i

Figure 32. Example of Computer Output for One Computer Run.



Figure 33. Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure 34. Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusi's Scaled to 20,000 Lbs.

4.2 Some Experimental Anomalies

Some of the results of this program depart from the expected trends. These anomalies are discussed in the following paragraphs.

The first anomaly occurred when the secondary flow velocity equaled the primary velocity. Evidence of this is to be found in Figures 33 and 34, for Run Numbers 102 to 117. Here, for unity velocity ratio, the scaled power level and perceived noise level values for the coaxial nozzles are in excess of the level produced by the primary nozzle alone. One would expect the scaled noise produced by nozzles with a velocity ratio of unity to be the same as that produced by the primary alone, if any interaction effect between the primary and secondary flow is neglected. In this case, the primary nozzle acting alone has been checked extensively and there is no doubt as to the reliability of the data. In addition, its noise output agrees well with the level predicted by Reference 5. This indicates that the increase in level is due to the presence of the coaxial nozzles. The exact cause of this discrepancy is not known, but an examination of the acoustic power spectra shows that additional noise sources are present which increase the middle and high frequency content of the spectra. The worst case is illustrated in Figure 35.

Examination of all of the power spectra show that this increase in the high frequency region of the normalized power spectra is most significant for the coplanar cold 1.6 pressure ratio nozzle. There is little or no effect for the lower area ratio secondary nozzles with the half and fully extended primaries, whereas with the coplanar nozzles these area ratios exhibited the largest effect. Also, examination of all the coplanar nozzle power spectra for lower velocity ratios indicated that this effect was generally not present.

These results are consistent with the possibility that the additional source is associated with the contraction of the external diameter of the primary nozzle pipe from 1.0 to 0.8 inches over a distance of 1/2-inch forward of the nozzle. This contraction in the primary pipe diameter creates a slightly divergent secondary nozzle for the coplanar case, with the throat approximately 1/2-inch upstream of the nozzle plane. The amount of divergence decreases with increasing area ratio, as can be seen by comparing the area ratios given in Section 3.2 for the coplanar and extended primary nozzle cases. Consequently, the effect would be expected to be greatest for the smaller area ratio secondary nozzles. Also, the acoustic effect would be expected to be greatest for flow velocities in the sonic region where the divergence acts as a supersonic diffuser, accelerating the flow rather than slowing it.

Additionally, as shown in Appendix B, the boundary layer shed from the outside of the primary nozcle prevents the coplanar nozzles from having completely uniform

į



Ì

Figure 35. Power Spectra for Unity Velocity Ratio and Various Area Ratios for Coplanar Jet with Primary Nozzle Pressure Ratio of 1.6 and Temperature of 60°F. (Data are normalized to primary nozzle thrust.)

velocity profiles. The turbulence induced by this velocity defect probably causes additional sound generation at a frequency associated with the scale of the contraction (≈ 0.1 inch) and with a magnitude which increases proportional to a power of the secondary flow velocity. A similar condition would undoubtedly be present on a real full-sized jet engine, although its magnitude may well vary considerably with variations in internal nozzle geometry, and hence would not necessarily scale correctly from these model jet results.

The second anomaly is associated with the secondary nozzle which has an area ratio of 2. This effect is illustrated in Figure 36 for Run Numbers 176 to 187, where the noise for the area ratio 2 nozzle is greater than would be expected. Examination of the nozzle showed that it had been over-heated when the primary was run with high temperature air, since the inside had become roughened and charred. This roughness would cause an unusually turbulent boundary layer to be produced inside the mouth of the nozzle. From an examination of the data, the over-heating appears to have happened just before Run 148; for all run numbers below 148, the nozzle produces results consistent with those produced by the other nozzles, but thereafter it appears to produce too much noise. The results for this area ratio for run numbers greater than 147 should therefore be treated with caution.

The third anomaly concerns a noise source associated with the two supersonic primary nozzles. In the majority of the runs with these nozzles, the model directivity pattern in the higher octave bands (principally the 31.5 kHz and 16 kHz bands) showed an unexpected increase at angles near 90 degrees to the jet axis. The maximum value of this increase was usually found near 110 degrees.

This phenomenon is shown in Figure 37 for Run 128, which had a primary nozzle with pressure ratio 2.5 and temperature of 64°F, together with a coplanar secondary nozzle of area ratio 10 and velocity ratio of 0.38. This figure illustrates one of the worst examples of this phenomenon. It shows the directivity patterns for the 5 highest octave bands. The 31.5 kHz and 16 kHz bands are seen to be 10 to 12 dB high at 110 degrees while the remaining bands form a consistent group, decreasing in level with increasing angle. It is obvious that such a discrepancy cannot be truly representative of a full-size jet and that the model data should be corrected. Fortunately, the peaks in the forward direction of directivity pattern for both the anomalous octave bands have not been affected (this was always the case in all runs with this fault) so that the correction to the directivity and computer corrections could be made.

As part of an evaluation of this phenomena, a one-tenth octave analysis was made of the sound at 110 degrees from the jet axis and at frequencies up to 25 kHz. This analysis is shown in Figure 38. It does not show any evidence of a discrete frequency, but rather a broad peak with a bandwidth of approximately one octave.







 $R_{\rm eff}$ and

Figure 37. Directivity Pattern for Run 128 (Coplanar) Showing the Anomaly in

the High Octave Bands,



٩.

Figure 38. One-Tenth Octave Band Analysis of Directivity Anomaly for Run 128 at 110 Degrees.

:

Finally, a check was made to determine the effect of this anomaly on the computed sound power levels and sideline perceived noise level. The corrections indicated in Figure 37 lowered the sound power level in both the 16 kHz and 31.5 kHz bands. The resulting overall sound power levels decreased 1.5 dB. The sideline perceived noise levels were then checked. It was found that the true maximum sideline perceived noise level was not altered by this change in the higher two bands for any of the full-scale jet thrusts. Similar checks were made on other runs where this problem was severe, with the same basic conclusions. These results indicate that the scaled overall sound power level plots, of which Figure 33 is an example, are altered slightly when the anomaly under discussion is at its worst, but that the maximum side-line perceived noise level is not altered. On this basis, all runs which showed this anomaly were corrected, as in Figure 37, and are noted by an asterisk (*) in the run number schedule in Appendix A.

The cause for this anomaly is not known. It appeared in almost all of the coplanar and extended nozzle configurations, in a few of the half retracted configurations, and none of the fully retracted runs. It was only slightly in evidence (≈ 2 dB at 31.5 kHz) when the 2.5 pressure ratio nozzle was run alone. However, it was present in most of the runs of the 3.5 pressure ratio nozzle which was operated at several pressure ratios. These latter results, shown in Appendix G, indicate that the effect is present for all pressure ratios greater than 2.0, generally increasing in magnitude with increasing pressure ratio. Intentionally Left Blank

の言葉を見ていた。

5.0 DISCUSSION OF RESULTS AND THEIR APPLICATION TO FULL-SCALE ENGINES

This section reviews the principal experimental results which are relevant to predicting the noise of full-scale engines and to understanding the noise reduction characteristics of coaxial jet flows in terms of the analysis in Section 2.3. These results are utilized to develop a prediction method for full-scale engines and to evaluate the differences between the noise characteristics of practical full-scale coaxial jet engines and those of idealized single-nozzle engines with a similar exit velocity.

5.1 Discussion of Principal Experimental Results

The experimental results will be examined with a view to determining the noise reduction characteristics of coaxial jets. The noise reduction of a coaxial jet is defined to be the difference between the noise of the coaxial jet and that of a single-nozzle jet which has the same thrust as does the coaxial jet and the same flow conditions as those of the primary nozzle of the coaxial jet. This comparison eliminates the gross variations due to changes in primary velocity and temperature. The following paragraphs examine the noise produced by the primary nozzles alone and the changes in sound power, directivity and maximum sideline perceived noise level resulting from the coaxial jet flow.

5.1.1 Noise of the Primary Nozzles.

Figures 39, 40 and 41 show the sound produced by the three primary nozzles with the three pressure ratios and three total temperatures as shown, as a function of the flow velocity. Figure 39 shows the measured overall sound power level for the isodel primary nozzles. The dashed curve for the overall sound power level shows reasonable agreement with the data except for the 60-degree, pressure ratio 3.5 nozzle. The line can be represented by:

$$\frac{U}{PWL} = 123.6 + 80 \log \frac{P}{1000} \text{ in dB re } 10^{-13} \text{ worts}$$
(39)

where Up is the primary velocity. This curve is 2 dB greater than that given in Figure 1, which is based principally on data obtained over a ground plane and assumed to have hemispheric radiation. Since the data in this report were taken in free spherical space, the difference might indicate that the typical ground plane in Figure 1 provided some absorption and reflection lasses, partially invalidating the assumption of ideal hemispheric radiation.

Preceding page blank

-71-



à.

this production of the second s

المناسمين الم

182. . 241.

Figure 39. Variation in Measured Overall Sound Power Level with Velocity for the Model Scale Primary Nozzles Alone.





Figure 41. Variation in Maximum 1500–Foot Sideline Perceived Noise Level for the Primary Nozzles Alone, Scaled to 20,000 Lbs Thrust.

Figure 40 shows the same data scaled to a thrust of 20,000 pounds. The dashed line represents the curve of Figure 39 scaled in accordance with the following expression:

$$PWL_{(20,000 \text{ lbs})} = 159 + 60 \log \frac{p}{1000}$$
 in dB re 10^{-13} watts (40)

The change in the velocity exponent from 8 to 6 results from the thrust scaling.

Figure 41 gives the maximum perceived noise level on a 1500-foot sideline for these jets scaled to a thrust of 20,000 pounds. These results appear to be proportional to U^8 . To calculate the overall power for a thrust other than 20,000 pounds, the results on Figure 40 can be scaled by ten times the ratio of the new thrust to 20,000 pounds. However, this simple rule does not apply to the maximum sideline perceived noise level because it is also a function of the noise spectrum. To determine the magnitude of this effect, the change in maximum 1500-foot sideline perceived noise level was calculated for thrusts of 10,000, 20,000, 40,000 and 80,000 pounds. The results are shown in Figure 42, along with the extent of the scatter. The mean line through the data has a slope proportional to F^{83} , indicating that doubling the thrust produces a maximum perceived noise level increase of approximately 2.5 PNdB.

5.1.2 Reduction in Overall Sound Power Level Associated with Coaxial Jet Flows.

To determine the sound power level reductions associated with the various coaxial jet configurations, the overall sound power level of each coaxial jet was compared to the overall sound power level of its primary jet operating alone, with all data scaled to a constant thrust of 20,000 pounds. Figure 43 shows the change in overall sound power level for the coplanar nozzles as a function of velocity ratio, with the simple theoretical estimates of Section 2.3 included for comparison. The data exhibited three anomalies: (1) an increase of noise for coaxial nozzles of unity velocity ratio, (2) relatively poor noise reduction performance for the nozzle with an area ratio of two, and (3) the distinctly lesser noise reductions associated with the primary nozzle with pressure ratio 2.5 when operating at 60° F.

The first two of these anomalies have been discussed in Section 4.2. The third anomaly is present throughout the data. It occurs for cold flows only, not being apparent in the cases when the jet was heated. A uniform reduction of approximately 4 dB would bring the cold 2.5 pressure ratio data into line, suggesting that the sound power of the primary nozzle might be low. However, inspection of Figure 39 shows the noise of this primary nozzle to be consistent with that of the other eight primaries, and other data. Thus, the cause remains a mystery and the data for the cold pressure ratio 2.5 nozzle will be ignored in the remaining discussions.

The data for the coplanar nozzles with area ratios of 5 and 10, and to a lesser extent area ratio 1, show rather good agreement with the simple theory. Further, there seem



Figure 42. Effect of Jet Thrust on Calculated Maximum Perceived Noise Level on 1500-Foot Sideline for the Primary Nozzles of Figure 41.



のないのであるというと

Contraction of the second s

her tils programmingen for del berente poor blegte blev verset versetter gebeger om det versette versette belev

Figure 43. Change in Overall Sound Power Level with Velocity Ratio for Coplanar Coaxial Jets of 20,000 Lbs Thrust and Four Area Ratios. Change is Measured Relative to a Single–Nozzle Jet which has Equal Thrust and a Velocity Equal to U_p.



and a second state of the second s

Figure 43. continued ... Change in Overall Sound Power with Velocity Ratio for Coplanar Coaxial Jets of 20,000 Lbs Thrust and Four Area Ratics. Change is Measured Relative to a Single–Nozzle Jet which has Equal Thrust and a Velocity Equal to U_p.

to be no consistent effects of either temperature or primary pressure ratio on the reductions of overall sound power level. Hence, it would appear that predictions of the sound power of the jet flow from modern high bypass ratio fan engines with coplanar coaxial nozzles can be made considering only the area and velocity ratios of the two flows.

The effect of extending the primary nuzzle 5.5 primary nozzle diameters beyond the plane of the secondary nozzle (one-half extension) and 11 diameters (full extension) is illustrated in Figures 44 through 47. Again, the theoretical estimate of the change in overall sound power is included for reference, although its derivation assumed coplanar nozzles.

The results for the nozzle of area ratio 1, given in Figure 44, show an improvement cver the coplanar performance. This may result from an improvement in its ability to shroud the core since the theory of Section 2.3 indicated that this nozzle was not large enough to shroud the primary core. However, with the extended primary nozzle, the secondary flow should be significantly larger than its initial diameter when it passes through the plane of the primary nozzle. At this plane, it will be slower than indicated by its exit velocity ratio, thus extending the shrouding regime to higher velocity ratios than predicted for coplanar nozzles in Section 2.3. Unfortunately, there are no velocity profiles available to test this conjecture.

The data for the nozzle with an area ratio of 2 are contained in Figure 45. They show improvement over the coplanar case, but the scatter is large and the roughness of this nozzle makes any evaluation questionable. The data for the nozzle with area ratio 5, Figure 46, show little change from the coplanar case, except for a small increase in scatter. However, the data for the fully extended nozzle of crea ratio 10, Figure 47, show a decrease in noise reduction relative to the coplanar case. The reason for this change in performance is not known; however, the data appear consistent and the results must be considered valid.

The acoustic performance of retracted primary nozzle configurations can be expected to differ from that of the coplanar and extended primary nozzle configurations since the flows and acoustic boundary conditions are different. During all tests, the nozzles were operated at a fixed set of total pressures such that the velocity ratios for the non-retracted cases were consistent among runs which had the same primary nozzle and temperature. However, when the primary nozzle is retracted within the secondary plenum, its effective pressure ratio is decreased and its velocity is similarly decreased. Its initial mixing with the secondary flow occurs within the plenum, where the secondary flow velocity is quite low; howaver, the noise generated by this mixing is somewhat shielded by the walls of the secondary nozzle.



San Barlin and the standing of the

Figure 44. Change in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 1 and Thrust of 20,000 Lbs. Change is Measured Relative to a Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to U_p.



Figure 45. Change in Overall Sound Power Level with Velocity Ratio for Coaxiai Jets with a Nozzle Area Ratio of 2 and Thrust of 20,000 Lbs. Change is Measured Relative to a Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to U_p.

a she was a state of the second state of the



and the second design of the second second

the stand of the second of the second states with the second second second second second second second second s

Figure 46. Change in Overall Sound Power Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 5 and Thrust of 20,000 Lbs. Change is Measured Relative to a Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to U_D.



Figure 47. Change in Overall Sound Nower Level with Velocity Ratio for Coaxial Jets with a Nozzle Area Ratio of 10 and Thrust of 20,000 Lbs. Change is Measured Relative to a Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to Up.

Assuming that the primary is retracted sufficiently that it completely mixes with the secondary flow price to exiting from the secondary nozzle and its noise does not radiate through the nozzle, the average or effective velocity through the nozzle would be expected to be given by:

$$U_{e} \cong \frac{4 (m_{s} + m_{p})}{g}$$
(41)

The acoustic power should be proportional to:

$$W_{\rm m} \sim d_{\rm s}^2 U_{\rm e}^8$$
 (42)

This situation represents a lower bound on the expected results which was not approached with the configurations evaluated in this study. The obvious upper bound is given by the assumption that all the noise of the primary mixing into the slow moving secondary is radiated through the nozzle and adds to the noise generated by the jet issuing from the secondary nozzle as it mixes with the ambient. Also, at low pressure ratios, additional noise might be expected from the turbulence generated in the plenum as it is convected downstream, outside of the secondary nozzle. The true situation for these data obviously lies between these two bounds.

The basic results for the variation of axial position, including the retracted primary nozzle cases, are given for the pressure ratio 3.5 primary nozzle in Figure 48. The 3.5 pressure ratio nozzle was chosen for discussion because its behavior is reasonably typical of the average data in the coplanar and extended primary nozzle cases, and because its flow parameters underwent the least change in the retracted cases. The changes in noise level for all cases in Figure 48 were based, as before, on the noise of the primary nozzle operating at a fixed total pressure and exhausting into the ambient.

The results illustrate the small increase in noise relative to the captanar configuration for the area ratio 10 nozzle at both extended positions, as previously discussed. No change is exhibited for the retracted nozzles. For the area ratio 1 and 5 nozzles, however, a small reduction relative to the coplanar case is found in the retracted cases, as well as in the extended cases. The more limited data for the pressure ratio 1.6 nozzle indicate slight increases relative to the coplanar cases for the retracted cases. However, it is not presented in detail since its flow was drastically affected by the secondary plenum pressure, and is not representative of full-scale engine design conditions.


Figure 48. Effect of Axial Position of the Primary Nozzle as Measured by the Change in Overall Sound Power Level Relative to the Level for Each Nozzle and Pressure Combination in the Coplanar Position. In all Cases, the Primary Total Pressure is 36.5 psig (Nominal 3.5 Pressure Ratio) and Total Temperature is Ambient. Lines Connect Averages for Each Area Ratio.

5.1.3 Changes in Sound Power Spectra and Directivity.

The principal changes in overall sound power associated with coaxial jet flows are also illustrated in examining a few typical sound power spectra. Figure 49 shows the sound power spectrum for a model jet configuration which has a primary nozzle with a pressure ratio of 1.6 and total temperature of 450°F, and a coplanar secondary nozzle with an area ratio of 10. The spectra are given for four velocity ratios ranging from zero (primary nozzle alone) to 0.88. The spectrum for the velocity ratio of 0.38 is 6 to 7 dB lower than the spectrum for the primary alone in the frequency range above the peak in the spectrum, while at low frequencies its sound power level is slightly above the level of the primary alone. The overall sound power level (at model thrust) is approximately 3 dB lower. This reduction of levels at the higher frequencies demonstrates that the low velocity secondary flow is shrouding the primary flow, thus reducing its acoustical output. The secondary flow would be expected to reduce the high frequencies in preference to the low frequencies because the high frequency acoustic sources are located in the initial mixing region of the primary jet where the effect of the secondary flow is greatest, whereas the lower frequencies are produced downstream where the secondary flow has less effect. Conversely, the increase in low frequency noise generation is to be expected as a result of the larger diameter equivalent jet resulting from the combined flows.

When the secondary velocity is increased to give a velocity ratio of 0.73, the equivalent jet becomes the dominating factor. The spectrum for this jet has a shape that should be expected of a single-nozzle jet with an exit area approximately ten times that of the primary nozzle alone. A final increase in secondary velocity to give a velocity ratio of 0.88 does not significantly change the spectrum shape, but does increase the spectrum levels in proportion to the eighth power of the change in secondary velocity. All of these results are consistent with the simple theoretical model postulated in Section 2.3.

Figure 50 shows the effect of extending the primary nozzle beyond the secondary. Comparing Figures 49 and 50, the effect of shrouding the primary can be seen to have nearly disappeared for the velocity ratio of 0.38 when the primary nozzle is extended eleven (11) primary diameters beyond the secondary nozzle. The spectra for velocity ratios of 0.725 and 0.878 appear to have a few dB less power when the primary is extended than in the coplanar case.

Comparison at a velocity ratio of 0.88 shows little change in the high frequency portion of the spectrum which should be controlled by the initial mixing region of the secondary jet. However, the low frequency portion of the spectrum of the extended primary is lower in level than that of the coplanar case. This result is consistent with the hypothesis that the equivalent jet for the extended nozzle cases should be defined by the flow conditions existing in the plane of the extended primary.









In Figure 51, the effect of area ratio on the shrouding of the primary nozzle is shown for coplanar nozzles at a velocity ratio of 0.38. These results demonstrate that flows from coaxial nozzles with small area ratios have little apparent shrouding effect, whereas flows from the larger secondary nozzles have considerable effect. This result is in accordance with the simple theory, since a secondary nozzle which has an area ratio of only 1 or 2 results in a fairly thin layer of secondary flow around the primary jet. Within a very few nozzle diameters downstream from the nozzle, the primary flow will have developed into a diameter larger than the original secondary nozzle diameter. The primary nozzle flow will therefore tend to behave over most of its length as if there were no secondary flow. It should be emphasized that these spectra are from the model jet, and hence no allowance has been made for varying thrusts. If they were corrected to an equal thrust, a small shrouding effect would be seen for the lower area ratios and an even larger effect would be observed for the higher area ratios.

Accompanying the changes in spectrum are changes in the acoustic radiation directivity pattern. Figures 52 through 55 show the directivity patterns for the flow cases whose spectra are shown in Figure 49. These results show that as the velocity ratio increases, proportionally more sound is radiated at larger angles from the jet flow direction. The 500 Hz and 1000 Hz octave bands do not change directivity very much, but the remainder are affected. The angular position of the peaks in the octave band directivity indices move by a maximum of 10 degrees; but at angles away from the peak, the directivity indices vary by as much as 12 dB in extreme cases near the jet axis and by up to 4 dB at 110 degrees from the axis. As a result, the angle at which the directivity index is zero increases by up to 30 degrees in both cases.

5.1.4 Change in Maximum Sideline Perceived Noise Level.

The effect of the various configurations on the maximum perceived noise level on the 1500-foot sideline is summarized for the coplanar nozzle configurations in Figure 56. This figure gives the data for perceived noise level in the same format used for the overall sound power level in Figure 43, and includes the theoretical prediction curve for power level reduction for reference. There is much similarity between the two sets of figures, although the maximum reduction in perceived noise level for area ratios 5 and 10 is nearly 1 dB greater than the maximum reduction in power level. All the general comments made previously concerning the change in overall sound power level are applicable to perceived noise level.



1.11.11.11.11





Figure 52. Directivity Pattern for a Primary Nozzle Alone.



いたがいいたいすいでももう

č

Ċ

1

รางรางการสารสารมาก เมษาสมบัตร์สารสารไป มีการสมบัตร์แหน่งสารสารสารใช้เพิ่มจะสิมบัตร์สารสารสารสารสารสารสารสารสาร

Figure 53. Directivity Pattern of Coplanar Nozzle with Velocity Ratio of 0.375.



and a set of the second se

an bane di katalan sanan ngangan katalan katalan katalan katalan katalan katalan katalan katalan katalan katala

Figure 54. Directivity Pattern for Coplanar Nozzle with Velocity Ratio of 0.731.



ł

Figure 55. Directivity Pattern for Coplanar Nozzle with Velocity Ratio of 0.88.



Ú.

Figure 56. Change in 1500–Foot Sideline Maximum Perceived Noise Level with Velocity Ratio for Coplanar Coaxial Jets of 20,000 Lbs Thrust and Four Area Ratios. Change is Measured Relative to a Single–Nozzle Jet which has Equal Thrust and a Velocity Equal to U_p.



Figure 56. continued ... Change in 1500-Foot Sideline Maximum Perceived Noise Level with Velocity Ratio for Coplanar Coaxial Jets of 20,000 Lbs Thrust and Four Area Ratios. Change is Measured Relative to a Single-Nozzle Jet which has Equal Thrust and a Vencity Equal to Up.

Figure 57 shows the data for one-half and full extension of the primary nozzle for area ratio 5. The data appear slightly more scattered than the associated overall sound power level data shown in Figure 46. However, they appear reasonably well represented by the coplanar data curve.

Figure 58 gives a presentation of the effect of axial position on the 1500-foot sideline perceived noise level, analogous to the sound power data of Figure 48. The results are similar, but with slightly more scatter.

rational second second with a state of the second second second second second second second second second secon

In general, the changes in maximum perceived noise level are very similar to those of overall sound power level, except that there is a tendency for slightly more scatter in the data. This tendency may result partially from the fact that the maximum perceived noise level is dependent upon a single angular measurement, whereas the sound power is a function of measurements at many angular positions. Also, the perceived noise level is more sensitive to the level in a few actave bands, whereas the overall sound power is a function of most of the actave bands. For both these reasons, the perceived noise level measurement may be inherently more susceptible to minor variations in the experimental process.

In addition, the perceived noise level is affected by any changes in directivity which are associated with the various configurations. The effect of the directivity changes on the variation in perceived noise level along the 1500-foot sideline, shown in Figures 52 through 55, is given in Figure 59. Zero feet on the sideline corresponds to the point directly apposite the jet (i.e., at 90 degrees to the jet axis). The data show that the maximum sideline level shifts along the sideline towards zero feet as the velocity ratio increases. In addition, there is a slight change in the rate at which the perceived noise level folls with increasing distance along the sideline.

5.2 Prediction of the Noise Characteristics of Full-Scale Coaxial Jets

The data and theory presented in this report can be used as the basis for the prediction of the noise characteristics of full-scale coaxial jets. This section develops prediction methods for bath the overall sound power level and the 1500-foot sideline maximum perceived noise level, the latter being the principal goal of this study. The principal emphasis is on coplanar coaxial nozzles which are of considerable practicul interest. For prediction of non-coplanar coaxial jets, the reader is referred to the discussions in Section 5.1 which consider the effects of axial displacement of the primary nozzle relative to the secondary nozzle. Caution should be exercised in applying this latter data to full-scale geometries which differ significantly from the models studied here.



Figure 57. Change in 1500-Foot Sideline Maximum Perceived Noise Level with Velocity Ratio for Cocxial Jets with a Nozzle Area Ratio of 5 and Thrust of 20,000 Lbs. Change is Measured Relative to a Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to U_p.



Figure 58. Effect of Axial Position of the Primary Nozzle as Measured by the Change in Maximum Perceived Noise Level in the 1500-Foot Sideline Relative to the Level for Each Nozzle and Pressure Combination in the Coplanar Position. In all Cases, the Primary Total Pressure is 36.5 psig (Nominal 3.5 Pressure Ratio) and Total Temperature is Ambient. Lines Connect Averages for Each Area Ratio.





The basic predictions for sound power level are developed directly from Figures 40 and 43. The results are given in Figure 60. When $\Delta_{DW} = 0$, the equation at the top of the figure represents the relationship found for zero secondary flow (primary only) from Figure 40 and the basic jet noise scaling laws. The curves which give the decrease of overall sound power level as a function of coaxial jet velocity ratio were obtained from Figure 43. They are based on the approximate data curves, where given in Figure 43, and are supplemented by the theoretical curves from Figure 21 as required for completeness.

Figure 61 presents a similar prediction method for the 1500-foot sideline maximum perceived noise level. When $\Delta_{DP} = 0$, the equation represents the results for the primary flows given in Figures 41 and 42. The curves giving the decrease of perceived noise level as a function of velocity ratio were obtained from Figure 56 in a manner similar to the development of the sound power level prediction curves.

There are two basic potential sources of error in using these two figures for prediction of the noise of a specific full-scale engine which has coplanar nozzles. First is the potential error in estimating the noise for the reference primary jet which may be expected to vary approximately ± 2 dB about the mean data line. Second, the model data for the coaxial flows, ignoring anomalous data, show a scatter of $\pm 1-2$ dB about the mean curves, describing the decrease of noise as a function of velocity and area ratio. These two errors might be expected to add in a mean square sense to give a total error of just under ± 3 dB for a specific prediction.

However, for tradeoff studies leading to engine optimization for noise, these prediction methods should be quite accurate, particularly since the simple theory appears to be in basically good agreement with the mean of the experimental results.

Figures 60 and 61 again demonstrate that the minimum noise for a coplanar coaxial jet with a fixed primary velocity and thrust occurs at a velocity ratio of approximately 0.5. The noise characteristics for a family of engines meeting this criterion and having 20,000 lbs. thrust and various area ratios are given in Figures 62 (power) and 63 (perceived noise).

From one viewpoint, however, the results presented in Figures 62 and 63 are more an indication of the potential remedial effects of a secondary flow. From a purely design viewpoint, the approach to optimization proceeds in a different manner. In engine design, the design thrust is one of the principal specified objectives. For a single nozzle with a given pressure ratio, the thrust produced is independent of the flow temperature. However, the noise produced by the jet is not independent of temperature since higher temperatures imply higher nozzle velocities, and hence more noise.



Figure 60. Prediction of Overall Sound Power Level Based on Primary Nozzle Data in Figure 40 and Mean Curves of Experimental Variation in the Decrease in Overall Sound Power with Velocity Ratio for Jets with Coplanar Coaxial Nozzles and Fixed Thrust. Decrease of Coaxial Jet Sound Power Level is Measured Relative to a Baseline Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to the Primary Velocity.



Figure 61. Prediction of 1500-Foot Sideline Perceived Noise Level Based on Primary Nozzle Data in Figures 41 and 42 and Mean Curves of Experimental Variation in the Decrease in 1500-Foot Sideline Maximum Perceived Noise Level for Jets with Coplanar Coaxial Nozzles and Fixed Thrust. Decrease of Sideline Perceived Noise Level is Measured Relative to a Baseline Single-Nozzle Jet which has Equal Thrust and a Velocity Equal to the Primary Velocity.



Figure 62. Variation of Minimum Sound Power as a Function of Area Ratio and Primary Velocity.



Figure 63. Variation of Minimum Value of the Maximum Perceived Noise Level on a 1500–Foot Sideline as a Function of Area Ratio and Primary Velocity.

From an acoustic viewpoint, it is therefore important to keep the jet temperature as low as possible, which means that a jet total temperature equal to ambient temperature is the possible minimum. The jet exhaust velocity, and hence the noise, can be reduced still further by reducing the nozzle pressure ratio as much as possible. There is obviously some practical minimum to the pressure ratio so that the resultant jet engine will produce the necessary thrust without becoming unnecessarily large with attendant weight and drag penalties. For takeoff conditions and engines with fixed convergent nozzles, that practical pressure ratio is most likely to be somewhere in the region of the critical pressure ratio, or slightly below, so that the engine is still efficient at cruise conditions. From an acoustic viewpoint, therefore, a practical optimum single jet engine should produce a jet exhaust at sonic pressure ratio with an exhaust temperature as near ambient as possible.

These conclusions were reached from an examination of a single jet nozzle. But for present-day jet engines, the requirements for optimum conditions actually apply to the bypass flow of practical bypass jet engines. For modern engines, the secondary flow temperature is only slightly greater than ambient temperature, and optimum overall performance generally dictates that the pressure ratio at takeoff thrust be slightly less than critical. For such designs, it is very instructive to interpret the noise reduction results using the secondary flow velocity as a fixed parameter and determine the amount of the detrimental effect on the engine's noise characteristics resulting from a higher temperature, higher velocity primary flow.

Figure 64 shows the predicted increase in overall sound power leve! (Δ_{1W}) for various values of area ratio as a function of the ratio of primary jet velocity to secondary jet velocity. The equation for the predicted sound power level and the design curves in the figure is a restatement of the information in Figure 60.

Figure 65 gives a similar presentation of the increase in the maximum perceived noise level on a 1500-foot sideline, developed from Figure 61.



Figure 64. Prediction of Sound Power Level for a Coaxial Jet Using Secondary Velocity for Reference.



Figure 65. Prediction of Maximum Perceived Noise Level on a 1500-Foot Sideline for a Coaxial Jet Using Secondary Velocity for Reference.

Tables 3 and 4 give the velocity ratios (primary velocity to secondary velocity) which cause either a 3 dB or 6 dB increase in overall sound power or maximum perceived noise level on a 1500-foot sideline relative to the noise at a velocity ratio of 1. It will be noticed that there is a considerable difference between power level and perceived noise level in sensitivity to the ratio of primary-to-secondary flow velocity. For example, for a coplanar nozzle with area ratio of 10 and primary velocity ratio of approximately 1.5, there is only a 3 dB increase in overall sound power output, but a 6 dB increase in 1500-foot sideline perceived noise level. These differences are the penalties resulting from having a primary velocity 50 percent greater than that of the secondary. This greater sensitivity of perceived noise level to changes in velocity ratio is due to the fact that the overall power level for fixed thrust increases in proportion to velocity to the sixth power, whereas the perceived noise level increases in proportion to velocity to the eighth power.

Table 3

Primary Velocity Ratios (U_p/U_s) for a 3 dB and 6 dB Increase in Overall Power Level from Figure 64

Power Level	Area Ratio				
Increase	1	2	5	10	
3 dB	1.14	1.17	1.40	1.52	
6 dB	1.34	1.42	1,70	1.85	

Toble 4

Primary Velocity Ratios (U_p/U_s) for a 3 dB and 6 dB Increase in Sideline Perceived Noise Level from Figure 65

Perceived Noise Level Increase	Area Ratio				
	1	2	5	10	
3 PNdB	1.11	1,12	1.18	1.23	
6 PNdB	1.29	1.22	1.37	1.53	

It should be noted that a primary-secondary velocity ratio of 1.5 is typical of current high bypass ratio engine design, primarily for non-acoustic reasons. This velocity ratio is in the region where the noise produced by the secondary mixing with the ambient and the equivalent jet control the overall sound power characteristics, except at low area ratios where the primary is inadequately shrouded. These concepts were developed in Section 2.3, particularly with reference to Figures 18 and 20.

.

5.3 Comparison of Noise Characteristics of Practical Coaxial Jet Engines with Optimum Single-Nozzle Engines

If the heat could be extracted from the primary jet, reducing its temperature to near ambient, its velocity would be approximately equal to that of the secondary. In this case, no noise penalty would be associated with the higher speed primary jet and the resulting engine would have near optimum performance and noise characteristics. The following paragraphs compare the acoustic characteristics of such an engine with those of currently feasible coaxial jet engines.

The model jet that comes nearest to this idealized jet is the primary nozzle with a pressure ratio of 1.6 and a flow temperature of 60° F. Its acoustic characteristics can be compared with those of the coplanar jets, which had area ratios of 5 and 10 and primary nozzles with pressure ratio 1.6 but flow temperature of 450° F. These two coplanar jets have primary velocity to secondary velocity ratios of approximately 1.35 and secondary flow velocities of 890 feet per second. The secondary flow theremfore has nearly the same flow conditions as the 60° F primary nozzle acting alone, so that these two coplanar runs correspond approximately to the conditions for the area ratio 5 nozzle indicated in Tables 3 and 4, which give a 6 dB increase in sideline perceived noise level and a 3 dB increase in overall sound power.

Figure 66 gives the acoustic power spectra for a 20,000 lb. thrust jet for these three configurations. It shows that the coplanar nozzles with the higher temperature and higher velocity primary jet produce considerably more high frequency noise than does the single-nozzle cold jet. This increased acoustic power at higher frequencies gives an indication as to why the coplanar jet shows a larger increase in perceived noise level than is indicated by the increase in overall sound power level. This conclusion is reinforced by a comparison of the actave band sound pressure spectra shown in Figure 67 for the position on the 1500-foot sideline which has the maximum perceived noise level. This position is at an angle of 55 degrees from the jet axis for the primary jet alone, but is at 65 degrees for the coplanar jets, which indicates the effect of changes in the directivity pattern.







all of the design of the second state of the s



Figures 68, 69 and 70 present the octave band directivity patterns for these three jets. The directivity maxima increase in angle by about 10 degrees when secondary flow is present. There is also a change in the shape of directivity pattern, so that the largest angle at which the directivity index is zero increases by as much as 25 degrees. These angular changes are reflected in Figure 71 which shows the 1500foot sideline perceived noise level for the three jets. The changes in directivity cause the perceived noise level to drop more rapidly from the maximum value for the coaxial jets than is the case for the primary nozzle alone. Thus, at the peak level the coaxial jets produce about 6 PNdB more noise, whereas at large distances along the sideline in the direction of the jet exhaust, the difference is only 2 PNdB.

The above results have been presented for a 20,000 lb. thrust engine. It has already been shown (Figure 42) that on the average, the maximum perceived noise level increases by 2.5 PNdB for every doubling of jet thrust in the range of thrust from 10,000 lbs. to 80,000 lbs. This scaling factor generally holds over the complete sideline curve, as demonstrated in Figure 72 which gives the calculated values of the sideline perceived noise levels for four different thrusts for the jet with pressure ratio of 1.6 and total temperature of 60°F.



Figure 68. Directivity Pattern for Primary Nozzle Alone.



Figure 69. Directivity Pattern for Coplanar Nozzles with Area Ratio 5, Velocity Ratio 0.722.



Figure 70. Directivity Pattern for Coplanar Nozzles with Area Ratio 10, Velocity Ratio 0.731.







E.



6.0 SUMMARY AND CONCLUSIONS

The acoustic output of more than three hundred (300) different model coaxial jet configurations has been analyzed and used to determine the effect of the slower secondary (or bypass) air flow on the sound output of coaxial jets. The model data were scaled to give the sound power output of a full-scale jet engine, and the sound pressure levels and perceived noise levels determined along a sideline parcilel to the jet axis. Noise reductions for coaxial jets were computed relative to a baseline single-nozzle jet which had the same thrust as that of the coaxial jet and a velocity equal to the coaxial jet primary. The principal conclusions from this work are as follows:

- The jet noise for coaxial jets with fixed thrust and area ratio was found to be minimum when the ratio of secondary velocity to prime y velocity was approximately 0.5.
- (2) The reduction in jet noise for fixed velocity ratio and thrust increases with increasing area ratio for coplanar or nearly coplanar nozzles. The maximum reduction in overall sound power was found to be 10 dB, and the maximum reduction in sideline perceived noise level was 11 PNoB for the nozzle which had a secondary to primary area ratio of 10. However, the reductions for an area ratio of 5 were approximately only 1 dB less than those for area ratio 10. These experimental results largely substantiate the simple theory of the noise of coplanar coaxial jets which was developed in this study.
- (3) These noise reductions were found to be largely independent of the pressure ratio of the primary nozzle and the total remperature of the primary flow over the range of pressure ratios (1.6 - 3.5) and temperatures (60°F - 800°F) covered in the experimental program.
- (4) While for a fixed velocity and temperature condition the overall sound power output is directly proportional to angine thrust, the maximum perceived noise level on a 1500-foot sideline was found to increase by only 2.5 PNdB each time the thrust was drubled, over a thrust range from 10,000 lbs. to 80,000 lbs. The maximum perceived noise level is therefore proportional to thrust raised to the power 0.833, when velocity and temperature conditions are held constant.
- (5) At fixed thrust and area ratio, the maximum perceived noise level on a 1500foot sideline was found to increase as the eighth power of velocity, whereas the overall sound power level increases as the sixth power.

- (6) An engine designed to have maximum thrust per unit area will generally have minimum weight and drag per unit thrust. Assuming the secondary flow is limited by aerodynamic constraints to sonic or nearly sonic flow speeds, the acoustically optimum engine will have primary and secondary velocities of approximately 1000 feet per second. This implies that the acoustically optimum primary flow temperature should be very near ambient and the pressure ratio no greater than that of the secondary. Although the pressure ratio condition is essentially achieved in current high bypass ratio jets, the temperature condition is difficult to achieve, although it represents increased cycle efficiency.
- (7) If all conditions of conclusion (6) could be achieved with a 20,000 lb, thrust engine, it would have an estimated overall sound power output of 159 dB relative to 10⁻¹³ watts and a maximum perceived noise level on a 1500-foot sideline of 88 PNdB.
- (8) Small increases in the primary velocity relative to the secondary velocity produce small increases in acoustic output, their magnitudes being a function of area ratio. Tables 5 and 6 give the ratios of primary velocity to secondary velocity which results in either 3 dB or 6 dB increase in noise relative to the noise of an engine which has a velocity ratio of unity and the same secondary velocity and total thrust. The tables therefore give the velocity ratio which results in the specified noise increase compared to the optimum engine discussed in conclusions (6) and (7). The marked differences between the two tables results from the difference between the sensitivities of the overall sound power and perceived noise level to changes in primary velocity.

Table 5

Secondary–Primary Velocity Ratio for Specified Increase in Overall Sound Power

Power Level Increase	Area Ratio				
	1	2	5	10	
3 අප	1,14	1.17	1,40	1.52	
ර	1.34	1,42	1,70	1.85	
Tab	e	6			
-----	---	---			
-----	---	---			

Secondary–Primary Velocity Ratio for Specified Increase in Perceived Noise Level

Perceived Noise		Area	Ratio	
Level Increase	1	2	5	10
3 PNdB	1,11	1.12	1.18	1.23
6 PNdB	1.29	1.22	1.37	1.53

- (9) It is probable that the noise reduction associated with coplanar coaxial jets can be increased by designing the nozzles to promote more rapid mixing. Such nozzles would probably enable the design of engines with jet noise characteristics approaching the optimums discussed in conclusions (6) and (7).
- (10) The extension of the primary nozzle up to eleven (11) primary nozzle diameters beyond the secondary nozzle produced an increase in the noise reduction for nozzles of area ratio 1, little change in the noise reduction for nozzles of area ratio 2 and 5, and a decrease in the noise reduction for nozzles with area ratio 10.
- (11) Significant retractions of the primary nozzle inside the secondary nozzle tended to decrease the noise relative to that produced by coplanar nozzles for all of the secondary nozzles when the primary nozzle had a high pressure ratio (3.5), but to increase the noise when the convergent primary nozzle with a subsonic pressure ratio (1.6) was used. However, the flow conditions for this latter primary nozzle were not characteristic of practical engines, and it is concluded that retracted primaries can be beneficial when properly designed.

Intentionally Left Blank

· "U

REFERENCES

- Lighthill, M.J., "On Sound Generated Aerodynamically: I, General Theory," Proceedings of the Royal Society, A, Vol. 211, p 564, 1952.
- Lighthill, M.J., "On Sound Generated Aerodynamically: II, Turbulence as a Source of Sound," Proceedings of the Royal Society, A, Vol. 222, p 1, 1954.
- 3. Plumblee, H.E., Wynne, G.A. and Zinn, B.T., "Effect of Jet Temperature on Jet and Fure Tone Noise Radiation," NASA CR-1472, 1969.
- 4. Plumblee, H.E., "Effect of Duct Heating on Jet and Fan Noise," Proceedings of Basic Aerodynamics Noise Research Conference, NASA Headquarters, Washington, D.C., NASA SP-207, July 1969.
- Eldred, K.M., White, R.W., Mann, M.A. and Cottis, M.G., "Suppression of Noise with Emphasis on the Near Field," Report ASD-TDR-62-578, Wright-Patterson Air Force Base, Ohio, 1963.
- 6. Eldred, K.M., "Review of the Noise Generation of Rockets and Jets," JASA, Vol. 32, No. 11, p 1502.
- Rollins, V.G., "Effect of Jet Temperature on Jet-Noise Generation," NACA TN-4217, March 1958.

- Eldred, K.M., "Acoustic Loads Generated by the Propulsion System," NASA SP-8072, June 1971.
- Grande, E., "Refraction of Sound by Jet Flow and Jet Temperature, 11" University of Toronto, Institute of Aerospace Studies, UTIAS Report 110, 1966.
- Atvars, J., Schubert, L.K., Grande, E. and Ribner, H.S., "Refraction of Sound by Jet Flow or Jet Temperature," NASA CR-494, May 1966.
- Lighthill, M.J., "The Bakerian Lecture, 1961, Sound Generated Aerodynamically," Proceedings of the Royal Society, A, Vol. 267, p 147, 1962.
- Howes, W.L., Callaghan, E.E., et al, "Near Noise Field of a Jet Engine Exhaust," NACA Report 1338, 1957.
- 13. Lourence, J.C., "Intensity, Scole and Spectra of Turbulence in Mixing Region of Free Subsonic Jet," NACA Report 1292, 1956.

Preceding page blank

- 14. Johannesen, N.H., "The Mixing of Free Axially Symmetrical Jets of Mach Number 1.40," ASTIA AD-203564, January 1957.
- 15. Pitkin, E.T., Glassman, "Experimental Mixing Profiles of a Mach 2.6 Free Jet," ASTIA AD-202294, August 1958.
- Anderson, Arthur, Johns, Frank, "Non-Dimensional Characteristics of Free and Deflected Supersonic Jets Exhausting into Quiescent Air," WADC-ED-5401, March 1954.
- 17. Dyer, I., "Distribution of Sound Sources in a Jet Stream," Journal of Acoustical Society of America, Vol. 31, p 1016, 1959.
- Ribner, H.S., "On the Strength Distribution of Noise Sources Along a Jet," University of Toronto, Institute of Aerospace Studies, UTIAS Report No. 51, 1958.
- Westley, R. and Lilley, G.M., "An Investigation of the Noise Field from a Small Jet and Methods for its Reduction," Report No. 53, College of Aeronautics, Cranfield, England, 1952.
- 20. Greatrex, F.B., "Aeronautica! Acoustics in Particular Jet Noise," Journal of Royal Aeronautical Society, London, April 1954.
- 21. Greatrex, F.B., "Engine Noise," Joint Symposium on Aeronautical Acoustics, London, May 1953.
- 22. Mercer, D.M.A., et al, "The Effect of 'Teeth' on the Noise from a Jet Engine," WADC TR 54–224, September 1954.

- 23. Collaghan, E.E., et al, "Tooth Type Noise Suppression Devices on a Full-Scale Axial Flow Turbojet Engine," NACA RM E54-801, March 1954.
- 24. North, W.J., "Summary Evaluation of Toothed Nozzle Attachments as a Jet Noise Suppression Device," NACA TN 3516, July 1955.
- 25. Laurence, J.C. and Benninghoff, J.M., "Turbulence Measurements in Multiple Interfering Air Jets," NACA TN 4029, December 1957.
- 26. Ciepluch, et al, "Acoustic, Thrust, and Drag Characteristics of Several Full-Scale Noise Suppressors for Turbojet Engines," NACA TN 4261, April 1958.

- 27. Greatrex, F.B., "Bypass Engine Noise," Society of Automotive Engineers, Reprint 162C, 1960.
- 28. Pao, S.P. and Lowson, M.V., "Spectral Techniques in Jet Noise Theory," Wyle Laboratories Research Staff Report WR 68–21, April 1969.
- 29. Lowson, M.V., "Some Applications of Jet Noise Theory," Wyle Laboratories Research Staff Report WR 69–21, October 1969.
- Aviation Week and Space Technology, "Forecast and Inventory Issue," March 10, 1969.
- Ames Research Staff, "Equations, Tables and Charts for Compressible Flow," National Advisory Committee for Aeronautics, Report 1135, Ames Aeronautical Laboratory, Moffett Field, California, 1953.
- 32. Society of Automotive Engineers, "Standard Values of Atmospheric Absorption as a Function of Temperature and Humidity for use in Evaluating Aircraft Flyover Noise," Aerospace Recommended Practice (ARP) 866, August 1964.

Sharen asha heela anda marka harareka alah madhera ada ana bar ana ana ada asha hikara nambaha

, se den bieren herrig bereite sitzen geschlichter ben inder biere finden bener filtert innen sitzer tiltetet s

- Evans, L.B. and Sutherland, L.C., "Investigation of Anomalous Behavior of Sound Absorption by Molecular Relaxation," Wyle Laboratories Research Staff Technical Memorandum, TM 70–4, May 1970.
- 34. Society of Automotive Engineers, "Definitions and Procedures for Computing the Perceived Noise Level for Aircraft Noise," Draft Revision ARP 865. Original date - October 15, 1964.
- Hermes, P.H. and Smith, D.L., "Measurement and Analysis of the J57-P21 Noise Field," AFFDL-TR-66-147, Wright-Patterson Air Force Base, Ohio, 1966.

Intentionally Left Blank

Sec. Same

1

ş

Mahan

Contention of

South Mary Sugar

APPENDIX A

DIRECTORY OF EXPERIMENTAL RUNS

This appendix contains a listing of the experimental runs that were made in the course of this project.

Table A-1 shows the groups of experimental runs associated with the various flow conditions of the primary nozzles and the relative position of the primary and secondary nozzles. Half extended or retracted means the primary nozzle is displaced 5.5 primary nozzle diameters axially from the exit plane of the secondary nozzle. Fully retracted or extended refers to 11 diameters. Within each group of runs indicated in the table, there is an equal number of runs for each of the four secondary nozzles. For each secondary nozzle, the secondary flow velocity is varied between that required to give a velocity ratio (secondary to primary) of one-third, up to a sonic velocity. Intermediate velocity ratios of two-thirds and unity are interposed between these extremes if this does not require the secondary velocity to be supersonic.

Table A-2 gives a detailed listing of the actual completed runs, giving the following data:

- Run Number (numbers in parentheses are reruns; asterisks indicate corrections were made to correct the high frequency data in the vicinity of 110 degrees for the third anomaly discussed in Section 4.2)
- Nozzle Position
- Nominal Area Ratio
- Primary Temperature
- Primary Pressure Ratio
- Primary Velocity
- Velocity Ratio (secondary/primary)
- Mass Flow Ratio (secondary/primary)

For additional detail, see Volume II.

Preceding page blank

127 A-1 Table A-1. Directory of Experimental Run Conditions

			æ	imary No	zzle Flow	Tempera	ture		
		60°F			450°F			800°F	
	Primary	y Pressure	Ratio	Primary	Pressure	Ratio	Primary	Pressure	Ratio
	1.6	2.5	3.5	1.6	2.5	3.5	1.6	2.5	3.5
Primary Nozzle Alone	ĩOI	118	131	144	157	166	175	188	197
_			909009						
Coplanar Nozzles	102-117	119-130	132-143	145-156	158-165	167-174	176-187	189-196	198-205
Fully Extended Primary	215-230	231-242	243-254	255-266	267-274	275-282	283-294 (1)	1	303-310 (1)
Fully Retracted Primary	311-326 (2)	327-338 (2)	339-350 (2)	1	ŧ	t	ł	ł	I
Half Extended Primary	407-422	423-434	435-446	447-458	459-466	467-474	ł	ŧ	ł
Half Retracted Primary	503-516	519-530	531-592	ŧ	I	I	1	I	I

(1) Some runs not made in these groups (see Table A-2 for details).

(2) Runs not made when secondary supply pressure greater than primary supply pressure. Primary plenum pressures for these runs were identical with those of all other runs; however, actual pressure ratios across primary exhausting into secondary were lower and are given in Table A-2,

7 ् ठ A-2

Table A-2

Experimental Run Conditions

itun Number	Nozzle Position	Nom. Area Ratio	Primary Temp_~F Average	Primary Pressure Ratio	Primary Velociły Ft/Sec	Velocity Ratio	Mass Flow Ratio	Run Number	Nozzle Position	Nom. Area Ratio	Primary Temp. ~ ^O F Average	Primary Pressure Ratio	Primary Velocity Ft/Sec	Velocity Ratio	Mass Flow Rotio
101	Primary Nozzla	-	64	1.5	899	-	-	149	Coplanar	2	450	1.6	1176	0.72	3.05
	Alone							150 (811)	Coplanar	2	450	1.6	1176	0.87	4.09
102	Ceptonor		64	1.6	900	9.35	0.58	151	Coplanor	5	450	1.6	1176	0.38	3.29
103	Coplenar	I	64	1.6	894	0.70	0.93	152	Coplanar	5	450	1.6	1176	0.72	6.65
104	Coplana	1	64	1.6	874	1.0	1.33	153	Coplanar	5	450	1.6	1176	0.87	9.04
105	Omitted	-	-	-	-	-	-	154 (727)	Coplanar	10	450	1.6	1176	0.38	6.02
165	Cop!anar	2	64	1.6	891	0.35	1.10	155	Coplanar	10	450	1.6	1176	0.73	13.38
107 (705)	Coplonar	2	64	1.6	893	0.70	1.99	156 (729)	Caplanar	10	450	1.6	1176	0.88	18.38
108 (722)	Coplanar	2	64	1.6	892	1.0	2.61	+157	frimary Nozzle	-	450	2.5	1590	i -	-
109	Copianar	2	64	1.6	893	1.15	3.25		Alone				1000		
110 (706)	Coplanar	5	64	1.6	872	0.25	1.63	•158	Coplanar	1	450	2.5	1590	0.41	0.08
111 (723)	Coplanar	5	64	1.6	892	0.70	3.79	•159	Coplanar	1	450	2.5	1590	0.00	1,18
112 (803)	Coplana	5	64	1.6	894	1.0	5.50	•160	Coplanar	2	450	2.5	1590	0.40	1,48
113	Omitted	-		-	-	-	-	161	Omitted	-	-	-	-	-	-
114 (708)	Coplanar	10	64	1.6	895	0.35	3.30	*162	Coplanar	5	450	2.5	1590	0.41	3.07
115 (804)	Coplanar	10	64	1.6	899	0.70	7.15	163	Coplanar	5	450	2.5	1590	0,65	5.10
116	Coplanar	10	64	1.6	878	1.0	11.03	*164 (812)	Coplanar	10	450	2.5	1590	0.40	5.42
117 (726)	Coplanar	10	64	1.6	881	1.16	14.28	165	Copianar	10	450	2.5	1590	0.66	10,22
118	Primary Nozzle Alone	-	64	2.5	1204	-	-	*166	Frimary Nozzle Alone	-	450	3.5	1817	-	-
119 (707)	Coplanar	1	64	2.5	1199	0.38	0.46	*167 (732)	Copinnar	1	450	3.5	1817	0.57	1.02
120 (710)	Coplanar	1	64	2.5	1199	0.73	0.79	168	Omitted	-	-	-	-	-	-
•121 (713)	Coplanar	1	64	2.5	1207	0.84	1.04	• 169	Coplanar	2	450	3.5	1817	0.42	1.38
• 122 (714,	Copionor	2	64	2.5	1195	0.38	0.52	•170 (734)	Coplanar	2	450	3.5	1817	0.57	1.96
•123	Copianar	2	64	2.5	1193	0.73	1.61	•171	Coplanar	5	450	3.5	1817	0.42	3.00
*124 (715)	Coplanar	2	64	2.5	1193	0.86	2.03	*172 (736)	Coplanar	5	450	3.5	1817	0.58	4.15
+125 (717)	Coplanar	5	64	2.5	1182	0.38	1.67	•173	Coplanar	10	450	3.5	1817	0.42	5.42
•126	Coplanar	5	64	2.5	1177	0,73	3.34	•174	Coplanar	10	450	3.5	1817	0.58	8.45
127 (719)	Coplanar	5	64	2.5	1177	0.86	4.14	175	itimary Nozzle	-	800	1.6	1401	-	-
•128	Coplanar	10	64	2.5	1184	0.38	3.18	174	Contener		800	1.6	1401	0.37	1.01
+129	Copionar	10	64	2.5	1194	0,73	6.69	177	Coolmor		800	1.6	1362	0.71	1.90
•130	Coplanur	10.	64	2.5	1195	0.87	8.38	179	Coolanor		800	1.6	1362	0.78	2.03
•131	Primary Nozzie	-	64	3.5	1380	-	-	170	Coplanat	2	800	1.6	1364	0.37	2.07
	Alone		1					180 (737)	Coplanar	,	800	1.6	1359	0.69	3.62
•132	Copianar	1	64	3.5	13/4	0.35	0.38	100 (7377	Contonat	2	800	1.6	1361	0.77	4.53
+133	Omitted	-	-	-			-	192	Conlong	•	800	1.6	1369	0.36	3.63
•134	Coplanar		64	3.5	1382	0.74	0.78	182 (738)	Contenes		800	1.6	1362	0.69	8.48
*135 (720)	Coplanur		64	3.5	13/0	0.35	0.08	163 (730)	Coolonar	5	800	1.6	1366	0.77	9.84
*136 (809)	Coptonor	2	54	3.5	1308	0.07	1.31	195 (915)	Conlorer	10	800	1.6	1366	0.36	7.67
*137	Coplanar	2	64	3.5	13/4	0.74	1,40	195 (015)	Contenet	10	800	1.6	1359	0.69	17.68
*138	Copiuner	5	64	3.5	1376	0.35	1,40	100	Coolanar	10	800	1.6	1359	0.77	21.85
*139	Coplonot	5	64	3.5	1376	0.07	2.70	*188	Primary Norvia		800	2.5	1864	-	
*140	Coplanar	5	64	3.5	1367	0.75	3,30	· ~	Alone	-					
*141 (721)	Coplanor	10	64	3.5	1387	0.35	2.04 # 08	•189	Coplanar	1	800	2.5	1853	0.39	0.78
*142	Coplanar	10	64	3.5	1383	0.07	2,00	+190	Coplanar	1	800	2.5	1856	0.58	1.38
•143	Coplanar	10	64	3.5	138/	0'5	0.0/	191	Coplanar	2	800	2.5	1855	0.38	1.84
144	Primary Nozzle Alone	-	450	1.0	11/0	-		192	Coplanar	2	800 800	2.5	1856	0,57	2.65
145	Copionar		450	1.6	1176	0.38	0.78	101 (*11)	Castanas		ann	2.5	1853	0.56	5.94
146	Coplanar	11	450	1.6	1176	0.73	1.43	104 (/41)	Copionar	[°]				-	
147	Coplaner	1	450	1.6	1176	0.88	1.89	175	Comitted	1.0	-	<u>.</u>	1942	0.64	1200
148	Coplanar	2	450	1.6	1176	0.38	1,38	1 196	Coplanar	110		1 4.3	1004	0.00	1 12.00

A-3

129

Table A-2 - continued

Bun Number	Nazzie Paition	Non, Anc Ratio	Primary Temp. ~ °F Average	Primary Pressure Natio	Primary Velocity Ft/Sec	Velocity Rotio	Mase Flow Ratio	Run Number	Nozzle Position	Nom. Ansa Rotio	Primary Temp, ~ F Average	Aimary Fressure Ratio	Primary Velocity Ft/Sec	Velocity Ratio	Mass Flow Ratio
•197	Primary Nazzle Alane	-	800	3.5	2169	-	-	•241	Fully Extended Primary	10	66	2.5	1208	0.72	5.31
*1#8	Copianar	1	800	3.5	2183	0.38	0.84	242	Omitted	-	-	-	-	-	-
*1 99	Copionar	1	800	3.5	2178	0.49	1.08	-243	Fully Extended	1	66	3.5	1378	0.36	0.27
*200	Coplanar	2	800	3.5	2179	0,38	1.74	1	Primory						
201	Copianar	2	800	3.5	2183	0.48	2.27	*244	Fully Extended	1	66	3.5	1333	0.67	0.59
202 (744)	Coplanar	5	900	3.5	2183	0.38	3.84	+245	Sully Feteraled	,	~	3.5	1384	0.74	0.70
203	Omitted	-	-	-	-	-	- 1		Primary	•	~	0.0	1444	•	
*204	Coplanar	10	800	3.5	2180	0.38	7.29	*246	Fully Extended	2	66	3.5	1367	0.36	0.51
205	Copianar	10	800	3.5	2184	0.48	10.06		Primary				1074		1.00
206 -214	Omitted	-	-	-		-	-	*247	Fully Extended Primary	2	~~	3.5	1374	0.00	1.05
215	Fully Extended Primary	1	66	1.6	898	0.35	0.32	*248	Fully Extended Primary	2	66	3.5	1380	0.74	1.29
216	Fully Extended Primary	1	66	1.6	896	0.70	0.82	*247	Fully Extended Primory	5	66	3.5	1374	0.35	1,30
217	Fully Extended Primary	1	66	1.6	900	0.99	1.21	*250 (747)	Fully Extended Primory	5	66	3.5	1370	86.0	2.81
218	Fully Extended Primory	1	66	1.6	896	1.14	1.49	*251	Fully Extended Primory	5	66	3.5	1378	0.76	3.23
219	Fully Extended Primary	2	66	1.6	892	C.35	0.60	*252	Fully Extended Primary	10	66	3.5	1388	0.36	2.47
220	Fully Extended Primary	2	66	1.6	892	0.70	1.34	*253	Fully Extended Primary	10	66	3.5	1368	0.68	5.46
221	Fully Extended Primary	2	66	1.6	895	1.0	2,79	*254	Fully Extended Primary	10	66	3.5	1387	0.75	6.31
222	Fully Extended Primary	2	66	1.6	895	1,14	3.36	255	Fully Extended Primary	1	450	1.6	1177	0.39	0.50
223	Fully Extended Frimary	5	66	1.6	894	0.34	1,60	256	Fully Extended Primary	1	450	1.6	1184	0.73	1.77
224	Omitted	-	-	-	-	-	-	257	Fully Extended	1	450	1.6	1183	0.88	1.56
225	Fully Extended Primary	5	66	1.6	895	0.99	6.09	258	Primary Fully Extended	2	450	1.6	1187	0.39	1.04
226	Fully Extended	5	66	1.6	900	1.15	7.01		Primory	-					
227	Fully Extanded	10	66	1.6	88.7	0.35	3.20	259	Primary	2	450	1.0	110/	0.73	2.04
228	Fully Extended Primary	10	66	1.6	895	0.70	6.65	260	Primary Evilue Extended	2	430	1.0	1185	0.38	2 45
229	Fully Extended Primary	10	66	1.6	898	1.0	10.58	201	Primary	5	430	1.5		0.50	
230	Fully Extended Primery	10	66	1.6	898	1.16	12.44	262	Fully Extended	5	450	1.6	1183	0.88	7.96
*231	Fully Extended Primory	1	66	2.5	1210	0.38	0.32	264	Fully Extended	10	450	1.6	1186	0,38	5.44
+232	Fully Extended Primory	1	66	2.5	1221	0.71	0.68	265 (750)	Fully Extended	10	450	1.6	1185	0.72	11.66
*233	Fully Extended Primory	1	66	2.5	1220	0,84	0.86	266	Fully Extended	10	450	1.6	1186	0.88	16.15
*234	Fully Extended Primary	2	66	2.5	1209	0,38	0.58	*267	Fully Extended Primary	1	450	2.5	1605	0.41	0.54
7235	Fully Extended Primary	2	66	2.5	1217	0.72	1.26	*268	Fully Extended Primary	1	450	2.5	1594	0.66	1,15
*236	Fully Extended Primary	2	66	2.5	1204	0.95	1.54	269	Fully Extended Frimary	2	450	2.3	1604	0,40	0.95
*237	Fully Extended	5	66	2.5	1200	0.38	1.60	1270	Omitted	-	- 1	-	-	-	_]
1238	Fully Extended	5	66	2.5	1187	0.73	3.24	*271	Fully Extended Primary	5	450	2.5	1601	0.40	2.62
+230	Canittani	_	-	_	_	-	_	*272	Fully Extended	5	450	2.5	1605	U.65	4.88
*240	Fully Extended	10	66	2.5	1208	0.38	2,44	*273	Frimary Fully Extended	10	450	2.5	1603	0,40	5.07
1	. ·	1	1	1	1	1	1	1	i cumery		1	1	í	Į	1

130 A-4

Table A-2 - continued

Run Number	Nozzle Position	Nom. Area Ratio	Primory Temp. ~ F Average	Primary Pressure Ratio	Primary Valacity Ft/Sec	Velocity Ratio	Mass Flow Ratio	Run Number	Nozzle Position	Nom. Area Ratio	Primary Temp, ~ ⁰ F Average	Primary Pressure Ratio	Primary Velocity Ft/Sec	Velocity Rotio	Mass Flow Ratio
•274	Fully Extended Primary	10	450	2.5	1604	0.65	9.62	330	Fully Retracted	2	74	2.23	1146	0.40	0.76
+275 (751)	Fully Extended	1	450	3.5	1802	0.42	0.64	331	Fully Retracted	2	74	1.61	906	0.96	1.66
•276	Fully Extended	1	450	3.5	1800	0.58	0.98	332	Fully Retracted	2	74	1.32	702	1.47	2.49
•277	Fully Extended	2	450	3.5	1800	0.42	0.72	333	Fully Retracted	5	74	2.23	1141	0.40	1.79
+278 (752)	Fully Extended Frimery	2	450	3.5	1800	0.58	1.07	334	Fully Retracted	5	74	1.56	872	1.03	3.68
•279	Fully Extended Primary	5	450	3.5	1798	0.42	1.52	335	Fully Retracted Primary	5	74	1.32	705	1.47	4.98
•280	Fully Extended Primary	5	450	3.5	1800	0.58	2.19	336	Fully Retracted Primary	10	74	2.23	1148	0.40	3.20
•281	Fully Extended Primory	10	450	3.5	1803	0.42	5.17	337	Fully Retracted Primary	10	74	1.61	904	0.96	6.34
•282	Fully Extended Primary	10	450	3.5	1802	0.58	7.%	338	Sully Retracted Primary	10	74	1.32	705	1.47	8.85
283-284	Omitted	-	-	-	-	-	-	339	Fully Retracted	1	74	3.06	1330	0.37	0.46
285	Fully Extended Primary	1	800	1.6	1372	0.78	1.74	340	Fully Retracted	1	74	2.09	1114	0.84	0.89
266	Fully Extended Primary	2	800	1.6	1374	0,37	1.33	341 (763)	Fully Retrocted	1	74	1.85	1025	1.01	1.12
287 (754)	Fully Extended Primary	2	800	1.6	1376	0.69	2.92	342	Fully Retracted	2	74	3.06	1331	0.37	0.74
268	Fully Extended Primary	2	800	1.6	1379	0.76	3.58	343	Fully Retracted	2	74	2.09	1114	0.84	1.43
289	Fully Extended Primity	5	800	1.6	1377	0.36	3.57	344	Fully Retracted	2	74	1.85	1024	1.01	1.68
290	Fully Extended Primory	5	800	1.6	1374	0.69	7.63	345	Fully Retrocted	5	74	3.06	1331	0.37	1.62
291	Fully Extended Primary	5	800	1.6	1374	0.76	8,82	346	Fully Retracted	5	74	2.09	1109	0.85	3.09
292-302	Omitted	-	-	-	-	-	-	347 (764)	Fully Retracted	5	74	1.85	1024	1.01	3.57
- 303 (733)	Primary Ecilie Entended	,	••••	3.5	2186	0.37	0,04	348	Primary Fully Retracted	10	74	3.06	1335	0.37	2.88
	Primary	,	~~~			0.47	0,70		Primary	10		2.00	1114		4.04
305~310	Omitted	-	-	-	-	-	-	349	Pully Retracted	10	14	2.05	1115	0.85	0.05
311	Fully Retracted Primory	1	74	1.52	846	0.37	0.71	350	Fully Retracted Primary	10	74	1.85	1023	1.02	6.69
312	Fully Retracted	1	74	1,29	664	0.94	1.80	351-407	Omitted	-	-	-	-	-	-
313-314	Omitted	-	-	-	-	-	-	408 (823)	Half Extended	1	60	1.6	896	0.70	0.63
315	Fully Retracted Primary	2	74	1.52	845	0.37	0.99	409	Half Extended Primary	1	60	1.6	894	1.0	1.04
316	Fully Retracted Primary	2	74	1.29	663	0.94	2,15	410	Half Extended Primary	1	60	1.6	895	1.15	1.28
317-319	Omitted	-	-	-	-	-	-	411	Half Extended	2	60	1.6	894	0.35	0.55
320	Fully Retracted Primary	4	74	1,29	664	0.94	4,34	412	Primory Half Extended	2	60	1,6	896	0.70	1.27
321-322	Omitted	-	-	1 49	-	-	-	412	Primary Nati European	,	40	14	807	1.0	2 19
323	Frimary Cultur Battantia	10	74	1.54	440	0.3/	3.03	413	Primory Half Extended	, ,	40	1.0	804	1.14	4.14 2.40
328 327	Primory	10		1.47				A15	Primary	5	40	1.4	844	0.35	1.54
328 (832)	Fully Retracted	- 1	74	1.61	896	0.96	1.16	414	Primary Malf Eutor de 1		40	14	887	0.70	3.94
329 (76.7)	Fully Retension	,	74	1.32	700	1.47	1.68	410	himary	5		1.0	UGU UGU	0.70	0.20
	Primary	•						417 (824)	Half Extended Primary	5	60	1.6	887	1.01	5.14

131

A-5

ENSUITATINE ROOM

Table A-2 - continued

Run Number	Nezzie Position	Nom. Area Ratio	Primary Temp. ~ ⁰ F Aversige	Primary Pressure Ratio	Primary Velocity Ft/Sec	Velocity Ratio	Mass Flaw Ratio	Run Number	Nazzle Position	Nom. Area Ratio	Primary Temp. ~ ^o F Average	Primary Pressure Retio	Primary Velocity Ft/Sec	Velocity Ratio	Mass Flow Ratio
418	Half Extended Primary	5	60	1.6	867	1.16	5.98	450	Half Extended Primary	2	460	1.6	1187	0.36	0.99
419	Half Extended Primary	10	60	1.6	867	0.35	2.94	451	Half Extended Frimary	2	460	1.6	1185	0.72	2,42
420	Half Extended	10	٥٥	۱.6	887	0.70	6.12	452	Half Extended	2	460	1,6	1188	0.87	3.14
421	Half Extended Primary	10	60	1.6	894	1,01	9.75	453	Half Extended Primary	5	460	1.6	1187	0.38	2.64
422	Half Extended Primary	10	60	1.6	892	1.16	11.90	454 (759)	Half Extended Primary	5	460	1.6	1186	0.73	5.81
•423	Half Extended Primary	1	60	2.5	1198	0.38	0.26	455	Half Extended Primary	5	460	1.6	1185	0.86	7.63
•424	Half Extended Primary	ł	60	2.5	1198	0.72	3.6 0	456	Half Extended Frimary	10	460	1.6	1143	0.38	5.38
•425	Half Extended Primary	1	60	2.5	1202	0.85	0.75	457	Half Extended Primary	10	460	1.6	1186	0.73	11.52
•425	Half Extended Primary	2	60	2.5	1202	0.38	0.56	458	Half Extended Primary	10	460	1.6	1186	0.88	15.09
*427	Half Extended Primary	2	60	2.5	1202	0.72	1.21	*459	Half Extended Primary	1	460	2.5	1601	0.40	0,52
*428	Omitted Mail Eutomated	-	-	-	-	-	-	•460	Half Extended Primary	1	460	2.5	1597	0.43	0.97
*427	Primary			2.5	1202	0.30	1.45	*461	Holf Extended Primary	2	440	2.5	1401	0.40	1.00
-4.50	Primory	2	2	2.3	1205	0.74	3.02	*462 (876)	Half Extended Primary	2	460	2.5	1596	0.45	1.09
*431	Half Extended Primary	5	80	2.5	1211	0.65	3.65	•463	Half Extended Primary	5	460	2.5	1400	0.40	2.68
*432	Half Extended Primary	10	6 0	2.5	1210	0.38	2,79	•464	Half Extended	5	440	2.5	1599	0.45	4.96
•433	Holf Extended Primary	10	60	2.5	1205	0.72	5.94	•465	Half Extended Disease	10	450	2.5	1601	0.40	3.10
434	Helf Extended Primery	10	60	2,5	1208	0.65	7.48	•445	Helf Extended	10	440	2.5	401	v.45	9.47
415	Holf Extended Primory	1	60	3.5	1354	0.34	0.24	•467	Half Extended	1	440	3.5	1829	0.42	0.57
N36	Half Extended Primory	T	60	3.5	1360	0.48	0.54	-441	Helf Extended	1	440	3.5	1824	0.57	0.74
N37	Helf Extended Primery	1	60	3.5	1363	0.75	0.62	•44?	Half Extended	2	440	2.5	1627	0.42	0.94
NOR	Helf Extended Primery	2	60	1.5	1366	0.38	0.48	+470	Helf Extended	2	440	3.5	1828	0.58	1.52
437	Half Extended Primery	2	60	3.5	1366	0.48	1.12	*471	Holf Estanded	\$	440	J. 5	1628	0.47	2.44
1440	Hulf Extended Primary	2	60	35	1367	0.75	1,24	*472	Half Extended Risearce	5	440	J.S	1825	0.58	2.86
1441	Half Extended Primary	5	60	3.5	1370	0.36	1,22	1473	Holf Entended	10	4 r0	J.S	1878	0.42	4.63
442	Helf Extended Primary	5	60	3.5	1374	0.48	2,73	*474	Half Entended	10		1.5	1878	0.58	7,75
443	Helf Extended Primary	5	60	J.5	1340	0.75	3.14	475-503	Omittad	•	-	-	-		-
444	Holf Extended Primory	10	40	3.5	1374	0.36	2.44	504	Hall Reinsclud Minary	1	*0	1.29	644	0. N	1,39
445	Half Extended Primery	10	60	J.5	1376	0.48	5.27	505 (765)	Half Retrocted Primary	2	70	1.52	847	0.37	1,04
446	Half Extanded Primary	10	40	J.5	1380	0.73	6.08	506 (766)	Helf Betractad Primary	3	70	3.29	643	0.14	2,24
447	Half Extended Primary	1	440	₹.6	1189	0.34	0.57	507-510	Omitted Half Retracted	- s	- x	 1.52	 M3	- 0.37	7.03
448	Helf Extended Primery	1	440	1.6	1189	0.73	1,14	512	Ninery Helf Batmeted	5	70	1,29	441	0.N	4.47
419	Holf Extended Primary		440	1,6	1185	0.40	1.63	\$13-514	Ninnry Onitted	_	-	_		-	-

A-6 2

e e la sub e e balance a la sub e la sub e la sub e la sub

Table A-2 - concluded

Rin Number	Nozzie Positien	Nom. Arca Ratio	Primory Temp. ~ ⁰ F Average	Primacy Pressure Batio	Primary Velocity Ft/Sec	Velocity Ratio	Mans Flow Ratio	Bun Number	Nozzie Positian	Non. Area Ratio	Primery Temp, ~ ^o F Averege	Primary Pressure Botie	Nimery Velocity Ft/Sec	Velocity Sutie	JA Fl Re
\$15	Half Estrected Primery	10	70	1.52	843	0.37	3.52	1604 (836)	Primary Nezzle Alene	-	74	4.0	1450	-	Γ
514	Half Retracted	10	70	1.29	667	0.N	7.68	405	Primary Neszle Alone	-	74	5.4	1570	-	
517-518	Omitted	-	-	-	-	-	-	+605	Primary Nezzle	-	74	7,8	1688	-	
519	Half Retrected Ninery	1	70	2.23	1149	0.40	0.49	L	Alene	L	L	L	L	L	L
*320	Helf Retrected Primary	1	70	1.41	103	0.97	1.33								
-521	Helf Betracted Primery	1	70	1.32	702	1.47	1.92								
\$22	Half Retrected	2	70	2.23	1144	0.40	1.01								
52 3	Helf Retrected	3	70	1.61	294	0.17	1.92								
-524	Half Katested	3	70	1.32	695	1,47	2.78								
-123	Helf Kansund	5	×	2.23	1130	0.40	1.95								
536	Half Satracted	3	×	1,81	393 3	0.97	3,49								
1527	Half Retracted	3	70	1.27	699	1.48	3.79								
524	Half Beinicind Namer	10	70	2.23	1144	0.40	2,19								
3279	Helf Botosciad Primary	1.9	70	1,41	249	0.97	4.03								
330	Half Retracted	10	x	1.22	101	2.45	9.14								
•531	Half Ratescind Bimmery	1	ניז	3.04	1314	Q.27	6.67								
-523	Helf Entracted Namery	1	210	2.09	1095	0.63	1.10								
-533 (Ne7)	Half Researched	1	*3	1.89 [L917	1,02	1.27								
utin Milita	Helt Retrested Manager	3	x 0	3.04	1320	9.27	ò.m								
נגר	Helf Besincered Relinery	>	2 70	3.09	1100	0.43	1.58	ļ							
-324	Holf Reinschall Rissory	T	, 70	1.65	OLDI	1.02	1.00								
-527	Half Hannated	3	70	3.06	1231	9.37	1.23								
538	Ontring	{ -	-	; - ;	- 1	-									
**	Helf Ritweind Binney	3	70	1.63	1673	ા.લ	3,47								
<u>مب</u>	Half Remotion	10	~	3.06	1222	9.27	2.67								
541	Helf Bewerned Rivery	13	≂	3.04	1258	9.13	3.84								
542	Half Revected	10	70	1.03	141210	1.62	4.77								
547-599	OniHed	-	-	-	••	-									
***	Nimply Nagalo Alune	-	N	3.0	1674	-	-								
60 1	Rivery Neesle Alene	-	74	8.5	1221	-									
433	Minney Hinesto Aluma] -	N 1	3,0	1248	-	-								
403	Aimery Marsta Alama	-	84	3.5	1307	-	-								

133

A-7

Intentionally Left Blank

.

APPENDIX B

JET FLOW PROFILE MEASUREMENTS

Measurements were made of the total pressure, static pressure and total temperature profile at the exit plane of the model jets prior to initiating the acoustic tests. These tests were conducted primarily to check the velocity profile at the exit of the primary and secondary jets. In addition, the data provided an independent check on the thrust measurements. A few profile measurements were also made well downstream of the exit plane to investigate the spreading characteristics of a coaxial jet. Typical results from these measurements are presented in this appendix. The instrumentation and apparatus used has been described in Sections 3.2.4 and 3.3.4.

B-1 Test Conditions

Test conditions for which significant flow profile data were obtained are summarized in Table B-1. The first series of runs were made without any secondary nozzle. The remaining runs were made with both nozzles in place, although in some cases there was no air flow through one of the nozzles. It is apparent that only a small number of the possible configurations were used for obtaining flow profile data.

B-2 Test Results

Figure B-1 presents a sample of the raw data taken directly from the chart recorder, showing a total pressure profile measurement for a primary nozzle with a pressure ratio of 1.6. Data of this type is the basis for the calculation of flow velocity. Figure B-2 shows the velocity profile at the exit plane of the nozzle with pressure ratio 1.6 as a function of total temperature of the flow. As would be expected from aerodynamic considerations, the boundary layer tends to become thinner with increasing temperature. Note that the abscissa is broken to expand the scale within the region of interest. Figure B-3 shows a similar set of profiles, except that in this case the flow total temperature is held at approximately 850°F while the nozzle pressure ratio is changed. It should be noted that these three profiles were made by three different nozzles, not by one nozzle with a varying pressure ratio.

8-1

Test Conditions for Flow Profile Data Table B-1

1

Sandan utarta harrigi ang katarta dan taka ta dang dikina matarta

	ŀ	_+				×						ľ	× (2)	×			×(2)	×	×	×	×
Profiles ⁽⁴⁾	٥	•											×	×	×		×			×	×
	C	r r	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×	×
ary ⁽¹⁾	Ratio	Actual	ue .						1.83	1.87	1.84	1.89	1.5	1.27	ff	1.89	ff	1.84	1.92	1.89	H.
Second	Pressure	Nominal	Z						1.89					1.27	0	1.89		1.89			•
	T	oF	58	460	008	§	ş	820				-	790	790	790	820	820	ff	790	880	880
٨	e Ratio	Actual	1.58	1.58	1.59	2.53	1.5-2.6	2.7-3.7					2.1	2.43	2.43	3.2-4.1	3.5		2.49	2.45	2.45
Primar	Pressun	Nominal	1.6			2.5	+	3.5	2.5							3.5		2,5			
	Nozzle	Position	Alone						Coplanar									Retracted(3)	Retracted(3)	Extended(3)	Extended(3)
Nominal	Area		0							8	S	0						ŝ	0	¥	-

Total Temperature of Secondary varied from 70°F to 90°F.

Profile Measurements also taken at Positions Downstream of Nozzle Exit Plane. <u> ୧୭୭</u>୫

Fully Retracted or Fully Extended Position. $P_{f} = Total Pressure, P = Static Pressure, <math>T_{f} = Total Temperature.$

145 B-2



Figure B-1. Typical Graphic Record of Total Pressure Profile for a Primary Nozzle with Pressure Ratio 1.6.



Figure B-2. Effect of Temperature on Boundary Layer at the Exit of Primary Nozzle.



Figure B-3. Typical Flow Velocity Profiles Through Exit Plane Boundary Layer for the Three Primary Nozzles.



Figure 8-4. Typical Protile of Total Pressure at Exit Plane of Supersonic Nozzle (Pressure Ratio 3.5) Operating at Otf-Design Conditions.

8-

The complete velocity profile for the supersonic nozzles was found to be very sensitive to changes in pressure ratio. Figure B-4 illustrates the changes that occur for the nozzle designed for a pressure ratio of 3.5. The correct plenum pressure for this nozzle is 37 psig, which unfortunately was not included in the series of runs. However, the figure still indicates that the discontinuities in the velocity profile decrease as the correct pressure ratio is approached.

Figure B-5 shows the total pressure profiles for the flow from the secondary nozzles with a primary nozzle in position but without any airflow through it. The profiles were made on a single traverse across a diameter of the jet. For presentation, the profiles for both sides of the jet are placed on the same graph. A slight lack of symmetry which decreases with area ratio is displayed on these figures. This was due partly to off-center fitting of the primary nozzle which was rectified by improving the location and fixing of the primary nozzles. Note that on the figure for area ratio 10, a dashed line is included to show the profile that was measured when the primary nozzle, without any airflow through it, was retracted inside the secondary nozzle. In this case, the flow at the exit plane is essentially uniform across the whole diameter of the nozzle.

The remaining figures show measurements made with airflow in both the primary and secondary nazzles. Figure 8-6 shows the total pressure and total temperature profiles at several axial positions for a nozzle with area ratio 10. The primary nozzle was designed to run at a pressure ratio of 2.5 but for this series of traverses was actually run at 2.1. The secondary nozzle was operated at a pressure ratio of 1.5. There is evidence of the operation of the primary nozzle below its design pressure ratio shown by the discontinuous form of the total pressure profile at the exit plane within the region of the primary flow, It will also be noted that there is a sharp dip in the total pressure profile of the edge of the primary nozzle caused by the finite thickness of the edge of the primary nozzle and the boundary layer shed from the inside and outside of the primary nazzle. As the profile station is moved downstream, the evidence of the shock structure in the primary flow grows weaker and the mixing process between primary and secondary flow becomes apparent. Figure 8-7 shows a similar series of profiles for the primary nozzle with pressure ratio 3.5. Again, the nozzle is operating slightly off design pressure ratio. It is also apparent that the rate of decay of total pressure in a rodial direction is much more rapid than the decay in temperature. This is a characteristic of the mixing process in jet flow.

> 138 B-5





1 ∪ 9 8-6



gure B-0. Total Pressure and Temperature Profiles at Several Axia Positions for Coplanar Nozzles with Area Ratio 10. Primary Pressure Ratio 2.1, Secondary Pressure Ratio 1.5. (Primary Operating Below Design Pressure Ratio.)





The profiles measured with the primary nozzle fully extended or fully retracted did not show particularly striking features. The only case illustrated (Figure B-8) shows the total pressure and temperature profile across the exit plane of the secondary nozzle (area ratio 10) with the primary nozzle fully retracted. The primary nozzle was run at a temperature of 790°F. There is a very slight bulge in the pressure and temperature profiles at the exit plane, which indicates that considerable mixing of the secondary and primary flows has taken place. Profile measurements made with the primary fully extended were distinguished by the absence of any significant difference between these profiles and the profiles obtained for the primary alone. This is not too surprising in view of the relatively long distance (8.5 inches) between the secondary and primary nozzles.



Figure B-8. Total Pressure and Temperature Profiles at Exit Plane of Secondary Nozzle (Pressure Ratio 1.89, Area Ratio 10) with Primary Nozzle (Pressure Ratio 2.5, Total Temperature 790°F) Fully Retracted.

B-9142



Intentionally Left Blank

APPENDIX C

COMPUTER PROGRAM FOR DATA ANALYSIS

Section 3.3 of this report has given details of the analytic methods used to process the experimentally-obtained data and to scale them to obtain results applicable to full-scale jets. The details of the programs used for accomplishing these analyses are given in this appendix.

The data obtained from the experiment was in two forms. The basic data giving flow conditions, pressures and temperatures for each run was noted on run sheets and then transcribed and punched on FORTRAN data cards. The remaining octave band and overall sound pressure data obtained from the traverse of the jet sound field was stored on magnetic tape and later processed by an automatic octave band analyzer which punched the data on eight-hole paper tape. A preliminary program was therefore required to reproduce this data on FORTRAN data cards. This reproduction program was written at the Control Data Corporation (CDC) Data Center at Los Angeles to run on a small CDC 160A computer. This was a machine language program and is not reproduced here.

The main portion of the data processing was done using a FORTRAN program written for the CDC 6600 computer at Los Angeles. Details of this program are given in the following pages. The program was written as a series of subprograms. The connections between these subprograms are shown in Figure C-1. A brief description of each of these subprograms is then given, followed by a complete listing.

C-1 Controlling Program A1901

Purpose

This is the controlling program for the reduction and analysis of the model jet acoustical data. It accepts the model data and scales it to equivalent full size bypass jet engine exhaust noise data.

Argument List

The elements in the argument list for this routine are control information required by the particular CDC 6600 Computer System on which this program was run.

143

C-1



Figure C-1. Basic Flow Diagram for Complete Data Reduction Computer Program

Common Storage

/DATA1/

FF contains one-third octave band center frequencies (these are set in a DATA statement).

MI, LI, LC, M2, L2 are constants (set in a DATA statement) utilized in subroutine OLLER for perceived noise level calculations.

/DATA2/

ICH is a counter used in subroutine AQUIRE (set as a DATA statement).

PI is the constant π .

ABO, AB1, AB2 are constants used in function FCTION (set as DATA).

/DATA3/

The items in this common data store are as defined below as input to this routine (except that at input, the first character is repeated; e.g., RRM at input is equivalent to RM in DATA3).

/DATA4/

RC is the value of specific acoustic impedance (ρc , cgs rayls).

R is the radius of the boom (ft) of the traverse microphone.

VM is the velocity of the microphone on the boom (ft/sec).

DT is the time to take one sound level reading with the traversing microphone (sec).

TH contains a list of the angles relative to the jet axis at which sound measurements were made.

XR1, XR2 are the two sideline distances.

THR1, THR2, THR3 and THR4 are the four full-scale jet thrusts.

F contains the octave band center frequencies.

All the above data were set as a DATA statement.

/DATA5/

N, N1, NC are integers controlling the amount of the experimental data that is used in the analysis (set as DATA). V is storage space for the experimental data.

DBMS is the instrumentation noise floor.

IND is used in subroutine STEP3.

Unnamed Common Storage

Used as required throughout the computer program.

Method

See Section 3.3.

Other Routines Used

This controlling program uses subroutines STEP1, STEP2, STEP3, STEP4, STEP5, PLOT2, OLLER and HEAD, and function FCTION. In addition, certain plotting routines are called which were incorporated within the software of the computer system used.

Data Required

Some of the invariant data used are set within this program. The remaining data required is information concerning the flow conditions of the runs being analyzed. The data is required in the following order:

NJOB: The total number of runs to be processed.

NCASES: Not used in this particular version of the program.

- J: Any integer. This is used purely to ensure that a read error will be produced should the data cards get out of sequence.
- RRN: The number assigned to the particular run.
- AAP: Axial position of primary jet with respect to the secondary jet (in inches).
- TTT1: Primary jet temperature (^OR).
- TTT2: Secondary jet temperature (°R).
- PPR: Pressure ratio across the primary nozzle.
- AAR: The ratio of the area of the secondary nozzle to the area of the primary.
- VVR: The static pressure (reservoir pressure, psig) for the secondary nozzle. Immediately after this number is read in, it is used in conjunction with the data in VV1 to calculate the ratio of the secondary nozzle velocity to the primary nozzle velocity. This velocity ratio is then stored in this location.

c-446

- VV1: The static pressure (reservoir pressure, psig) for the primary nozzle. Immediately after this information is read in, it is used to calculate the exit velocity for the primary nozzle. This velocity is then stored in this location.
- RRMM: The mass flow through the secondary nozzle (lbs/min).
- EEMI: The mass flow through the primary nozzle (lbs/min).
- TTHR: The thrust of the model jet (lbs).
- TT: The ambient temperature (^OR).
- PP: The ambient pressure (inches of mercury).
- HH: The ambient relative humidity (percent).
- CCF: The calibration factor relating millivalt output of the octave band analyses to sound pressure (dynes per cm²/millivolt).
- TTB: The angle (in degrees) from the jet exhaust axis at which the total sound power integration was begun. In the present series of tests, this was constant at 12.5 degrees.

Further data is then required for subroutine AQUIRE.

Printing

This routine prints the number of experimental runs to be processed, followed by the invariant data set within the program. As each run is processed, it prints out the data previously read in (as in the preceding section). Additional output is produced by subroutines STEP1, STEP2, STEP3, STEP4 and HEAD.

Miscellaneous

There is a limit of 200 on the total number of experimental runs that can be processed in any one run on the computer.

C-2 Subroutine AQUIRE

Purpose

Reads in, checks and converts to floating point form the sound level data from each experiment.

Argument List

NR:

The number of the run for which acoustic data is sought.

- VX: Storage for the acoustic data as processed by this subroutine.
- ISTOP: This is set equal to +1 if an error is found in the data (in which case the controlling program A1901 does not attempt to process the data). Otherwise, ISTOP equals zero.

Common Storage

For details, see Controlling Program A1901.

Method

The routine searches for the character "L." Having found one, it checks to see if the next three (3) characters from the run number NR. This process is continued until the correct run number is found. The following thirty-six (36) characters are then processed in groups of three (3) to form millivolt values corresponding to measured sound pressure levels. It searches for 21 groups of such data and processes them.

Data Required

This routine requires 21 groups of forty (40) characters. Each group should begin with the character "L." The next 3 characters in each group constitute the run number. The remaining twelve (12) sets of 3 characters correspond to 12 sound pressure level readings. These 12 are a reference level, ten (10) actave band levels (up to 31.5 kHz band) and the overall sound pressure level. The 21 groups of 40 characters are the sound levels at 5-degree intervals, starting at 15 degrees from the axis of the jet.

Miscellaneous

There are three (3) error stops in this routine. They occur if (1) there is too much data, (2) there is too little data, or (3) a character other than a numeral or "L" is found. In the first two cases, ISTOP is set equal to unity. In the second, the whole computer run is aborted.

C-3 Subroutine STEP1

Purpose

Converts the millivolt sound level readings to decibel readings relative to 0.0002 dynes/cm².

Common Storage

See Controlling Program A1901.

148 C-6

Method

The data obtained by subroutine AQUIRE is converted first to sound pressure in dynes per square centimeter (arrays X and V) and then to sound pressure levels re 2×10^{-4} dynes per square centimeter (array Y). In this conversion, the following correction factors determined from the instrumentation checkout (see Section 3.2.2) were included.

Octave Band Center Frequency (kHz)	0.5	1	2	4	8	16	31.5	Overall
Correction	0	0.5	1.5	1.0	2.0	3.0	4.5	6.0

The reference level and 63 Hz, 125 Hz and 250 Hz octave bands are ignored since they are not relevant to this experiment. In addition, the overall level is calculated from the sum of the octave band levels. This value replaces the experimentallydetermined value which will not be correct because of the coupling between the spectrum shape and the correction factors given in the above table.

Printing

The octave band and overall sound pressure levels in decibels for the model jet are printed for twenty-one (21) angular positions at 5-degree intervals, starting 15 degrees from the axis of the jet exhaust.

C-4 Subroutine STEP2

Purpose

To determine the total power radiated from the model jet.

Common Storage

See Controlling Program A1901.

Method

The radiated sound from the jet is assumed to be cylindrically symmetric about the jet axis. The total sound power (in watts) is therefore found by integrating the sound intensity over a sphere centered on the jet nozzle. This integration is carried out both for overall sound power radiated and for the power in the octave bands. The power level is then determined by converting to decibels relative to 10^{-13} watts.

Finally, the overall sound power level is scaled to give the values for full-scale engines with 10,000 lbs, 20,000 lbs, 40,000 lbs and 80,000 lbs thrust.

Printing

The overall and octave band sound power (in watts) and sound power level (in dB relative to 10^{-13} watts) are printed for the model jet. The overall sound power levels for the four (4) full-scale jet engines are listed.

C-5 Subroutine STEP3

Purpose

To calculate the directivity pattern for the jet noise.

Common Storage

See Controlling Program A1901.

Method

The overall and actave band directivity are calculated based on the definition of directivity as the difference between the actual sound pressure level and the sound pressure level that would pertain if the total sound power output of the jet were radiated equally in all directions.

Printing

The overall and octave band directivity indices are printed, along with an indication of the noise floor for that particular octave band.

C-6 Subroutine STEP4

Purpose

To calculate the jet mechanical power and acoustic conversion efficiency.

Common Storage

See Controlling Program A1901.

15() C-8

Method

The mechanical power is defined in Section 3.3 and is calculated in watts. The acoustic conversion efficiency is the ratio of the total acoustic power to the jet mechanical power.

C-7 Subroutine STEP5

Purpose

To calculate the sideline overall and octave band sound pressure levels and the sideline perceived noise levels.

Common Storage

See Controlling Program A1901,

Method

The theoretical basis for this subroutine is given in Section 3.3. The four (4) extra octave bands are added in this fautine, assuming a constant decrease in level of 3 dB per actave above the 31.5 kHz Land. The routine allows for attenuation due to distance (inverse square law) and atmospheric attenuation. Two sideline distances (500 feet and 1500 feet) and four full-scale jet thrusts (10,000 lbr, 20,000 lbs, 40,000 lbs and 80,000 lbs) are considered. The positions on the sidelines correspond to five-degree angle increments starting at 15 degrees from the jet axis. The attenuation effects are colculated in the main part of this routine. Subrouting OLLER is used to calculate the perceived noise levels.

Other Routines Used

This routine calls subroutines OLLER, HEAD and FLOT2, and function subroutine FCTION.

Printing

Subroutine HEAD prints the sideline sound level information. (For further details see the specification for subroutine HEAD.)

C-8 Subroutine OLLER

Purpose

Calculates the sideline perceived noise levels (PNL in units of PNdB), given the sideline octave band sound pressure levels.

Argument List

F is a one-dimensional array giving the center frequencies of the actave bands.

TrIR is the thrust of the model jet.

THRI is the thrust of the full-scale engine.

St is the two-dimensional array containing the sideline actave band information.

PN is the array which contains (on exit from the routine) the sideline perceived noise level (PNdB).

Common Storage

See Controlling Program A1901.

Method

Outlined in Society of Automotive Engineers "Definitions and Procedures for Computing the Perceived Noise Level of Aircraft Noise," Aerospace Recommended Practice (ARP) 865, dated October 15, 1964.

Miscellaneous

- (1) The noy curves (tables) exist only for frequencies in the range 50 Hz ≤ f ≤ 10,000 Hz. The routine allows for lower frequencies by extending the noy curves as a constant (independent of frequency) below 50 Hz. At frequencies above 10,000 Hz, the contribution to the perceived noise is assumed to be negligible.
- (2) If any octave band sound pressure level should exceed 136.0 dB, the routine places 777.0 PNdB in the appropriate place in the PNdB table.

C-9 Subroutine HEAD

Purpose

a served the weather the provide served as

Prints the sideline octave band and overall sound pressure levels and the perceived noise level.



C-10

Argument List

THR is the thrust of the model jet.

THRI is the thrust of the full-scale jet.

SL is a two-dimensional array containing the octave band and overall sideline sound pressure levels.

PN is a one-dimensional array containing the sideline perceived noise levels.

AL is a one-dimensional array listing the positions along the sideline at which the sound levels, et cetera, are determined.

Common Storage

See Controlling Program A1901.

Printing

This routine prints sideline sound pressure level and perceived noise levels at twentyone (21) positions along the sideline. These positions are at 5-degree intervals from the jet axis, starting at 15 degrees from the axis. For each sideline position, the printing consists of:

- (1) The distance in feet to the point on the sideline at which the sound levels are calculated.
- (2) The perceived noise level at that point. The first value is obtained by including eleven (11) octave bands in the perceived noise level calculation. The value in parenthesis is the value obtained without using the top four (4) octave bands. These are the bands that were extrapolated in Subroutine STEP5.
- (3) The overall sound pressure level.
- (4) Eleven (11) octave band sound pressure levels. It should be noted that the center frequencies of these octave bands will vary because they are scaled from the model frequencies by the square root of the ratio of the model thrust to fullscale thrust.

C-10 Subroutine PLOT2

Purpose

Plots the perceived noise level on the 1500-foot sideline for full-scale jet engines with thrusts of 10,000 lbs, 20,000 lbs, 40,000 lbs and 80,000 lbs.

153 C-11

Argument List

X is the distance from the jet to the point on the 1500-foot sideline.

Y1, Y2, Y3 and Y4 are the column arrays containing the perceived noise levels for the four (4) full-scale jet engines.

RN is the run number being analyzed.

Method

The four (4) sideline perceived noise levels are plotted and labeled appropriately. The scales are not marked on the axes, but the vertical axis has a range of perceived noise levels from 50 PNdB to 130 PNdB while the horizontal axis gives distance along the sideline in the range -1000 feet to +6000 feet. The zero point on this axis corresponds to the point on the sideline nearest to the jet, i.e., the point at which a line from the jet, at right angles to the jet axis, intersects the sideline.

Other Routines Used

Subroutines PLOT, SYMBOL and NUMBER were used. These were part of the FORTRAN operating system on which this complete computer program was run and therefore were not provided as separate routines with the program card deck. Sub-routine PLOT is the routine for drawing lines and moving the pen and subroutines SYMBOL and NUMBER are the routines for writing Hollerith characters and numbers, respectively, on the graph.

Miscellaneous

The output from this routine was plotted off-line on a CALCOMP plotter.

C-11 Function FCTION

Purpose

Used in computation of atmospheric attenuation.

Method

Calculates an attenuation factor (rom a measured attenuation curve as outlined in Section 5.

N ale C-12
Argument List

J is an integer indicating the frequency being considered. THRI is the full-scale jet thrust.

Common Storage

See Controlling Program A1901.



:

Intentionally Left Blank

_	PROGRAM A1901(INPUT.OUTPJT.TAPE60=INPUT.TAPE61=OUTPJT.TAPE7)
	REAL MI&M2&LI&L2&LC
<u>c</u>	****
	CU4MON/DAT1/ FF(24), M1(24), L1(24), LC(24), M2(24), L2(24)
<u></u>	COMMON/DATE/ ICH,PI,ABD,AB1,AB2
	COMMON/DAIJ/ RNAPJTIITTZ-PRARAVAVIARMAEMIATARATAPAACFAB
	$\frac{COMMON/DAITY}{COMON(DAITS)} = \frac{COMON(DAITS)}{COMON(DAITS)} = \frac{COMON(DAITS)}{COMON(DAITS)$
с	SPARA COUNTRY AND
	DIMENSION BUFF (1024)
	DIMENSION DO(1)
	EQUIVALENCE (DD(1),RN)
<u>_</u>	****
	$COMMON EM_{EM2}V2 + X(12,21) + Y(12,21) + A(21) + W(11) + PW(11) + DI(11,21)$
	$\frac{\text{COMMON CE(1)} + \text{SLL1}(15+21) + \text{SLL2}(15+21) + \text{AL2}(21) + \text{AL2}(21)}{\text{COMMON CE(1)} + \text{AL2}(21)}$
	COMMON PNIL(21), PNI2(21), PNI3(21), PNI4(21), PNI(21), PN2(21)
~~~	$\begin{array}{c} COMMON \ PN21(21), \ PN22(2), \ PN23(21), \ PN24(21) \\ COMMON \ S111(15, 21), \ S112(15, 21), \ S112(15$
	COMMON SLIT(15+21/+SLI2(15+21/+SLI3(15+21/+SLI4(15+21) COMMON SLIT(15+21+SLI2(15+21)+SLI3(15+21)+SLI4(15+21)
C	****
-	DIMENSION RRN(200) + AAP (200) + TTT1 (200) + TTT2 (200) + PPR (200) + AAP (200) +
	1VVR(200) + VV1(200) + RRMM(200) + EEM1(200) + TTHR(200) + TT(200) + PP(200) + HH
	2(200) + CCF (200) + TTR (200) + JJ (200) + KK (273) + ODBNS (200)
_	EQUIVALENCE (JJ(1), HH(1)) • (KK(1) • V(1))
<u>_C</u>	****
	$DATA (AB0=0.127032) \cdot (AB1=1.338286) \cdot (AB2=-0.207140)$
*	$\frac{0.414}{(1-1)} (1-1) = 1 = 1 = 1 = 0 = 1 = 0 = 0 = 0 = 0 = 0$
	$DATA (B=10.667) \cdot (V = 1.2) \cdot (DT=0.273) \cdot (DT=43.5)$
	DATA (ICH=81) • (PI=3.14159265)
	DATA (XR1=>00.0) + (XR2=1500.0)
	DATA (THR1=10000.0).(THR2=20000.0).(THR3=40000.0).(THR4=30000.0)
	DATA ((F(I),I=1,14)=63.0,125.0,250.0,500.0,1000.0,2000.0,4000.0,90
	UALA ((RR(1)))=+1=19+0)=+1+1=15+01 IAF(+4=EC0)++1+1[++++NS T++++ER+++E
*****	234HEODM AHEO ILAHNERAH CON AHEACAH FOAAA HI A SAHARAAA PER
	3ROSR 44AM W 4HRITT 4HEN 5 4HOR T 4HHE C 4HOCA ANGRO A 4H A FT AJPET
	4 .4HTEST.4H ON .4H12/1.443/68.4H.)
	DATA ((FF(I)+I=1+24)=50.0+63.0+80.0+100.0+125.0+160.0+200.0+250.0+
	1315.0,400.0,000.0+630.0,800.0+1000.0+1250.0+1500.0,2000.0,2500.0,3
	1150.0.4200.0.5000.0.6300.0.8000.0.10000.0)
	DATA ((M1(1), I=1, 24)=0.043478, 0.04057, 2(0.035331), 0.035336, 2(0.033))
~~~	(333) (0.032031 (0.030613) (0.030103) (0.02996) (2(0.042285))
•	DATA ((1,C(1), 1=1),24) = 91,01,85,88,87,32,79,85,79,75,75,75,96,73,96,74,
	191+94+63+13(0+0)+44+29+59+72)
	()ATA ((M2(1)+1=1+24)=15(0+030103)+9(0+029960))
	DATA ((L2(1)+I=1+24)=52.0+51.0+49.0+47.0+46.0+45.0+43.0+42.0+41.0+
	15(40.0),38.0,34.0.32.0.12.(29.0),30.0,31.0,34.0,37.0)
<u></u>	******
	0170100100 001280140314001028(6999)
	EWITVALENCE (AP + TAP) + (TTP) + (TAP, AP) + (TUD, U3) + (T300, 300)
	CALL PLOTS(BUFF+1024+7)

で、生活に注意になって

CALL PLOT(0.0,0.3,-3)
READ (60-1) NJOB+NCASES
1 FVKMAI (213) Waite (414600)
500 FORMAT (1)
WRITE (61+2) NJ03
2 FORMAT (104.34HANALYSIS OF JET NOISE EXPERIMENTS./10X.33(1H-)///10
1X+15+2X+28HEXPERIMENTS TO BE PROCESSED.////)
IF (NJOB.GT.200) 3.4
<u>3 STOP</u>
4 WRITE (61,5) (DD(1), $i=17,20$) E contration for structure of structure of the set of the structure of the set of the s
1/10/40HSPECIFIC ACOUSTIC IMPEDANCE OF ATP AT STO CONSCIENTS
2-FIG.4/10X -28HMTCR0PHOVE BOOM SWEEF BADIUS.32X-2HA -FIG.4/10X-25HM
3ICROPHONE SWEEP VELOCITY, 35X, 2H= +F10.4/10X, 334TIME BETWEEN SEAJEN
4TIAL PRINTOUTS 27X, 2H= 1510.4)
WRITE (61+0) (DD(I)+I=42+47)
6 FORMAT : 10%, 24HE IRST SIDE-LINE DISTANCE, 36%, 24= +F10, 4/10%, 25HSECO
IND SIDE-LINE DISIANCE+35X+2H= +F10+4/10X+34HFINST THRUST REFERENCE
2 FOR SCALING-25X+2H= +F11+4/10X+35HS2COND_IHUJST REFERENCE FOR SCA
JLING;234;27= ; IV+4/IV4;34HIMIKU (RKJS) KEFENCE VCE FUK SCHLING;20X;
WRITE (61.7) (TH(I).I=1.2))
7 FORMAT (10%,54HANGULAR POSITIONS OF MICROPHONE TO JET STREAM (THET
14)// 7X.21F6.1////
WRITE $(61+8)$ (F(I)+I=4+10)
B FORMAT. (104.444041GINAL FREQUENCIES OF OCTAVE BAND CENTERS.//10X.
READ (60.11) J. RRN(I)
111 FORMAT (15+F10+0)
READ (60+9) AAP(I)+TTT1(I)+TTT2(I)+PPR(I)+AAR(I)+VVR(I)+VV1(I)+RRM
9 FURMAI (85400) DEAN 144,2015EMI/IN THUD/IN TT/IN-BO/IN HU/IN-CARINA ATM/AN
10 CONTINUE
D0 16 I=1,NJ08
RN=RRN(I)
TT1=TTT1(I)
IF (AAP (1) . GE 0. 5) 400 . 401
<pre>>>1 A1E(AA1(1) + 14*0)\(AA4(1) + 14*0)</pre>
90 10 402
400 CONTINUE
PR=PPR(I)
V1 = (VV1(1) + 14.6)/14.6
V1=(5,0*(V1++0,2357) = 1,0))+#0.5
VX= (5.0+(VX++0.28571 - 1.0))++0.5
VV1(1) = TTT1(1)/(1.0 + (V1**2)/5.0)
VVR(1)= TTTZ(1)/(1.0 + (VR++2)/5.0)
<u>VV1(I)=1120.0*(VV1(I)/520.0)**0.5</u>
VV4([/=1]20.0"(VVR(]/520.0)**0.5

409	5 CONTINUE
	TdA = (VV) (I) + VVA (I)
	Z CONTINUÉ
C 11	DTETHE NUZZLE VELOCITIES ARE CALCILATED FROM THE MEASURED PRESSURE
<u> </u>	RATIOS AND STATIC TEMPERATURES. WITH RETRACTED PRIMARY THE
C	CALCULATED SECONDARY VELOCITY WILL BE WRONG IF THE PRIMARY
<u> </u>	TEMPERATURE DIFERRS FROM THE SECONDARY TEMPERATURE. FOR
C	COPLANER NOZZLES AND EXTENDED PRIMARY, HRY, IR SI IS CALCULATED
~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	THUS IS THE MEASURED THINKING IN THE MEASURED
Ŷ	
	$EE^{(1)}(I) = EE^{(1)}(I) / 60_{0}(I)$
	EM1=EEM1(I)
<u> </u>	<u>T=TT(I)</u>
	P=PP(I)
	· 버글러버(I)
	CF=CCF(I)
	DB NS= (LCF (1)/U,033245) == 2,0
	IF (172/1)20.25.20
ZŽ	IAPEITT2=IA3=EVR=IRMM=104 NONE
	<u>60 TO 30</u>
50	ENCODE (8,930+AP) 4AP (1)
· · · · · · · · · · · · · · · · · · ·	ENCODE (8,930+1(2) TTT2(1)
	ENCODE (8+980+AN) AAR(I)
080	ENGLUE (8170)KMM) KKMM(1)
	CONTAILE
5.	WRITE (61.600)
•	WRITE (61.11) (DO(1).1=1.6)
	FORMAT (10% JOHRUN NUMBER. SOX, 2H= + F8.3/10X. STHAXIAL POSITION OF
	IPRIMARY WRT. SECONDARY (INS.) + 13X+2H= +AB /10X+23-PRIMARY TEMPER
·	2ATJRE (R) + 37X+2H= + F8+3/10X+25HSECONDARY [EHPERATURE (R) + 35X+2H=
	3+A9 /10X+22HPRIMARY PRESSURE RATIO+39X+2H3 + F8+3/10X+10HAREA RA
	-113,502,214 + 48 /102,164VELOCITY RAIL0,462,242 + AB /108,254Pq1
	DMART VELOCITY (FI/DECI):304:244 + F0:3)
12	PRATE LOTTEL MASS FLOW BATTOLESKAPHE AR (LAX-200001043) MACA
• •	IFLOW (LASEC) +31x-24± + Fq.3/10X-124THRUST (LASEADX-24± - Fq.3
	2/10X, 29HENVIRONIENTAL TEMPERATURE (R), 31X, 2H= + F8, 3/10X, 30HENVIDO
	IN TENTAL PRESSURE (IN H3, 30X, 2H= , FB, 3/10X, 33 HENVIRONYE VTAL HUMID
	AITY (PER CENT) +2/X+24= + F8-3/10X+35+CALIBAATION FACTOR (MV TO DY/
	550 C41,25X,2H= , FR.3)
	WHITE (61+970) DUNS
- 913	FUMAL (104 32HIVSTRUMENTATION NUISE FLOOR (33) + 26X 2H= +
	1 FO.38 ////95% 45(1H=)8 ////) 16/7779/111 95.44.96
	AFTILIGIL// 32440430
	RHMERRHM(1)

40	CONTEN
40	NRNATHY (RN)
15	CALL AQUIRE (NRN+V+ISTOP)
	IF(ISTOP,E3,1) 16,777
777	CONTINUE
	CALL STEP1
	CALL SIGHE
	CALL STEP4
	CALL STEPS
16	CONTINUE
<del></del>	CALL PLOT (0.0; 0:0; 999)
17	FORMAT (114) OF RUNN
	STOP
	END
	, , , , , , , , , , , , , , , , , , ,
	SHUDANTINE ADUTOE INDAWA TETADA
	DIMENSION V(13-22) ATT (801-04/13-21)
	CUMMON/DAT2/ ICH
	ISTOP=0
	<u>KJ=1</u>
	PFAD (60.3) (TT(T), T=1, 00)
3	FURMAT (8081)
	ICHel
	I=II(ICH)
	NIRO & KIRO
5	
	IF (ICH.GT.80) 6.7
6	READ (60,3) (11(1),1=).80)
0	IF (NX.EQ.NR) 16.9
9	IF (MK.EQ.1) 1.10
10	1F (KJ-22) 30, 31, 34
30	WRITE(61:998) NR Fromat (196) 108 100000 Million (197) 15, 7 100100 (1990) 15, 1000
32	ISTORES
	GO TO 36
31	JS¤1
	JE#21
38	
	<u>Kakel</u>
	D0 33 I=1+13
-	_VX(I+N)#V(I+J)

adr. and manifester floor

C-18

33	CONTINUE
36	
34	1F(KJ.GT.23) 35,37
35	WRITE (61+997) MR
997	FORMAT (1H0,10X,10HRUN NJMBER,15,2X,14HT00 MJCH DATA,)
	60 TO 32
37	JS=2
	JE=22
	WRITE (61,996)
996	FORMAY (2X+48HBEMARE. THERE ARE 22 PIECES OF DATA IN THIS RUN. )
	60 10 32
C++++	******
<u>c 1</u>	THE ABOVE THREE CARDS SHOULD BE REPLACED BY THE STATEMENT 30 TO 38
C	IF THE OPTION OF PROCESSING 22 PIECES OF DATA PER RUN IS REQUIRED.
C	
16	
	15 (NX) 27165 27
60	$V(X_1, K_2) = 0.2$
67	V(RIORU) = FLOAT(NA)
<u> </u>	
<u> </u>	
10	4071 4071 (0033) (11(1)/12/202)
10	
4 7	
	1F(1,40, 18(1)55,23
55	
	G0 T0 5
23	IF(1.4GT-36) 21+24
24	NX=NX=10+I-27
25	CONTINUE
	GO TO 16
21	WRITE (61122)
22	FORMAT (10X,27HINPUT ERROR IN SUBR. AQJIRE)
	STOP
	510

C-19

\$

SUBROUTINE STEPT
REAL M1.M2.L1.L2.LC
COMMON/DAT1/ FF (24) + M1 (24) + L (24) + LC (24) + H2 (24) + L2 (24)
COMMON/DATZ/ ICH.PI.AB0.AB1.AB2
COMMON/DAT3/ RN. AP. TT1. TT2. PR. AR. VR. VI. RMM. EMI. THR. T. P. H. CF. TB
COMMON/DAT4/ RC,R+VM+DT+TH(21)+XR1+XR2+THR1+THR2+THR3+THR4+F(14)
COMMON/DAIS/_N+N1+NC+M+V(13+21)
COMMON EM+EM2+V2+X(12+21)+Y(12+21)+A(21)+W(11)+PW(11)+DI(11+21)
OMMON_CE(1)+SLL1/15+21)+SLL2(15+21)+AL1(21)+AL2(21)
COMMON PN11(21), PN12(21), PN13(21), PN14(21), PN1(21), PN2(21)
COMMON PN21(21) PN22(21) PN23(21) PN24(21)
COMMON SLI1(15+21)+SL12(15+21)+SL13(15+21)+SL14(15+21)
COMMON_S[21(15+21)+S[22(15+21)+S[24(15+21)+S[24(15+21)-
V(I+J)=V(I+J)*CF
100 CONTINUE
D0 1 I=1,N1
KeI
X(1+K)=V(2+I)
Y(1,K) = ALQG10(X(1+K) = X(1+K)) = 10+0+74+0
00 1 3=3,13
X (L gK) = V (J f 1) V (L gK) = V (J f 1)
Ceeese
C THIS LOOP INSERTS THE FOLLOWING OCTAVE BANG CORRECTION FACTORS
C 250HZ 02H.
C 1000HZ +0.50H.
C 2000HZ +1.52H.
C 4000HZ +1.02H.
6000HZ +2.038.
<u> </u>
Y ( 6+13) = Y ( 6+13) + 0.5
Y(7,13)=Y(7,13) + 1.5
$Y(-\theta + 13) = Y(-\theta + 13) + 1 + 0$
Y ( 9913)¤Y ( 9913) ◆ C+U
Y(10+13)=Y(10+13) + 3+0
$\frac{Y(10,13) \pm Y(10,13)}{Y(11,13) \pm 3.0}$
$\begin{array}{c} Y(10,13) = Y(10,13) & 0 \\ Y(11,13) = Y(11,13) & 0 \\ Y(11,13) = Y(11,13) & 0 \\ Y(11,13) = Y(12,13) & 0 \\ Y(12,13) = $
$\begin{array}{rcl} Y(10,13) \equiv Y(10,13) & < 2.0 \\ Y(10,13) \equiv Y(10,13) & < 3.0 \\ Y(11,13) \equiv Y(11,13) & < 4.5 \\ Y(12,13) \equiv Y(12,13) & < 6.0 \\ V(17,13) \equiv X(16,13) & = X(16,13) & < (10.0) & < (0.5/20.0) \end{array}$
$\begin{array}{c} Y(10,13) \equiv Y(10,13) + 2.0 \\ Y(10,13) \equiv Y(10,13) + 3.0 \\ Y(11,13) \equiv Y(12,13) + 6.0 \\ Y(12,13) \equiv Y(12,13) + 6.0 \\ V(7,13) \equiv X(6,13) \equiv X(6,13) + (10,0) + (0,5/20,0)) \\ V(8,13) \equiv X(7,13) \equiv X(7,13) + (10,0) + (10,5/20,0)) \\ V(8,13) \equiv X(7,13) \equiv X(7,13) + (10,0) + (10,0) \\ V(10,0) + (10,0) + (10,0) + (10,0) \\ V(10,0) + (10,0) + (10,0) + (10,0) \\ V(10,0) + (10,0) \\ $
$\begin{array}{c} Y(10,13) \equiv Y(10,13) + 2+0 \\ Y(10,13) \equiv Y(10,13) + 3+0 \\ Y(11,13) \equiv Y(11,13) + 4+5 \\ Y(12,13) \equiv Y(12,13) + 6+0 \\ V(17,13) \equiv X(16,13) \pm X(16+13) + (10+0) + (0+5/20+0)) \\ V(17,13) \equiv X(17,13) \pm X(16+13) + (10+0) + (10+0/20+0)) \\ V(17,13) \equiv X(17,13) \pm X(17,13) + (10+0) + (10+0/20+0)) \\ V(17,13) \equiv X(17,13) \pm X(17,13) + (10+0/20+0)) \\ V(17,13) \pm X(17,13) + (10+0/20+0)) \\ V(17,13) \pm X(17,13) + (10+0/20+0)) \\ V(17,13) \pm X(17,13) + (10+0/20+0) \\ V(17,13) + (10$
$\begin{array}{c} Y(10,13) \equiv Y(10,13) + 2.0 \\ Y(10,13) \equiv Y(10,13) + 3.0 \\ Y(11,13) \equiv Y(12,13) + 4.5 \\ Y(12,13) \equiv Y(12,13) + 6.0 \\ V(7,13) \equiv X(10,13) + X(10,13) + (10,0) + (10,0) \\ V(7,13) \equiv X(10,13) + X(10,13) + (10,0) + (10,0) \\ V(10,13) \equiv X(10,13) + X(10,13) + (10,0) + (10,0) \\ V(10,13) \equiv X(10,13) + X(10,13) + (10,0) \\ V(10,13) \equiv X(10,13) + X(10,13) + (10,0) \\ V(10,13) \equiv X(10,13) \pm X(10,13) + (10,0) \\ V(10,13) \equiv X(10,13) + (10,13) + (10,13) \\ V(10,13) \pm X(10,13) + (10,13) + (10,13) \\ V(10,13) \pm X(10,13) + (10,13) + (10,13) + (10,13) \\ V(10,13) \pm X(10,13) + (10,13) + (10,13) + (10,13) \\ V(10,13) \pm X(10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) \\ V(10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) \\ V(10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) \\ V(10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) \\ V(10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,13) + (10,$
$\begin{array}{c} Y(10,13) \equiv Y(10,13) + 2.0 \\ Y(10,13) \equiv Y(10,13) + 3.0 \\ Y(11,13) \equiv Y(12,13) + 4.5 \\ Y(12,13) \equiv Y(12,13) + 6.0 \\ Y(12,13) \equiv X(10,13) \pm X(10,13) + (10,0) + (0,5/20,0)) \\ Y(10,13) \equiv X(10,13) \pm X(10,13) + (10,0) + (10,0/20,0)) \\ Y(10,13) \equiv X(10,13) \pm X(10,13) + (10,0) + (10,0/20,0)) \\ Y(10,13) \equiv X(10,13) \pm X(10,13) + (10,0) + (10,0/20,0)) \\ Y(11,13) \equiv X(10,13) \pm X(10,13) + (10,0) + (10,0/20,0)) \\ Y(12,13) \equiv X(11,13) \equiv X(11,13) + (10,0) + (10,0/20,0)) \\ Y(12,13) \equiv X(11,13) \equiv X(11,13) + (10,0) + (10,0/20,0)) \\ Y(12,13) \equiv X(11,13) = X(11,13) + (10,0) + (10,0/20,0)) \\ Y(12,13) \equiv X(11,13) = X(11,13) + (10,0/20,0) \\ Y(12,13) \equiv X(11,13) = (10,0/20,0) \\ Y(12,13) = (11,0/20,0) \\ Y(12,13) = (11,0/20,$
V(10,13) = V(10,13) = 2.0 $V(10,13) = V(10,13) = 3.0$ $V(11,13) = V(10,13) = 3.0$ $V(11,13) = V(12,13) = 6.0$ $V(7,13) = x(12,13) = 6.0$ $V(7,13) = x(12,13) = x(16,13) = (10,0) = (1,5/20,0)$ $V(10,13) = x(16,13) = x(17,13) = (10,0) = (1,5/20,0)$ $V(10,13) = x(16,13) = x(16,13) = (10,0) = (1,5/20,0)$ $V(10,13) = x(10,13) = x(10,13) = (10,0) = (1,5/20,0)$ $V(11,13) = x(10,13) = x(10,13) = (10,0) = (4,5/20,0)$ $V(12,13) = x(11,13) = x(11,13) = x(12,13) = (10,0) = (4,5/20,0)$ $V(13,13) = x(12,13) = x(12,13) = (10,0) = (4,5/20,0)$

	DD 9 I=19NC
	SUM=0
	00 8 K=5,11
9	SUM=SUM+10+0++(Y(K+I)/10.0)
	Y(12,1)=10.0*AL0010(SUM)
9	CONTINUE
	WRITE (61+2)
2	FORMAT (104.17 HEXPERIMENTAL DATA /104.79 HTABLE SHOWING DETANE BAND
	AND OVERALL SOUND PRESSURE LEVEL VARIATION WITH ANGLE ///)
	WRITE (61+12) (F(1)+1=4+10)
12	FURMAT (6X+SHANGLE+17X+65HOCTAVE BAND SOUND PRESSURE 1
	1EVELS +17X++HOVER /6X+5H(DEG)+ 5X+7(F6+0+4X)+5X+3HALL /)
	D0 3 1=1+NC
	Jat
	WRITE(61.4) TH(J)+(Y(K.1)+K=5.12)
4	FORMAT (5x+F6.1+ 5x+7(F6.1+4x)+2x+F6.1)
3	CONTINUE
•	WRITE (61-5)
5	FORMAT (/////)
•	DQ A JataN
• • •	XXEV (2a.)
	V(2.J) #V(13.J)
_	V(13.J) #XA
6	CONTINUE
	15.[s[ 7 00
	XX=x (1, J)
	(L+51)X#(L+1)X
	X(12+1)=XX
	X4=Y(1+J)
	(L.S()Ya(L.)Y
	Y112+J)=XX
7	CONTINUE
	RETURN
	ENO

CHAGANTTAC PTCA-
RCAL MISMERLISLESLE
CUHUN/DATI/ FF (24) , 41 (24) , L1 (24) , L2 (24) , M2 (24) , L2 (24)
COMMUN/DATE/ ICH+FI,AJ0+AJ1+AB2
COMMON/DAI3/ RNOAPOTILOTIZOPROAROVIORIMAEMIOINROTOROTOROTORO
COMMON/DAT4/ RC+4+VH+OT+TH(21)+XA1+XA2+THA1+THA2+THA3+THA4+F(14)
COMMON/DATS/ N+41+NC+H+V(13+21)
COMMON EM+EM2+V2+X(12+21)+Y(12+21)+A(21)+W(11)+B(11)+D1(11-21)
COMMON CE(1)+3LL1(15,21),SLL2(15,21),AL1(21),AL2(21)
COMMON PN11 (21) PN12(21) PN11211 PN14(21) PN1 (21) PN2(21)
COMMON PN21 (21) . "N22(21) . GN23(21) . BN24(21)
COMMON SI 11 13-21 ACT 21 ACT SI 17 15-16 TAKE 21
COMON SUST 115-2115-2116-2116-2116-2116-21-6126(13461)
Rudano-29472/P-34R1(1/519-0)
16-12-5-P//180-0
<u></u>
T2 = (TH(1, +2,3)) PI/180,
$A(1) = PI = 2 \cdot Q = R = R = (COS (T1) = COS (T2))$
1 CONTINUE
₩(i)=0.0
D0 3 J=5.11
W(J) = 0 = 0
00.2 1=1.00
MET3MMET3+ME(1)
PW(J)=ALOGIV(W(J))=10.0+130.0
J CONTINUE
$= \frac{P_{W}(1) = 10, 0^{4} + 0010(10, 0^{4} + 0^{2})/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(5)/10, 0) + 10, 0^{4}(0^{4}(0^{4}(5)/10, 0)) + 10, 0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0^{4}(0$
17)/10.0) +10.0** (P*(8)/10.0) +10.0** (P#(9)/10.0) +10.0** (P#(10)/10.0)
2+10.2++(Pw(11)/10.0))
WRITE (61+45)
AS FORMAT (1M1///S14 ACOUSTIC POWER AND SOUND POWED LEVEL COD MODEL )
PET 1
NR17F (A) (4)
A FURNATIZZATA ONFUFUIEVEN, AV. 1 DADONED / MATTEL AND THE AND THE
MITTER ALLAND ALLANDA
A CIDMAY FAR WIGHTAN
0 / VIGR: (04//10/258L108/216-5//8///4//
001(1-2+13
WHITE (61,8) F(J) + (1) + 4(1)
10 CONTINUE
8 FORMAT (7X) F6.0.84 E12.5,7X F7.1)
WRITE (61+100)
100 FORMAT (///444 OVERALL SOUND POWER LEVEL.SCALED FOR THE STATES ANT
ARUST 6X. 154PUSET LEVEL (24)
¥(2)=***(1) + (1)+(1)0(201A+(1)+(1)+(1)+(1)+(1)+(1)+(1)+(1)+(1)+(1)
WHITE (61,101) THH: w(2)
W(2) = 4 10 = 10 = 10 = 10 = 10 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 = 2 =
WRITE ISI INI THE ATEN
HTALE LOJELUJI ETNJENEZI NITVETNI OVALA V LODEN NI ODANI U STATU.
HELLE, (01:01) THE # (2)
101 FUHAT (74. F6.0, HX. F7.)
WHITE (61+7)
ゲージのヨロムアー トレスノノノノン
HETUAN

SUBROUTINE STEP3
REAL M1.M2.L1.L2.LC
CO4MON/DAT1/ FF(24),41(24),L1(24),LC(24),M2(24),L2(24)
544.144.014.1514.1514.014.00
COMMON/DAT3/ ANAAPATTIATT2+PR+AR+VR+VI+RHH+EMI+THR+T+P+H+CF+TB
COMMON/DAT4/ RC. 7. VM. 01. TH(21). XA1. XA1. THA1. THA2. THA3. THA4. F 114)
COMHON/DAT5/ N:N1:NC:4.9(13.21).JBNS:1 VD(11)
CO4NON E4.EH2.V2.X112.21).V(12.21)A.(21)A.(21).H(11).04(11).01(11.21)
COMMON CE (1) + SLL1 (15+21) + SLL2 (15+21) + AL1 (21) + AL2 (21)
COMMON PHI1(21) + PHI2(21) + PHI2(21) + PHI2(21) + PHI2(21)
CUMMON PN21(21). PN22(21), PN23(21) PN24(21)
CUMMON SLI1(15+21)+SL12(15+21)+SL12(15+21)+SL14(15+21)
CONNON SL21(15,21), SL22(15,21), SL23(15,21), SL23(15,21)
DIMENSION ISLLI(209)
EQJIVALENCE (ISLL1()) SEL1(1)
COV=AL0G1019)+20.0+10.5
<u>90 i I=1.11</u>
CC=P#(I)
DO 1 JelovC
DI(1,J)=Y(1,J)=CC+COV
1 CONTINUE
¥7178 (5+18)
2 FORMAT 11M1/10X.29MTABLE OF DIRECTIVITY INDICES./10A.29(1H-1///10X_
1.6HTHETA 37 THIVERALL. AX.6HSOO HZ. SX.5H1 KHZ. SX.6HZ KHZ. SX.5H
24 KHZI 5XI5HR KHZI 4XI6HI6 KHZI 3XI843145 XHZI /1
DO 3 J#1+NC
ENECOP (110, 4, ISLL) (DILLAJAIRIAL)
4 FORMAT (1)(FO.2. 4K))
KEJ.
WRITE (61+5) TH(K)+ISLL1(1)+(ISLL1(I)+I#S+11)
<u>5 FOGHAT (108+FG-1+2X-8(2X+AG))</u>
3 CONTINUE
00.20 [Els]]
- P#(1)=08NS+P#(1)+CDN
SO CONTINUE
NRITE (61+61) PH(1)+(P#(1)+1=5+11)
21 FORMAL JULIES LASH TABULATED VALUES ARE DOUGTFUL IF THEY ARE LESS
THAY THESE VALUES (//161.8(F10.2))
END

0.0

	SUAROUTINE STEPA
	REAL MILERELIELES
	CUMMON/DAT12 \$F1241.231:241+11(241+1C(24)+H2(24)+12(24)+12(24)
	60 MMON/DATE (54:5: 400.55), 482
	CONTRACTOR TAL REAR TELETTERENARE VALUE AND VALUE AND THE PART OF THE CONTRACT OF THE CONTRACT.
	COMPOSION AC, 4, VH. OT, TH(2); + +41== 42+THR1+T42+TH43+T44+F11+1
	19990N204152 NoN1+NC+M52113:211
	QUARDE EM-EM2+V2+X122+21)+Y12+211+A(21+++11)+P#(11)+)(11+21)
	20 MON (2 11) + 21 1 1 1 2 11 5 1 2 11 5 1 2 11 5 1 2 1 1 2 1 1 4 2 2 1 2 1
	60-400 44114211, 34121211, P414(21), P414(21), P41(21), P42(21)
	CONNE PREI (211. PREZ (211. PREZ 1211. PREZ (211
	0==10= Stil: 15.21).Sti2:15.21).Sti3:15.21).Sti4:15.21)
	10 10 1 1 2 1 1 1 4 2 1 1 5 2 2 1 1 3 2 1 1 5 1 2 3 1 1 5 2 3 1 1 5 1 2 4 1 3 2 5 1
	291146LERE : F4444AMA
	<u>\$9481-0619+7376+2.8+32.171</u>
	SF (1444 .2.2. 104 NONE 1 20110
È.	204 × V2 = 0.
	00 10 30
10	CONTINUE
	Енгентент
	ASEATEAN
30	Enacom+1Ex7+A1+A1+EnS+A5+45+45+
	##17E (61+E) E4+CE(1)
<u>.</u>	FORMAT 141/1 102 20H JET SECHANICAL POWER, DE 244 + E14451 2
	2104 21-4 CONVERSION EFFICIENCY 24 2H+ + E14.51
	RETURN
	END .

inthi i

ຳອ່ງ C-24

SUBROUTINE STEPS
REAL HI.+H2+L1+L2+LC
COMMON/DAT1/ FF (24) + 11 (24) + L1 (24) + L2 (24) + L2 (24)
COMMON/DATZ/ ICH.PI.AB0.A31.AB2
COMMON/DAT3/ RN. AP. TT1 . TT2 . PR . AR . VR . VI . RMM . EMI . THR . T. P. H. CF . TB
COMMON/DA14/ RC+R+V4+DY+TH(21)+XR1+XR2+THR1+14R2+THR3+T4R4+F(14)
COMMON/DAIS/ NeNLENCEMEV(13:21)
COMMON CC(1) = C(1) +
COMMON PAIL 1211 - PAIL 2/211 - PAIL 4/211 - PAIL (211 - PAIL 211
COMMON PN21 (21) . PN22 (21) . PN23 (21) . PN24 (21)
COMMON SILL (15,21) + SL12(15,21) + SL13(15,21) + SL14(15,21)
COMMON SI 21 (15, 21) + SL22/15, 21) + SL23/15, 21) + SL24/15, 211
DO 3 1=1,tC
AL1(1)=XR1/SIN (TH(1 )+P1/180.0)
AL2(I)=XR2/SIN (TH(I )+PI/180.0)
WZ1=AL0G10(AL1(1)/R)*20,0
WZ = ALOGIO(ALZ(I) + R) + 20.0
SLIS(1,1)=1(JFI)=₩Z]
1 CONTINUE
$DO \land J=1.4$
 1 + ا = 1 ا
J2=J+10
SLL1(J1,1)=SLL1(J2,1)=3.0
SLL2(J1,I)=SLL2(J2,I)-3.0
4 CONTINUE
<u>\$511#5512#5513#5514#5521#5523#5524#0.0</u>
WZ1=AL1(]'/(UV).
00.2 122.15
ALF = 10.**FCTION(J.THR1)
SF = 10.*ALOGIO(THR1/THR)
SL11(J,I)=SLL1(J,I)-ALF##Z1 +SF
SL21(J+1)=SLL2(J+1)=ALF++Z2 +SF
ALF = 10. #*FCTION(J, THR2)
SF = 10.4ALOGIO(THR2/THR)
SLCC(J)=SLLC(J+I)=ALF##22 +SF ALF = 10 ##F(TION/1,TH33)
SE = 10, #410/10/10/10/07/1001
$SL13(J_0I)=SL11(J_0I)=\Delta LF##71 + SF$
SL23(J+1)=SL(2(J+1)=ALF*#Z2 +SF
ALF = 10. **FCTION(J. THR4)
SF = 10.*ALOGIO(THR4/THR)
SL14(J,I)=SLL1(J,I)=ALF##7.1 +SF
SL24(J+I)=SLL2(J+I)=ALF+#Z2 +SF
SS11=SS11+10.0**(SL11(J+1)*0.1)
5512=5512+10+0##(5112(J+1)#0+1)
2013#2012*10 08#(SL)4(1.1)#0.1)
5521=5521+11,0##(5121(1+1)#0.1)
SS22=SS22+10+0** (SL22(J+1)*0.1)
5523=5523+10.0**(5L23(J.1)*0.1)
\$\$24=5\$24+10.0**(\$L24(J+1)*0.1)
ALF = 10.**FCTION(J.THR)

SLL1(J+I) = SLL1(J+I) = ALF#WZ1
SLL2(J+I) = SLL2(J+I) - ALF + HZ2
2 CONTINUE
SL11(1+I)=AL0G10(SS11) #10.
SL12(1+I)=ALOG10(SS12) #10.
SLI3(1+I) = 4LOG10(SS13) + 10
SL14(1+1)=4L0G10(5514) +10.
SL21(1)1)=AL0G10(SS21) #10.
- SL22(1+I)=4L0G10(SS22) #10.
SL23(1+1)=AL0610(SS23) +10.
\$L24(1+1)=4L0G10(\$\$24) +10+
3 CONTINUE
CALL OLLER (F.THR.+THRI+SLI1.PN11)
CALL OLLER (F.THR. THR. S' 12. PN12)
CALL OLLER (F.THR.THR3.5113.PN13)
CALL OLLER (F.THR.THR4.SL14.PN14)
CALL OLLER (F, THR, THR, SL21, PN21)
CALL OLLER (F, THR, THR2, SL22, PN22)
CALL OLLER (F.THRATHRASSIZS,PN23)
CALL OLLER (F,THR,THR4,S', 24, PN24)
CALL HEAD (THR, THRI, SLII, PNII, ALI)
CALL HEAD (THR, THR2, SL12, PN12, ALL)
CALL HEAD (THR:THR:SSLI:SPN:SVALI)
CALL HEAD (THR, THR4, SL14, PN14, AL1)
CALL HEAD (THR, THRI, SL21, PN21, AL2)
CALL HEAD (THR, THR2, SL22, PN22, AL2)
CALL HEAD (THR;THR;SSL23;PN23;AL2)
CALL HEAD (THR.THR.4.SL24.PN24.AL2)
CALL PLOT2(AL2, PN21, PN22, PN23, PN24, RN)
RETURN
END ,

	SUBROUTINE OLLER (F, THR, THRI, SL, PN)
	REAL M1+M2+L1+L2+LC+LCC
	COMMON/DAT1/ FF(24) + 1(24) + L1(24) + LC(24) + M2(24) + L2(24)
	DIMENSION F(1) + SL(15+1) + PN(1)
	D0 5 I=1.21
	DO 100 J=2+15
	IF (SL (J+I)+GT+135+0) 101+100
101	PN(I)=777.7
	GO TO 5
100	CONTINUE
	XMAX=SMAX=0.0
	00 4 J=2,15
	FREQ=F(J-1)+SQRT(THR/THRI)
	IF (FREQ.GE.10000.0) 4,26
26	IF (FREQ.LT.50.0) 27.6
5	AW=AM+(SL(J,I)-ALO)
	Aw=10,0**Aw
	SMAX=SMAX+Aw
<u> </u>	IF (AW.GT.XMAX) 3.14
3	
14	1+ (J, E0, 11) 30, 31
30	AW#XMAX+()+2*(SMAA*XMAA)
	SL(2)1/=40.0453,34ALUGIO(AW)
31	
4	
	AW#AMAAY()+37 (3784778MAA) Dalitiyaa (3433,384) (3410,43)
9	
27	EFF0#52.0
6	
	$IF (FHEQ_GT_FF(K)) 7.8$
7	CONTINUE
	STOP
8	K1=K-1
	K2=K
	F1=FF(K1)
	F2≖FF (K2)
	Y1=LC(K1)
-	Y2=LC(K2)
	LCC=Y1+(Y2=Y1)*(FREQ=F1)/(F2=F1)
	IF (SL(J,I).GE.LCC) 9,10
2	AM=M2(K1)+(M2(K2)-M2(K1))*(FREQ-F1)/(F2-F1)
	ALO=L2(K1)+(L2(K2)-L2(K1))+(FREQ-F1)/(F2-F1)
	<u>GO TO 2</u>
10	AM=M1(K1)+(M1(K2)-M1(K1))*(FREQ-F1)/(F2-F1)
	ALJEL1(K1)+(L1(K4)+L1(K1))+(FREQ+F1)/(F2+F1)

-----

7414

BRANNIN WAN HISAN WANNA

÷

-

:

.....

168 C-27

SUBROUTINE HEAD (THR. THRI, SL. PN. AL) DIMENSION AL(1) . PN(1) . SL(15,1) . FM(14) REAL MI +M2+L1+L2+LC COMMON/DAT3/ RN COMMON/DAT4/ RC. R. VM. DT. TH (21) , XR1 , XR2 , THR1 , THR2 , THR3 , THR4 , F (14) DO 10 J=1+14 FM(J)=F(J)*SORT(THR/THRI) 10 CONTINUE WHITE (61+1) RN+THR, THR1 1 FORMAT (6(/) 1H1: 10X 10+RUN NUMBER: F10.0: 4(/) * 10X 15HMODEL THRUST = + F10.3.5X 20HFULL SCALE T 1HRUST = + F10.3./// 5X 2HL., 7X 5HPND3., 5X.5HDASPL.12X 45HOCTAVE BAND SOUND PRESS PRESSURE LEVELS ) 7 FORMAT (29X,11(F8.1)/) D0 2 I=1+21 WRITE (61+6) AL(I)+PN(I)+SL(2+1)+SL(1+I)+(SL(J+I)+J=5+15) 2 CONTINUE 6 FORMAT (1X+F7.1+F7.1+1H(+F5.1+1H)+F7.1+11(F8.2)) RETURN END FUNCTION ECTION (J+THRI) REAL M1+M2+L1+L2+LC COMMON/DATZ/ ICH, PI. ABO. ABI. ABZ COMMON/UAT3/ RN+AP+TT1+TT2+PR+AR+VR+V1+RMM+EM1+THR+T+P+H+CF+TB COMMON/DAT4/ RC.R.VM.DT.TH(21) +XR1+XR2+THR1+THR2+THR3,THR4+F(14) FREQ = F(J-1)+SQRT(THR/THR1)++001 XZ=ALOGIO (FREQ) FCTION = ABO+XZ*(AB1+A92*XZ) RETURN END

	SUBROUTINE PLOTE (X. YI. YE. YE. YE. RN)
	DIMENSION X(21), Y1(21), Y2(21), Y3(21), Y4(21), ICHAR(4)
	ICHAR(1)=44 10K\$ ICHAR(2)=44 20K \$ ICHAR(3)=44 40K
	ICHAR(4)=4H 80K
	D0 75 1=1,21
	X(I) = SQRT(X(I) + 2 - 2250000.0)
_75	CONTINUE
	D0 76 I=17,21
	X([) == X([)
76	CONTINUE
	CALL PLOT(7.09 0.0. 2)
	$CALL PLOT(ADJU_0U) = J$
1	CONTINUE
	CALL PLOT (0.0. 0.0. 3)
	CALL PLOT(0.0. 8.0. 2)
	YD=8.0
	00 2 I=1.8
	CALL PLOT(0.0. YD. 3)
	CALL PLOT(0.05, YD, 2)
	YD=YD-1.0
	CONTINUE
	A7740UU177(1)+100
	$CA(1 - D) OT (XDA, 364YD_ 30A3)$
	CALL SYMBOL (XP+. 32. YP 47 16. ICHAR(1). 0. 4)
	CALL PLOT(XP, YP, 3)
	00 3 1=2+21
	XP=_001#X(I)+1.0
	YP=0,1+Y1(I)=5.0
	CALL PLOT(AP+ YP+ 2)
3	CONTINUE MB- AAAM(1)
	VP=0.1=V2/11_5.0
	CALL PLOY (XP+.05+YP-3)
	CALL PLOT (AP+. 35. YP 13. 2)
	CALL SYMROL (XP+, 32, YP-, 21, , 16, ICHAR(2), 0, 4)
	CALL PLOT (AP, YP, 3)
	00 4 1=2,21
	XP#,001*X(I)+1.0
	YPE0.1*Y2(1)-5.0
	CALL PLUI(AP) TP) 2)
	XP=.001+X(11+1.0
	YP=0.1*Y3(1)=5.0
	CALL PLOT(AP+.05, YP, 3)
	CALL PLOT (XP+. 35. YP+. 13. 2)
	CALL SYMBOL (XP+. 32, YP+. 05 16. ICHAR(3). 0. 4)
	CALL PLOT(AP, YP, 3)
	DC DC DC DC DC DC
	Are 001 WAIL10100
	TE-NALTIJIA/77AU CALI DLOY/XDA VDA DI
	ማግኘው የተለማት የግን የግን የግን የሚ

0.000

NG YOUNG DO

•••

C-29

50	CONTINUE
	XP=.001+X(1)+1.0
	YP=0.1*Y4(1)-5.0
	CALL PLOT (XP++05+YP+3)
	CALL PLOT (XP+.35+YP+.39-2)
	CALL SYMBOL (XP+, 32, YP+, 31, , 16, ICHAR (4), 0, 4)
	CALL PLOT (XP, YP, 3)
	00 60 1=2+21
	XP=,001+X(I)+1,0
	YP=0.1+Y4(I)-5.0
	CALL PLUT (AP. YP. 2)
60	CONTINUE
	ICHAR(1) = 4HRUN S ICHAR(2) = 4HNO.
	CALL SYMBOL (.25. 8.515. ICHAR(1). 0. 4)
	CALL SYMBUL (999., 999., .16. ICHAR (2), 0, 4)
	CALL NUMBER (999. + 999. + 16+ PN+ 0+ -1)
	CALL PLOT (10.0.0.03)
	RETURN
	FND

大きなななないとしていたいできたのというですのでき

÷

171 C-30

.....

#### APPENDIX D

# SUMMARY OF OVERALL SOUND POWER LEVEL DATA

This appendix contains a set of 28 graphs which present the overall sound power level data, scaled to a 20,000 pound thrust engine, as a function of the ratio of secondary mass flow to primary mass flow. Each graph corresponds to one of the principal groups of runs defined in Table A-1.



いたいのないで、「たい」

Figure D-1. Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.





はたちたいもうちょう ちょうちょう

いたいない、「ないないないないない」としていました。

Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Construction Come

and the bound of the content of the bound of the bound of the bound of the state of the state of the second of t

Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



í.

「おいろいろ」というです。

2

ためのは、「ない」となったいであるないで、ないないないので、

STORY REPAIRS

Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure D–1. continued ..., Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.





Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



•• •

ľ

.....

Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



and a second second

and a state of the second s

Design of the second







CALLOR AND A PARTY

Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.

7/83 D-12



Figure D-1. continued ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.







Figure D-1. concluded ... Variation in Overall Sound Power with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.

# Intentionally Left Blank

### APPENDIX E

مريان ترتيا فالمحافظ

いたい、日本の人名の日本により

### SUMMARY OF MAXIMUM PERCEIVED NOISE LEVEL DATA

This appendix contains a set of 28 graphs which present the maximum perceived noise level data on the 1500-foot sideline, scaled to a 20,000 pound thrust engine, as a function of the ratio of secondary mass flow to primary mass flow. Each graph corresponds to one of the principal groups of runs defined in Table A-1.

187

E-1



A STATE STATE OF

Section Section of the

States and the Children of a

Figure E-1. Variation in Maximum Perceived Noise Level on a 1500-Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.

E-2


ないとない日本の方法

ł

والمعمد بمتراد متحدر

Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500-Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure E-1. continued ... Variation in Maximum Perceived Naise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scoled to 20,000 Lbs.



E--5



Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500-Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.





2000

. . . . . . .

Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500-Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



gure E–1. continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



and the address in the particular sector and so it is a sector of the base

Figure E-1. continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.



りたけおれていてい

Ş

1

Figure E–1, continued ... Variation in Maximum Perceived Noise Level on a 1500–Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.

24) () E-14



Figure E-1. concluded ... Variation in Maximum Perceived Noise Level on a 1500-Foot Sideline with Bypass Ratio for Thrusts Scaled to 20,000 Lbs.

.

201

Intentionally Left Blank

e en presidente en estadada da da data en en da bienen (press data), da data da data da da da da da da da da da

## APPENDIX F

# PRIMARY JET SOUND POWER SPECTRA AND DIRECTIVITY DATA

This appendix contains twelve (12) graphs which give the sound power spectra and directivity data measured for the three (3) primary nozzles. For each nozzle, the power spectra are given for three (3) temperatures, followed by three (3) graphs containing the associated octave band directivities for each temperature condition.



Scholaron Paintana



293

F-2





F-3



Figure F-3. Directivity Pattern for Run Number 144 (Pressure Ratio 1.6, Temperature 450°F).

F-4









F--6

intervention some some





F-7



Figure F-7. Directivity Pattern for Run Number 157 (Pressure Ratio 2.5, Temperature 450°F).

ာရှုမှ F-8



,

Figure F-8. Directivity Pattern for Run Number 188 (Pressure Ratio 2.5, Temperature 800°F).







'' ( ]} F−10





Figure F–11. Directivity Pattern for Run Number 166 (Pressure Ratio 3.5, Temperature 450°F)

# F-12 ?13





?14

F-13

Intentionally Left Blank

.

#### APPENDIX G

### NOISE OF A SUPERSONIC NOZZLE OPERATING AT OFF-DESIGN PRESSURE RATIOS

It is to be expected that supersonic nozzles of jet engines, if designed for high-speed cruise at high altitude, may be operating at off-design pressure ratios at takeoff. Therefore, it was of interest to explore the noise generated during off-design operation to determine whether any anomalies may be present. A small number of exploratory runs were conducted for this purpose.

It was felt that if any anomalies (such as discrete tones) existed, they would certainly be apparent in a cold flow (i.e.,  $60^{\circ}$ F). Therefore, the exploratory runs were made with a cold flow, using a supersonic nozzle designed for pressure ratio 3.5. When discrete tones occur in a test, even if they are not detectable by ear, they are very apparent from spikes in the directivity pattern for the octave band in which they occur. The computer printout data from these runs was reviewed and no pure tone anomalies were found.

Seven (7) runs were performed within a one-hour pariod, using operating pressure ratios ranging from 2.0 to 7.8, and repeating the on-design pressure ratio condition. Since the computer printout of the data lists the design pressure ratio but not the operating pressure ratio, that information is given in the table below. (For all other runs in the project, operating pressure ratio and design pressure ratio were nominally the same.)

Run Number	Primary Supply Pressure, psig	Operating Pressure Ratio
600	14.6	2,0
601	21,9	2,5
602	29,2	3.0
603	36.5	3.5
604	43.8	4.0
605	63.0	5.4
606	99.5	7,8

Run 131 provides another data point of on-design aperation for the same nozzle. However, it was run at a different time, with a lower supply temperature, and had a slightly higher mass flow and a 2.1 dB higher acoustic power when compared to Run 603 in this series. Therefore, it is not used for comparison with these sequential results.

The directivity index values are shown for alternate runs in Figure G-1. There is a general trend to a sharper directionality (without shift in angle) as the operating pressure ratio is increased, but there are no apparent anomalies or disagreements with normal jet noise data. In terms of the octave band spectra at  $0 = 45^{\circ}$ , Figure G-2, there is no significant change in spectrum shape for off-design operation.

Figure G-3 shows the effect of off-design operation on the model scale sound power generation with the  $U^8$  slope mean data line for the primary nozzles for reference. Figure G-4 shows similar data for the 1500-foot sideline maximum perceived noise level. Here, the  $U^8$  slope data curve for the other primary nozzles almost fits, indicating that the off-design conditions affect the power level more than they uffect the perceived noise level. The reason may be that the sharper directivities for the nozzle, when operated above design velocity, tend to reduce the relative amount of noise radiated to the sideline.

It should be mentioned that in all runs except Run 600, the original data showed broadband noise in the 16 kHz and 31.5 kHz bands, which was removed in the same way as in the production runs for on-design cases (see Section 4.2). It is known that over-pressure operation of a convergent nozzle can produce what is called "shock cell noise," Reference G-1, and has even done so on full-scale engines operating at altitude, Reference G-2. Such noise is characterized by discrete tones of u highly directional nature, whose fundamental frequency (for a stationary jet) can be predicted from:

$$f_1 = \frac{\alpha}{3d (R - R_c)^{1/2}}$$
 (G-1)

where

f₁ = fundamental frequency, Hz

R = operating pressure ratio

R_c = critical pressure ratio, normally 1.89 (but should be design for a convergent divergent nozzle)

a molent speed of sound, ft/sec

d = jet exit diameter, ft

The original data (before correction) was reviewed. Anomalies were noted in most of the runs for the upper two octave bands of data (and sometimes in the upper three octave bands: B kHz, 16 kHz and 31.5 kHz) at  $\theta = 100 - 110$  degrees.

The results for 110 degrees and the upper three octave bands are given in Figure G-5. No anomalies in the directivities were observed at the two lower pressure ratios.

> ີ່ໄດ້ G-2



Figure G-1. Directivity Index for Overall Sound Pressure Level for a Supersonic Nozzle Operating at Off-Design Pressure Ratios.



Figure G-2. Octave Band Sound Pressure Spectra at 45 Degrees to Jet Axis for Supersonic Nozzle Operating at Off-Design Pressure Ratios.



Figure G-3. Effect of Off-Design Operation of the Sound Power Generation by a Supersonic Jet Nozzle.

alan yenentika eta mila







Figure G-5. Octave Band Sound Pressure Level at 110 Degrees, Showing Onset of the Third Anomaly Above 1200-ft/sec from Measured Raw Data.

ALL CONTRACTOR OF ALL

Therefore, the  $U^8$  slope reference line has been fitted through these points. Between the pressure ratios of 2.5 and 3.0, the anomaly just appears in the top two octave bands, with an increase of about 10 dB. At the design pressure ratio of 3.5, the 8 kHz octave band begins to increase; by a pressure ratio of 4.0, it is approximately 12 dB higher than might have been expected. Above the pressure ratio of 4.0, all three bands appear to be increasing with increasing pressure ratio more slowly – on the order of  $U^8$  or less.

These data clearly show that this anomaly is a function of pressure ratio and begins in the vicinity of the design pressure ratio. The two higher frequency bands were affected at a slightly lower pressure ratio than the 8 kHz band – roughly in accordance with the expectations of Equation (G-1). However, the 1/10th octave analysis shown in Section 4.2 did not indicate that pure tones are present, but that the phenomenon is at least an octave wide.

Although the actual optimum pressure ratio for the nozzle is probably slightly different than 3.5 because of boundary layer effects, it is considered that it could not be below 3.0 since the nozzle thrust coefficient was about .94, indicating a relatively thin boundary layer.

To obtain better insight on the problem requires additional runs using noise, schlieren photography and pressure probes as investigative tools.

G-8
## REFERENCES

- G-1. Powell, A., "On the Mechanics of Choked Jet Noise," Proc. Physical Society, B, Vol. LXVI, pp 1039–1056, 1953.
- G-2. Hay, J.A., and Rose, E.G., "In-Flight Shock Cell Noise," J. Sound Vib., Vol. II, Part 4, pp 411-420, 1970.



G-9